Access Services Tariff Section 8 4th Revised Sheet 1 Replacing 3rd Revised Sheet 1

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES

The Telephone Company may, at the option of an IC or end user, provide Billing & Collection services as specified by contract. Such services may include:

- Recording Service
- Billing Service
- Billing Analysis Service
- Billing Information Service
- Account Maintenance

Unless otherwise stated in this section, the term IC includes an end user.

8.1 Recording Service

Recording Service is the recording of the details of an IC message, and when requested by the IC, the provision of those details to the IC.

8.1.1 Rates and Charges

The rates and charges for recording service for all contract lengths up through five (5) years are listed below.

(AT) (AT)		1 Year <u>Rate</u>	More Than One <u>Year Rate</u>
(AT)	Recording Per Customer Message	\$0.0250	(NR) ICB
	Assembling & Editing Per Message	0.0050	
	Provision Of Message Detail-Per Message	0.0050	
	Per Tape Charge	45.0000	
	Data Transmission Sent or Received		
	Per Record using CMDS Network	0.0030	İİ
	Data Transmission Sent or Received Per		
	Record Using Customer Provided Network	0.0010	İİ
(AT)	Overnight Delivery of Mag Tape	90.0000	(NR) ICB

8.2 Billing Service

8.2.1 Message Processing Service

Message Processing Service is the transforming of the recorded IC call details into rated messages in preparation for billing.

8.2.2 Bill Processing Service

Bill Processing Service is the preparation of bills for message-billed service and bulk-billed service, mailing of statements of the amounts due for service received from the IC and the collection of the deposits and monies due from the End Users.



P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 8 except for the purpose 3rd Revised Sheet 1 Replacing 2nd Revised Sheet 1 of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES The Telephone Company may, at the option of an IC or end user, provide (AT) (AT) Billing & Collection services as specified by contract. Such services may include: RECEIVED - Recording Service SEP 6 1991 - Billing Service - Billing Analysis Service - Billing Information Service MISSOURI - Account Maintenance Public Service Commission (AT) Unless othwerwise stated in this section, the term IC includes an end user. 8.1 Recording Service Recording Service is the recording of the details of an IC message, and when requested by the IC, the provision of those details to the IC. 8.1.1 Rates and Charges The rates and charges for recording service for all contract lengths up through five (5) years are listed below. Rate Recording Per Customer Message \$0.0250 Assembling & Editing Per Message 0.0050 Provision Of Message Detail-Per Message 0.0050 CANCELLED Per Tape Charge 45.0000 Data Transmission Sent or Received Per Record using CMDS Network 0.0030 Data Transmission Sent or Received Per 0.0010 JUL 141994 90.0000 JUL 141994 Record Using Customer Provided Network BY 4 Th R.S#1 Overnight Delivery of Mag Tape Public Service Commission 8.2 Billing Service MISSOURI 8.2.1 Message Processing Service Message Processing Service is the transforming of the recorded IC call details into rated messages in preparation for billing. 8.2.2 Bill Processing Service Bill Processing Service is the preparation of bills for messagebilled service and bulk-billed service, mailing of statements of the amounts due for service received from the IC and the collection of the deposits and monies due from the End Users. FILED Issued: SEP 0 9 1991 Effective: NOV 0 2 1991 1991 NOV 2By R. D. BARRON, President-Missouri Division

> Southwestern Bell Telephone Company St. Louis, Missouri

MO. PUBLIC SERVICE COMM.

Access Services Tariff Section 8 2nd Revised Sheet 1 Replacing 1st Revised Sheet 1 and Original Sheet 1.01 **RECEIVED**

(CP) ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES

NOV 2 4 1987

The Telephone Company may, at the option of an IC, provide Billi MSSOURI Collection services. Such services may include: Public Service Commission

- Recording Service
- Billing Service
- Billing Analysis Service
- Billing Information Service
- Account Maintenance
- 8.1 Recording Service

Recording Service is the recording of the details of an IC message, and when requested by the IC, the provision of those details to the IC.

8.1.1 Rates and Charges

The rates and charges for recording service for all contract lengths up through five (5) years are listed below. CANCELLED

Recording Per Customer Message Assembling & Editing Per Message Provision Of Message Detail-Per Message Per Tape Charge Data Transmission Sent or Received Per Record using CMDS Network Data Transmission Sent or Received Per Record Using Customer Provided Network Overnight Delivery of Mag Tape Rate 1991 \$0.0250 Y 3 R.S. 1 (CR) 0.0050 Y 3 R.S. 1 0.0050 Service Commission (CR) 45.0000 MISSOURI (NR) 0.0030

0.0010 (NR) 90.0000

8.2 Billing Service

8.2.1 Message Processing Service

Message Processing Service is the transforming of the recorded IC call details into rated messages in preparation for billing.

8.2.2 Bill Processing Service

Bill Processing Service is the preparation of bills for messagebilled service and bulk-billed service, mailing of statements of the amounts due for service received from the IC and the collection of the deposits and monies due from the End Users.

Issued: DEC 1 1987 By R. D. BARRON, President-Missouri Division 87-95 Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

he option of an IC, pro	AUG 2.8 1986 vide the following MISSUURI Public Service Commission
SERVICES	REGEIVED
	ccess Services Tariff Section 8 lst Revised Sheet 1 cing Original Sheet 1

ACCESS

8. BILLING AND COLLECTION SERVICES

The Telephone Company will, at t services:

- Recording Service
- Billing Service
- Billing Analysis Service
- Billing Information Service
- 8.1 Recording Service

(AT)

(CT)

(MT)

The Telephone Company will provide Recording Service in association with the offering of Feature Groups C and D Switched Access Service for IC messages that can be recorded by the Telephone Company provided automatic message accounting equipment. In addition, where the Telephone Company records the IC messages on manual tickets, the Telephone Company will provide Recording Service for the manual tickets.

The Telephone Company will provide Recording Service in its operating territory. The minimum territory for which the Telephone Company will provide Recording Service is all the appropriately equipped offices in a state operating territory for which the IC has ordered Feature Group C or D Switched Access Service.

For Feature Group C and D Switched Access Service, when answer supervision is provided by the IC premises, the term "IC message" used herein denotes a completed call originated by an IC's End User. An IC message begins when answer supervision from the premise of the ordering IC is received by Telephone Company recording equipment indicating that the called party has answered. A message ends when disconnect supervision is received by Telephone Company recording equipment from either the premise of the ordering IC or the IC's End User premise from which the call originated.

CANCELLED

JAN 1 1988

BY 2nd R.S.#1 Public Service Commission MISSOURI

F	RIL	E	Ū
---	-----	---	---

SFP 29 1986

Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 1

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES

DEC 20 CED

..... in marthe and

The Telephone Company will, at the option of an IC, provide the following services:

- Recording Service
- Billing Service
- Billing Analysis Service
- Billing Information Service

8.1 Recording Service

The Telephone Company will provide Recording Service in association with the offering of Feature Groups C and D Switched Access Service for IC messages that can be recorded by the Telephone Company-provided automatic message accounting equipment. In addition, where the Telephone Company records the IC messages on manual tickets, the Telephone Company will provide Recording Service for the manual tickets.

The Telephone Company will provide Recording Service in its operating territory. The minimum territory for which the Telephone Company will provide Recording Service is all the appropriately equipped offices in a state operating territory for which the IC has ordered Feature Group C or D Switched Access Service.

The term "IC message" used herein denotes a completed call originated by an IC's End User. An IC message begins when answer supervision from the IC terminal location is received by Telephone Company recording equipment indicating that the called party has lanswered. FAU IC message ends when disconnect supervision is received by Telephone Company recording equipment from either the IC terminal location or the IC End User's premises from which the call originated. SEP 2 9 1980

8.1.1 General Description

General Description BY $f \in S$, $f \in S$ and when requested by the IC, the provision of those details to the IC. Recording Service includes recording, assembly and editing and provision of recorded IC message detail.

Recording is the entering on magnetic tape or other acceptable media the details of IC messages originated through Switched Access Service for which appropriate answer and disconnect supervision $-i\frac{1}{2}\frac{1}{1+1}\frac{1}{2}$ has been received. Recording is provided 24 hours a day, 7 days - 1 1934

Assembly and editing is the aggregation of the recorded IC message 253 details to create individual messages and the verification that the data required for rating, in accordance with the standard format established by the Telephone Company, is present. Assembly and editing is performed at least once a week.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.1 Recording Service-(Continued)
- (AT) For Feature Group D Switched Access Service when answerphice Germinission not provided by the IC premises, the term "IC message" used herein denotes a call originated by an IC's end user where a start time (the recorded time at which the Feature Group D entry switch receives the first wink supervisory signal forwarded from the IC's point of termination) and a disconnect time are received by Telephone Company recording equipment. For purposes of measurement, the message begins when the wink supervisory signal is received and ends when disconnect supervision is received by Telephone Company recording equipment from either the premises of the ordering IC or the IC's End User premises from which the call originated.
- (AT) Feature Group C Switched Access Service calls where the IC premises does not provide answer supervision are not permitted.
- (MT) 8.1.1 General Description

Recording Service is the recording of the details of an IC message, and when requested by the IC, the provision of those details to the IC. Recording Service includes recording, assembly and editing and provision of recorded IC message detail.

Recording is the entering on magnetic tape or other acceptable media the details of IC messages originated through Switched Access Service for which appropriate answer and disconnect supervision has been received. Recording is provided 24 hours a day, 7 days a week.

Assembly and editing is the aggregation of the recorded IC message details to create individual messages and the verification that the data required for rating, in accordance with the standard format established by the Telephone Company, is present. Assembly and editing is preformed at least once a week.

CANC	ELLED	
3 A KI 1	1440	

JAN 1 1988

BY <u>2nd R.S. #1</u> Public Service Commission MISSOURI

FALED

SEP 29 1986

Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 1.01

REGEIVED

AUG 28 1986

MISSOURE

Access Services Tariff Section 8 5th Revised Sheet 2 Replacing 4th Revised Sheet 2

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Bill Processing Service-(Continued)

Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting of credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's services and/or Local Telephone Exchange Services to an End User. Where Local Telephone Exchange Service access is denied, access to IC service will also be denied.

8.2.3 Rates and Charges

The rates and charges for billing service are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

performed on an overtime basis.		
		More Than
	<u>1 Yr. Rate</u>	<u>1 Yr. Rate</u>
Message Rating Per Message	\$0.0050	(CR) ICB
Message Bill Processing Per Message	0.0300	
Message Bill Processing -		
Per Phrase Summary Records	0.0045	
Bulk Bill Processing Per Message	0.0200	
Message Bill Inquiry Per Message	0.0400	
Bulk Bill Inquiry Per Message	0.0040	
Private Line Bill Inquiry Per		i i
Bill Rendered	0.3500	İİ
Bill Rendering - Message Bill	0.4500	i i
Bill Rendering - Bulk Bill	0.4000	İİ
Bill Rendering - Private Line Bill	2.0000	İİ
Bill Phrase 5 Lines Per Phrase		i i
Per Bill	0.0200	İİ
Bill Page Per Page Per Bill	0.0350	i i
End User Adjustment Per Adjustment	2.5000	i i
Post Bill Message Investigation		i i
Per Case	3.0000	i i
Program Development Charge Per Hour	60.0000	i i
Data Transmission Sent or Received		İİ
Per Record Using CMDS Network	0.0030	İİ
Data Transmission Sent or Received		i i
Per Record Using Customer Provided		i i
Network	0.0010	i i
Clerical Staff Utilization Per Hour	45.0000	i i
CPU Utilization Per Hour	850.0000	i i
Retention of Records Per Tape		
Per Month	1.0000	(CR) ICB

Issued:

May 11, 1994 Effective: July 14, 1994 By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri



(CT) (CT)

(CT)

Access Services Tariff Section 8 4th Revised Sheet 2 Replacing 3rd Revised Sheet 2

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

- 8.2 Billing Service-(Continued)
- 8.2.2 Bill Processing Service-(Continued)

Public Service Commission Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting of credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's services and/or Local Telephone Exchange Services to an EMANCELLED User. Where Local Telephone Exchange Service access is denied access to IC service will also be denied.

8.2.3 Rates and Charges

The rates and charges for billing service are listed below BY _____ elements charged on an hourly basis, the charges are do Public Service Commission MISSOURI performed on an overtime basis. 5 Yr. Rate 1 Yr. Rate

	<u>1 Yr. Rate</u>	<u>3 Yr. Rate</u>	5 Yr. Rate
Message Rating Per Message	\$0.0050	\$0.0050	\$0.0050
Message Bill Processing Per Message	0.0300	0.0275	0.0250
Message Bill Processing -			
Per EMI Text Record	(NR)0.0045	(NR)0.0045	(NR)0.0045
Bulk Bill Processing Per Message	0.0200	0.0175	0.0150
Message Bill Inquiry Per Message	0.0400	0.0350	0.0300
Bulk Bill Inquiry Per Message	0.0040	0.0035	0.0030
Private Line Bill Inquiry Per			
Bill Rendered	0.3500	0.3000	0.2500
Bill Rendering - Message Bill	0.4500	0.4000	0.3500
Bill Rendering – Bulk Bill	0.4000	0.3500	0.3000
Bill Rendering - Private Line Bill	2.0000	1.5000	1.0000
Bill Phrase 5 Lines Per Phrase			
Per Bill	0.0200	0.0200	0.0200
Bill Page Per Page Per Bill	0.0350	0.0350	0.0350
End User Adjustment Per Adjustment	2,5000	2.5000	2.5000
Post Bill Message Investigation			
Per Case	3.0000	3.0000	3.0000
Program Development Charge Per Hour	60.0000	60.0000	60.0000
Data Transmission Sent or Received			
Per Record Using CMDS Network	0.0030	0.0030	0.0030
Data Transmission Sent or Received			
Per Record Using Customer Provided			
Network	0.0010	0.0010	0.0010
Clerical Staff Utilization Per Hour		45.0000	45.0000
CPU Utilization Per Hour	850.0000	850.0000	850.0000
Retention of Records Per Tape			FILED
Per Month	1.0000	1.0000	1 0000

Issued:

Effective:

NOV 1 2 1993 DEC 1 2 1993 DEC 1 2 1993 By M. H. SCHULTEIS, Division Manager-Regulatory & Industry Relations Southwestern Bell Telephone Company St. Louis, Missouri

MISSOURI Public Service Commission

RECEIVED NOV 08 1993

MISSOURI

JUL 141994 STAR,S#2

AT) (AT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Section 8 3rd Revised Sheet 2 Replacing 2nd Revised Sheet 2 RECEIVED JAN 16 1990

Access Services Tariff

MISSOURI Public Service Commission

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Bill Processing Service-(Continued)

Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting of credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's services and/or Local Telephone Exchange Services to GANAGELLED User. Where Local Telephone Exchange Service access is denied, access to IC service will also be denied.

8.2.3 Rates and Charges

Issued: JAN 1 7 1990

DEC 121993

4Th R.S.

The rates and charges for billing service are listed being Service Gomemission elements charged on an hourly basis, the charges are doubled MSSOURI performed on an overtime basis.

	<u>1 Yr. Rate</u>	<u>3 Yr. Rate</u>	5 Yr. Rate
Message Rating Per Message	\$0.0050	\$0.0050	\$0.0050
Message Bill Processing Per Message	0.0300	0.0275	0.0250
Bulk Bill Processing Per Message	0.0200	0.0175	0.0150
Message Bill Inquiry Per Message	0.0400	0.0350	0.0300
Bulk Bill Inquiry Per Message	0.0040	0.0035	0.0030
Private Line Bill Inquiry Per			
Bill Rendered	0.3500	0.3000	0.2500
Bill Rendering - Message Bill	0.4500	0.4000	0.3500
Bill Rendering - Bulk Bill	0.4000	0.3500	0.3000
Bill Rendering - Private Line Bill	2.0000	1.5000	1.0000
Bill Phrase 5 Lines Per Phrase	•		
Per Bill	0.0200	0.0200	0.0200
Bill Page Per Page Per Bill	(CR) 0.0350	(CR) 0.0350	(CR) 0.0350
End User Adjustment Per Adjustment	2.5000	2,5000	2.5000
Post Bill Message Investigation			
Per Case	3.0000	3.0000	3.0000
Program Development Charge Per Hour	60.0000	60.0000	60.0000
Data Transmission Sent or Received			
Per Record Using CMDS Network	0.0030	0.0030	0.0030
Data Transmission Sent or Received			
Per Record Using Customer Provided			
Network	0.0010	0.0010	0.0010
Clerical Staff Utilization Per Hour	45.0000	45.0000	45.0000
CPU Utilization Per Hour	850.0000	850.0000	850.0000
Retention of Records Per Tape			
Per Month	1.0000	1.0000	1.0000

Effective: FED 1 7 1990

FILED

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

FEB 17 1990

Public Service Commission

and the second second second second second second second second second second second second second second second

and the second second second second second second second second second second second second second second second

of dolineum

(CP) ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

8.

NO

RECEIVED

NOV 241987

MISSOURI Public Service Commission

8.2 Billing Service-(Continued)

8.2.2 Bill Processing Service-(Continued)

BILLING AND COLLECTION SERVICES-(Continued)

Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting of credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's services and/or Local Telephone Exchange Service to an End User. Where Local Telephone Exchange Service access is denied, FEB 17 1990 access to IC service will also be denied.

8.2.3 Rates and Charges

The rates and charges for billing service are Served below. For elements charged on an hourly basis, the charges be doubled if appw. For rate performed on an overtime basis.

Issued: DEC 1 1987	Effective: JAN	1 1988	JAN 1 1988
Per Month	(NR) 1.0000	(NR) 1.0000	(NR) 1.0000 FILED
Retention of Records Per Tape			, , , , , , , , , , , , , , , , , , , ,
CPU Utilization Per Hour	850.0000	850.0000	850.0000
Clerical Staff Utilization Per Ho		45,0000	45.0000
Per Record Using Customer Provid Network	(NR) 0.0010	(NR) 0.0010	
Data Transmission Sent or Receive			
Per Record Using CMDS Network	(CR) 0.0030	(CR) 0.0030	(CR) 0.0030
Data Transmission Sent or Receive		•••••••	
Program Development Charge Per Ho		60.0000	
Per Case	3.0000	3.0000	3.0000
End User Adjustment Per Adjustmer Post Bill Message Investigation	2+000	2.000	2.5000
Bill Page Per Page Per Bill End Uger Adjustment Per Adjustmer	(CR) 0.0500 at 2.5000	(CR) 0.0500 2.5000	(CR) 0.0500 2.5000
Per Bill	(NR) 0.0200	(NR) 0.0200	`
Bill Phrase 5 Lines Per Phrase			
Bill Rendering - Private Line Bil	ll (CR) 2.0000	(CR) 1.5000	(CR) 1.0000
Bill Rendering - Bulk Bill	0.4000	0.3500	0.3000
Bill Rendering - Message Bill	0.4500	0.4000	
Bill Rendered	0.3500	. 0.3000	0.2500
Bulk Bill Inquiry Per Message Private Line Bill Inquiry Per	0.0040	0.0035	0.0030
Message Bill Inquiry Per Message		0.0350	
Bulk Bill Processing Per Message		0.0175	0.0150
Message Bill Processing Per Messa		(CR) 0.0275	(CR) 0.0250
Message Rating Per Message	_ (NR)\$0.0050	(NR)\$0.0050	(NR)\$0.0050
-	<u>1 Yr. Rate</u>	<u>3 Yr. Rate</u>	<u>5 Yr. Rate</u>
performed on an overerme babio.	•		

JAN 1 1988 87-95

Southwestern Bell Telephone Company St. Louis, Missouri

By R. D. BARRON, President-Missouri Division

Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.1 Recording Service-(Continued)
 - 8.1.1 General Description-(Continued)

Public Service Commission Provision of the recorded detail is the provision of magnetic tapes containing the assembled and edited IC message detail and, when requested by the IC, data-transmitting the assembled and edited IC message detail to the IC. Except for lost or damaged records, the recorded detail will be available to the IC not more than five business days after the date all the detail requested by the IC was processed by the Telephone Company.

- 8.1.2 Undertaking of the Telephone Company
- (AT)A. When answer supervision is provided by the IC premises, the Telephone Company will record all IC messages carried over Feature Groups C and D Switched Access Service that are available to Telephone Company-provided (AT) recording equipment or operators. When answer supervision is not provided by the IC premises, the Telephone Company will record Feature Group D Switched Access Service messages that are available to Telephone Company-provided recording equipment or operators. Special arrangements must be made to provide these recordings; therefore, the time necessary to implement the service and the charges to provide these recordings will be determined on an individual case basis. The Telephone Company will record IC messages carried over a Feature Group A Switched Access Service. Unavailable IC messages (i.e., certain Feature Group C operator and TSPS messages which are not accessible by Telephone Companyprovided recording equipment or operators) will not be recorded. The recording equipment will be provided at locations selected by the Telephone Company. Assembly and editing will be performed on all IC messages recorded during the billing period established by the Telephone Company. Except as set forth in Paragraphs 8.1.2, F. and 8.1.3, following, recorded message detail from previous billing periods will not be recovered and made available to the IC.
 - B. A standard format for the provision of the recorded IC message detail will be established by the Telephone Company and provided to the IC. If an che course of Telephone Company business, it is necessary to change the iormat, the Telephone Company will notify the involved [5] six months prior to the change. JAN 1 1988

SEP 29 1986

Access Services Tariff

Replacing Original Sheet_2-

lst Revised Sheet 2

AUG 2 8 1980

MISSOURI

Section 8

Public Service Commission

Issued: AUG 29 1986

BY 2md R.S. #2 Public Service Commission

MISSOURI

(MT)

Effective: SEP 29 1986

Access Services Tariff Section 8 Original Sheet 2

BEC 20 000

83-253

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

8.1.1 General Description-(Continued)

Provision of the recorded detail is the provision of magnetic tapes containing the assembled and edited IC message detail and, when requested by the IC, data-transmitting the assembled and edited IC message detail to the IC. Except for lost or damaged records, the recorded detail will be available to the IC not more than five business days after the date all the detail requested by the IC was processed by the Telephone Company.

8.1.2 Undertaking of the Telephone Company

- A. The Telephone Company will record all IC messages carried over Feature Groups C and D Switched Access Service that are available to Telephone Company-provided recording equipment or operators. Unavailable IC messages (i.e., certain Feature Group C operator and TSPS messages which are not accessible by Telephone Company-provided recording equipment or operators) will not be recorded. The recording equipment will be provided at locations selected by the Telephone Company. Assembly and editing will be performed on all IC messages recorded during the billing period established by the Telephone Company. Except as set forth in Paragraphs 8.1.2, F. and 8.1.3, following, recorded message detail from previous billing periods will not be recovered and made available to the IC.
- B. A standard format for the provision of the recorded IC message detail will be established by the Telephone Company and provided to the IC. If, in the course of Telephone Company business, it is necessary to change the format, the Telephone Company will notify the involved IC's six months prior to the change.
- C. The recorded IC message detail provided to the IC will not be sorted to furnish detail by specific End Users, groups of End Users, by office or by location.
- D. Recorded IC message detail will be provided to an IC as set forth in Paragraph 8.1.2, E., following. The Telephone Company will determine the number of magnetic tapes or detailfiles required to provide the recorded message detail to the IC.

SFP 2 9 1986 HCE COMMISSION OF MISSOUN

JAN 0 1 1984

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

DE. HOUIS, HISSOUI

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 Original Sheet 2.01

REGEIVED

AUG 28 1986

MISSOURI Public Service Commission

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.1 Recording Service-(Continued)
- 8.1.2 Undertaking of the Telephone Company-(Continued)

(MT)

- C. The recorded IC message detail provided to the IC will not be sorted to furnish detail by specific End Users, groups of End Users, by office or by location.
- D. Recorded IC message detail will be provided to an IC as set forth in Paragraph 8.1.2, E., following. The Telephone Company will determine the number of magnetic tapes or data files required to provide the recorded message detail to the IC.

CANCELLED

JAN 1 1988

BY <u>2nd R.S.#1</u> Public Service Commission MISSOURI

SEP 29 1986

Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986

Access Services Tariff Section 8 3rd Revised Sheet 3 Replacing 2nd Revised Sheet 3

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.3 Rates and Charges-(Continued)

(CT) (CT)		<u>1 Yr. Rate</u>	More Than <u>1 Yr. Rate</u>
	Service Order:		
	Establishment of and Change to		
	Calling Cards Per Order	\$25.0000	(CR) ICB
	Private Line End User Activity		
	Per Order	40.0000	
	Bulk Billed End User Activity		
	Per Order	40.0000	İİ
	Other End User Order Activity	ICB	(CR) ICB

8.3 Billing Analysis Service

Billing Analysis Service is the provision of detection, investigation and deterrence of billing evasion activities.

8.3.1 Rates and Charges

The rates and charges for billing analysis for all contract lengths up through five (5) years are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

	<u>Yr. Rate</u>	More Than <u>1 Yr. Rate</u>
30 Day Report for Detection Per Office Continuous Scan Per Office Per Week Documentation Scan Per Line Per Case	\$700.0000 700.0000 700.0000	(NR) ICB
Hourly Rate *	75.0000	(NR) ICB
* To Include: Collection of Evidence Preparation of Affidavits Prosecution Summary Assisting Law Enforcement Expert Witness Analysis Expert Witness Testimony Coordination of Sources Review Services Identification Report Services Preservation of Evidence		

Issued:



(AT) (AT)

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

(CP) ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

8.2 Billing Service-(Continued)

8.2.3 Rates and Charges-(Continued)

Access Services Tariff Section 8 2nd Revised Sheet 3 Replacing 1st Revised Sheet 3

RECEIVED

NOV 2 4 1987

MISSOURI Public Service Commission

		<u>1 Yr. Rate</u>	<u>3 Yr. Rate</u>	<u>5 Yr. Rate</u>
Service Order:				
Establishment of and Change to				
Calling Cards Per Order	(NR)	\$25.0000	(NR)\$25.0000	(NR)\$25.0000
Private Line End User Activity			• • • •	
Per Order		40.0000	40.0000	40.0000
Bulk Billed End User Activity				
Per Order		40.0000	40.0000	40.0000
Other End User Order Activity	(NR)	ICB	(NR) ICB	(NR) ICB

8.3 Billing Analysis Service

Billing Analysis Service is the provision of detection, investigation and deterrence of billing evasion activities.

8.3.1 Rates and Charges

The rates and charges for billing analysis for all contract lengths up through five (5) years are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

	,	•
30 Day Report for Detection Per Office	(CR)	\$700.0000
Continuous Scan Per Office Per Week		700.0000
Documentation Scan Per Line Per Case		700.0000
Hourly Rate *	(CR)	75.0000

* To Include: CANCELLED Collection of Evidence Preparation of Affidavits Prosecution Summary Assisting Law Enforcement Expert Witness Analysis Expert Witness Testimony Public Service Commission Coordination of Sources **Review Services** Identification Report Services Preservation of Evidence

JUL,141994

Rate

FILED

JAN 1 1988 -95 Public Service Commission

Issued: **DEC** 1 1987 Effective: JAN 1 1988

MISSOURI

Access Services Tariff Section 8 1st Revised Sheet 3 Replacing Original Sheet 3

REGEIVED

APR 1.5 ----

MISSOURI

Public Service Commission

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.1 Recording Service-(Continued)
 - 8.1.2 Undertaking of the Telephone Company-(Continued)
 - E. At the request of an IC, magnetic tapes containing the recorded IC message details will be provided to the IC as part of Recording Service. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. Unless specified otherwise by the IC, the magnetic tapes will be sent to the IC via first-class U.S. Mail Service. However, the IC may pick up the magnetic tapes at a location designated by the Telephone Company or request that the detail on the magnetic tapes or in a data file be data-transmitted to the IC. When the IC message details are datatransmitted to an IC location, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis. When the IC does not wish to receive the recorded IC message details, and the Telephone Company receives notice from the IC at least two weeks prior to the date the details would be sent to the IC, the charge as set forth in Paragraph 8.1.7, following, does not apply.
 - F. At the IC's request, the Telephone Company will make a reasonable effort to recover recorded IC message detail previously made available to the IC and make it available again for the IC. The charge as set forth in Paragraph 8.1.7, following, will apply for all such detail provided. When the IC message details are data-transmitted to an IC location, the data transmission charges will be determined on an individual-case basis. Such a request must be made within 30 days from the date the details were initially made available to the IC.
 - 8.1.3 Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, the Telephone Company's liability for Recording Service is as follows:

CANCELLED

JAN 1 1988

	BY <u>2nd RS#3</u> Public Service Commission	
······································	MISSOURI	FILED
Issued:	Effective: MA	
APR 1 9 1985	By R. D. BARRON, President-Missouri Southwestern Bell Telephone Com St. Louis, Missouri	

(AT)

Access Services Tariff Section 8 Original Sheet 3

DEC 2

1 (93 -

83-253

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

- 8.1.2 Undertaking of the Telephone Company-(Continued)
 - E. At the request of an IC, magnetic tapes containing the recorded IC message details will be provided to the IC as part of Recording Service. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. Unless specified otherwise by the IC, the magnetic tapes will be sent to the IC via first-class U.S. Mail Service. However, the IC may pick up the magnetic tapes at a location designated by the Telephone Company or request that the detail on the magnetic tapes or in a data file be data-transmitted to the IC. When the IC message details are data-transmitted to an IC location, the data transmission charges will be determined on an individual-case basis. When the IC does not wish to receive the recorded IC message details, and the Telephone Company receives notice from the IC at least two weeks prior to the date the details would be sent to the IC, the charge as set forth in Paragraph 8.1.7, following, does not apply.
 - F. At the IC's request, the Telephone Company will make a reasonable effort to recover recorded IC message detail previously made available to the IC and make it available again for the IC. The charge as set forth in Paragraph 8.1.7, following, will apply for all such detail provided. When the IC message details are data-transmitted to an IC location, the data transmission charges will be determined on an individual-case basis. Such a request must be made within 30 days from the date the details were initially made available to the IC.
- 8.1.3 Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, the Telephone Company's liability for Recording Service is as fallows:

MAY 2 0 1985 Mr. PUBLIC SERVICE COMMISSION OF MISSOURI

DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

Access Services Tariff Section 8 4th Revised Sheet 4 Replacing 3rd Revised Sheet 4

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service

Billing Information Service is the provision of information to the IC from Telephone Company record systems labeled as Customer Records Information System (CRIS) and Customer Name and Address Bureau (CNA).

8.4.1 Rates and Charges

Rates and charges for Billing Information Service for all contract lengths up through five (5) years are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

performed on an overtime basis.	1 Year <u>Rate</u>	More Than <u>1 Yr. Rate</u>
Program Development Charge Per Hour	\$60.0000	ICB
Data Transmission Sent or Received		
Per Record Using CMDS Network	0.0030	
Data Transmission Sent or Received		
Per Record Using Customer Provided		
Network	0.0010	
Clerical Staff Utilization Per Hour	45.0000	
CPU Utilization Per Hour	850.0000	
Standard SWBT CRIS Billing Information		
Per Record	0.0006	
CNA Information Per Request	1.5000	
CNA Info Written Confirmation Per Request	0.7500	
Overnight Delivery of Mag Tape	90.0000	
Billing Name and Address (BNA), per request (CR)	8000	ICB
(DR)		

8.5 Account Maintenance

Account Maintenance is the provision of customer information to update the IC's data base, for example, billing name and address changes, number changes, etc.

8.5.1 Rates and Charges

Rates and charges will be determined on an Individual Case Basis (ICB).

8.6 Liability of the Telephone Company

If detailed information required to provide Billing and Collection Service is not available because the Telephone Company lost or damaged records or tapes or incurred system outages, the Telephone Company will attempt to



(AT) (RT)

No Supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 3rd Revised Sheet 4 Replacing 2nd Revised Sheet 4 RECEIVED

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service

MAY 10 1994

MISSOURI

Billing Information Service is the provision of information to the the Commissi from Telephone Company record systems labeled as Customer Records Information System (CRIS) and Customer Name and Address Bureau (CNA).

8.4.1 Rates and Charges

(AT)

Rates and charges for Billing Information Service for all contract lengths up through five (5) years are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

(AT) (AT)		l Year More Than Rate <u>1 Yr. Rate</u>
	Program Development Charge Per Hour	\$60.0000 (NR) IÇB
	Data Transmission Sent or Received	INTILED
	Per Record Using CMDS Network	0.0030 CANCELLED
	Data Transmission Sent or Received	0
	Per Record Using Customer Provided	
	Network	0.0010 IN 201985
	Clerical Staff Utilization Per Hour	45.0000 JAN 5 # 4
	CPU Utilization Per Hour	850.0000 HTAR.S
	Standard SWBT CRIS Billing Information	BY 4 Commission
	Per Record	0.0006 Service UBI
	CNA Information Per Request	0.0000 1/2 1.5 0.000 BY 1/2 1.5 1.5000 MISSOURI
	CNA Info Written Confirmation Per Request	0.7500
	Overnight Delivery of Mag Tape	90.0000
	Billing Name and Address	
	- mechanized, per request	0.3000
	- mechanical, per request	1.0000 (NR) ICB

8.5 Account Maintenance

Account Maintenance is the provision of customer information to update the IC's data base, for example, billing name and address changes, number changes, etc.

8.5.1 Rates and Charges

Rates and charges will be determined on an Individual Case Basis (ICB).

8.6 Liability of the Telephone Company

If detail∈d information re is not available because t tapes or incurred system o	he Telephone Company lost	or damaged records or
Issued: MAY 1 1 1994 By M. H. SCHULTEIS, Divisi Southwest	Effective: JUN ion Manager-Regulatory & In ern Bell Telephone Company	ndustry Relations 4 1994
	t. Louis, Missouri	MISSOURI Public Service Commission

				P.S.C. MoNo. 36			
t	ariff	lement to i vill be is:	sued		Ac		Services Tariff Section 8 Revised Sheet 4
		for the pur ling this		ACCESS SERVICES	Replacing		Revised Sheet 4
8	. BILI	LING AND C	OLLECTION SI	ERVICES-(Continued)			RECEIVED
	8.4 Bi	llling Infe	ormation Se	rvice			JAN 25 1989
	fı	rom Teleph	one Company	rvice is the provisi record systems labe IS) and Customer Nam	led as $Custo$	Hehd	Records Commission
	8.4.1	Rates and	d Charges				
۲		lengths	up through a bourly	or Billing Informati five (5) years are J y basis, the charges	isted below. are doubled	Fo	r rate elements performed on an
		Drognon		Charge Per Hour		.000	
		Data Trai Per Rec	nsmission S ord Using C	ent or Received MDS Network	-	.003	
		Per Rec Netvork	ord Using C	ent or Received ustomer Provided		.001	
		CPU Util	ization Per	ization Per Hour Hour Billing Information		000 000 000	
		CNA Info Overnigh	rmation Per Written Co t Delivery	nfirmation Per Reque of Mag Tape	est O	.000 .500 .750 .000	BY 3 NAR. S. # 4 Bublic Service Commission
(NR) (NR)				dress ized, per request ical, per request		.300 .000	
	8.5 A	ccount Mai	ntenance				
	I		ase, for ex	the provision of cu ample, billing name			-
	8.5.1	Rates an	d Charges				
		Rates an	d charges w	ill be determined o	n an Individu	ual (Case Basis (ICB).
	8.6 L	iability o	f the Telep	hone Company			
۲	1	s not avai	lable becau	on required to provi se the Telephone Co em outages, the Tel	mpany lost o	r da	aged records or
		JAN 27	1020	Rffective:	FEB 26 198		FILED
- 1			•	ARRON, President-Mi	-		FEB 26 1989
			•	-	one Company		lic Service Commission

-

-1

8.

Access Services Tariff Section 8 1st Revised Sheet 4 Replacing Original Sheet 4

Rate

(CP) ACCESS SERVICES

NOV 241987

MISSOURI

RECEIVED

BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service

Billing Information Service is the provision of information to the IC Commission from Telephone Company record systems labeled as Customer Records Information System (CRIS) and Customer Name and Address Bureau (CNA).

8.4.1 Rates and Charges

Rates and charges for Billing Information Service for all contract lengths up through five (5) years are listed below. For rate elements charged on an hourly basis, the charges are doubled if performed on an overtime basis.

•	Program Development Charge Per Hour Data Transmission Sent or Received Per Record Using CMDS Network Data Transmission Sent or Received Per Record Using Customer Provided Network Clerical Staff Utilization Per Hour CPU Utilization Per Hour Standard SWBT CRIS Billing Information	(CR) (NR) (NR) (NR)	\$60.0000 0.0030 0.0010 45.0000 850.0000	
	Per Record CNA Information Per Request	(CR)	0.0006 1.5000	CANCELLED
	CNA Info Written Confirmation Per Request Overnight Delivery of Mag Tape	(CR) (NR)	.0.7500 90.0000	FEB 26 1989

8.5 Account Maintenance

Public Service Commission

#4

Account Maintenance is the provision of customer information to MISBOURIE IC's data base, for example, billing name and address changes, number changes, etc.

8.5.1 Rates and Charges

Rates and charges will be determined on an Individual Case Basis (ICB).

8.6 Liability of the Telephone Company

If detailed information required to provide Billing and Collection Service is not available because the Telephone Company lost or damaged records or tapes or incurred system outages, the Telephone Company will attempt to

		PILED
Issued: DEC 1		JAN 1 1988
	By R. D. BARRON, President-Missouri Div Southwestern Bell Telephone Company St. Louis, Missouri	isionPublic Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 4

DEC 20 CLJ

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

- 8.1.3 Liability of the Telephone Company-(Continued)
 - A. If IC message detail is not available because the Telephone Company lost or damaged tapes or incurred recording system outages, the Telephone Company will estimate the volume of lost IC messages and associated revenue based on previously known values. This estimated IC message volume will be included along with the IC message detail provided to the IC and/or provided for Message Processing Service. In such events, the extent of the Telephone Company's liability for damages shall be limited to the granting of a corresponding credit adjustment to the IC amounts due to account for the unbillable revenue.
 - B. When the Telephone Company is notified that, due to error or omission, incomplete data have been provided to an IC, the Telephone Company will make a reasonable effort to locate and/or recover the data and provide new magnetic tapes to the IC at no additional charge. Such request to recover the data must be made within 30 days from the date the details were initially made available to the IC. If the data cannot be recovered, the extent of the Telephone Company's liability for damages shall be limited as set forth in Paragraph 8.1.3, A., preceding.
 - C. In the absence of willful misconduct, no liability for damages to the IC or other person or entity other than that as set forth in Paragraphs 8.1.3, A. and B., preceding, shall attach to the Telephone Company for its action or the conduct of its employees in providing Recording Service.

8.1.4 Obligations of the IC

A. The IC shall order Recording Service under a Special Order for each state where the service is desired. By let kS. #4

The IC shall order Recording Service at least one month prior to the date when the IC message detail is to be recorded MISSOURI

B. The IC shall order provision of recorded IC message detail at least one month prior to the date when it wishes to receive the recorded message detail. However, the IC may wish to receive magnetic tapes of the recorded detail or have the recorded detail data-transmitted to an IC location at some times and not at others. Therefore, change in the provision of recorded IC message detail to the IC will be accommodated provided the IC gives two weeks' advance written notific**et gon 20519** Telephone Company.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

CANCELLED

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.6 Liability of the Telephone Company-(Continued)

recover the lost IC detail or will estimate the lost volume of messages. In such events, the extent of the Telephone Company's liability for damages shall be limited as provided in the contract between the Telephone Company and the IC.

When the Telephone Company is notified that, due to its error or omission, incomplete detail has been provided to an IC, the Telephone Company will make every reasonable effort to recover the detail at no additional charge to the IC. Such requests to recover the detail must be made within 30 days from the date the details were initially made available to the IC. If the detail cannot be recovered, the extent of the Telephone Company's liability for damages shall be limited as provided in the contract.

In the absence of willful misconduct, no liability for damages to the IC or other person or entity other than that set forth in the preceding paragraphs, shall attach to the Telephone Company for its actions or the conduct of its employees in providing Billing and Collection Services.

Any additional provisions limiting the liability of the Telephone Company shall be provided for in the contract.

8.7 General Regulations

Each Billing and Collection Services contract signed by an IC will specify the specific Rates and Regulations, as well as appropriate methods, practices and procedures that will apply to the IC. Major changes to the service arrangement made by the customer will require review by the Telephone Company and may result in changes of rates or provisions in the contract and may require regulatory approval.

At the conclusion of the contract, the customer must elect one of the following options:

- Renew the contract at the terms, conditions and rates that are tariffed or otherwise available at the time.
- Discontinue the service.
- (AT) When the Company furnishes a facility or service for which a rate or charge is not specified in the Company's tariffs, charges will be determined on an Individual Case Basis (ICB). Such charges will be made available to similarly situated customers on a nondiscriminatory basis. ICB
 Rates will be structured to recover the Company's cost of providing the services. Terms of specific ICB contracts will be made available to the Missouri
 (AT) Public Service Commission upon request on a proprietary basis.

Issued:	May 11, 1994	Effective:	July 14, 1994	
	By M. H. SCHULTEIS	, Division Manager-Regu	ulatory & Industry Relations	F
	Southweste	ern Bell Telephone Comp	bany	M
		St. Louis, Missouri		

Access Services Tariff Section 8 1st Revised Sheet 5 Replacing Original Sheet 5

(CP) ACCESS SERVICES

RECEIVED

8. BILLING AND COLLECTION SERVICES-(Continued)

NOV 2 4 1987

8.6 Liability of the Telephone Company-(Continued)

MISSOURI

recover the lost IC detail or will estimate the lost volume Service Commission In such events, the extent of the Telephone Company's liability for damages shall be limited as provided in the contract between the Telephone Company and the IC.

When the Telephone Company is notified that, due to its error or omission, incomplete detail has been provided to an IC, the Telephone Company will make every reasonable effort to recover the detail at no additional charge to the IC. Such requests to recover the detail must be made within 30 days from the date the details were initially made available to the IC. If the detail cannot be recovered, the extent of the Telephone Company's liability for damages shall be limited as provided in the contract.

In the absence of willful misconduct, no liability for damages to the IC or other person or entity other than that set forth in the preceding paragraphs, shall attach to the Telephone Company for its actions or the conduct of its employees in providing Billing and Collection Services.

Any additional provisions limiting the liability of the Telephone Company shall be provided for in the contract.

8.7 General Regulations

Each Billing and Collection Services contract signed by an IC will specify the specific Rates and Regulations, as well as appropriate methods, practices and procedures that will apply to the IC. Major changes to the service arrangement made by the customer will require review by the Telephone Company and may result in changes of rates or provisions in the contract and may require regulatory approval.

JUL 141994

MISSOU

JAN 1

At the conclusion of the contract, the customer must elect one of the following options:

- Renew the contract at the terms, conditions and rates that are tariffed or otherwise available at the tipe.

- Discontinue the service.

FILED

JAN 1 1988 87-95

Service Commission Public Service Commission

1988

Issued: DEC 1 1987

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective:

Access Services Tariff Section 8 Original Sheet 5

DEC 20

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

- 8.1.4 Obligations of the IC-(Continued)
- C. The IC's terminal location shall provide such signals as may be required for the proper operation of the Telephone Company's automatic message accounting equipment used to perform the detail recordings.

8.1.5 Payment Arrangements and Audit Provision

A. Audit Provision

Upon reasonable written notice by the IC to the Telephone Company, the IC shall have the right through its authorized representative to examine and audit, during normal business hours and at reasonable intervals as determined by the Telephone Company, all such records and accounts as may under recognized accounting practices contain information bearing upon the recording of messages for which amounts may be payable to the IC. Adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit. Neither such right to examine and audit nor the right to receive such adjustment shall be affected by any statement to the contrary, appearing on checks or otherwise, unless such statement expressly waiving such right appears in a letter signed by the authorized representative of the party having such right and delivered to the other party.

All information received or reviewed by the IC or its authorized representative is to be considered confidential and is not to be distributed, provided or disclosed in any form to anyone not involved in the audit, nor is such information to be used for any other purpose.

CANCELLED

JAN 1 1998

BY <u>Lat R.S. #5</u> Public Service Commission MISSOURI

JANT 1 (SO) 83 - 253

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 1st Revised Sheet 6 Replacing Original Sheet 6 1st Revised Sheet 8 Original Sheet 9 Original Sheet 10 Original Sheet 11 Original Sheet 12 Original Sheet 13 Original Sheet 14 Original Sheet 15 1st Revised Sheet 16 Original Sheet 17 Original Sheet 18 1st Revised Sheet 19 1st Revised Sheet 20 Original Sheet 20.01 Original Sheet 20.02 Original Sheet 21 1st Revised Sheet 22 Original Sheet 23 Original Sheet 24 Original Sheet 25 Original Sheet 26 Original Sheet 27 1st Revised Sheet 28 1st Revised Sheet 29 1st Revised Sheet 30 1st Revised Sheet 31 1st Revised Sheet 32 1st Revised Sheet 33 1st Revised Sheet 34 1st Revised Sheet 35 Original Sheet 36 1st Revised Sheet 37 Original Sheet 37.01 1st Revised Sheet 38 Original Sheet 39 Original Sheet 40 2nd Revised Sheet 41 Original Sheet 41.01 Original Sheet 42 Original Sheet 43 Original Sheet 44 Original Sheet 45 Original Sheet 46 Original Sheet 47 Original Sheet 48 Original Sheet 49 Original Sheet 50 Original Sheet 51



Issued: December 1, 1987 Effective: January 1, 1988 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 6

DEC 20100

83-253

 (\mathbf{r}, \mathbf{r})

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

8.1.5 Payment Arrangements and Audit Provision-(Continued)

B. Minimum Period and Minimum Monthly Charge

The minimum period for which Recording Service is provided and for which charges apply is one month for each state in which the service is ordered.

The minimum monthly charges are the charges for IC messages recorded, IC message assembled and edited (except when Message Processing Service is ordered for the same monthly period) and when ordered by the IC, provision of the IC message detail on magnetic tape or data file. If the service is cancelled or discontinued prior to entering the detail on a magnetic tape or data file, the minimum monthly charge will be the charge for all IC messages recorded, assembled and edited for a 30-day period. The Telephone Company will use the most recent 30-day period for which data is available to determine the minimum charge.

C. Cancellation of a Special Order

An IC may cancel a Special Order for Recording Service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for Recording Service is the date the IC requests the recordings to start.

When an IC cancels a Special Order for Recording Service after the order date but prior to the start of service, the minimum monthly charges will apply.

BY Lat. R.S. #6 Fublic Service Commission

D. Changes to Special Orders

MISSOURI

When an IC requests changes to a pending Special Order for Recording Service, the pending Special Order will be cancelled, and the requested changes will will be undertaken if they can be accommodated by the Telephone Company under a new Special Order. All cancellation charges as set forth in Paragraph 8.1.5, C., preceding, will apply for the cancelled Special Order.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 2nd Revised Sheet 7 Replacing 1st Revised Sheet 7 Original Sheet 52 Original Sheet 53 Original Sheet 54 Original Sheet 55 Original Sheet 56 1st Revised Sheet 57 1st Revised Sheet 58 1st Revised Sheet 59 1st Revised Sheet 60 1st Revised Sheet 61 1st Revised Sheet 62 Original Sheet 62.01 Original Sheet 62.02 Original Sheet 62.03 Original Sheet 62.04 Original Sheet 62.05 Original Sheet 62.06 Original Sheet 62.07 Original Sheet 63 Original Sheet 64 Original Sheet 65 Original Sheet 66 Original Sheet 67 Original Sheet 68 Original Sheet 69 Original Sheet 70 Original Sheet 71 **Original Sheet 72** 1st Revised Sheet 73 2nd Revised Sheet 74 1st Revised Sheet 75 2nd Revised Sheet 76 2nd Revised Sheet 77 2nd Revised Sheet 78 2nd Revised Sheet 79 1st Revised Sheet 80 2nd Revised Sheet 81 1st Revised Sheet 82 1st Revised Sheet 83 Original Sheet 84 1st Revised Sheet 85 1st Revised Sheet 86 Original Sheet 87

Issued:

December 1, 1987

Effective: January 1, 1988



No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

Access Services Tariff

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.1 Recording Service-(Continued)
- 8.1.6 Rate Regulations

- REGEIVED JUN 27 1986
- A. For each message recorded, the recording and the assembling and editing charges apply except when the IC orders Message Processing Service as set forth in Paragraph-8-2-1, following, is ordered for the same state and month that Recording Service is ordered, the assembling and editing charge does not apply per IC message.

The charges for recording and for assembly and editing apply per message recorded and per message assembled and edited whether or not the IC's schedule of rates specifies billing on a per-message basis or any other basis.

- B. The per-Special-Order Charge applies for each Special Order accepted by the Telephone Company for Recording Service.
- C. When message detail, with or without sorting and/or name and address information is entered on a data file or magnetic tape for provision of message detail to an IC, the per-tape charge applies for each data file or magnetic tape prepared, and the per-record charge applies for each record processed. A record is a logical grouping of information as described in the programs that process the information and load the magnetic tapes or data file. The Telephone Company will determine the charges based on the number of data files or magnetic tapes prepared and on its count of the records processed. The number of records input to or the number of records output from the programs that process information and load the magnetic tapes or data file, whichever number of records is higher.
- 8.1.7 Rates and Charges

CANCELLED

The rates and charges are: JAN 1 1988

per IC message	BY <u>And l.s. #19</u> US blic Service Commission 	ARR+ (CR)	120 10 1 1986
Assembling and Editing, per IC message(1)	BA	ARA+	86-84 Public Soggice Commission
 Not applicable when Message P 8.2.1, following, is provided 8.2.1, B.1.c. and B.2.d., following 	to the IC, except as se		
Issued: JUN 2 7 1986	Effective: JUL	1 1986	

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 7

ACCESS SERVICES

DEC 2 . Los

8. BILLING AND COLLECTION SERVICES-(Continued)

8.1 Recording Service-(Continued)

- 8.1.6 Rate Regulations
- A. For each message recorded, the recording and the assembling and editing charges apply except when the IC orders Message Processing Service. When Message Processing Service as set forth in Paragraph 8.2.1, following, is ordered for the same state and month that Recording Service is ordered, the assembling and editing charge does not apply per IC message.

The charges for recording and for assembly and editing apply per message recorded and per message assembled and edited whether or not the IC's schedule of rates specifies billing on a per-message basis or any other basis.

- B. The per-Special-Order Charge applies for each Special Order accepted by the Telephone Company for Recording Service.
- C. When message detail is entered on a data file or magnetic tape for provision of message detail to an IC, the per-tape charge applies for each data file or magnetic tape prepared, and the per-record charge applies for each record processed. A record is a logical grouping of information as described in the programs that process the information and load the magnetic tapes or data file. The Telephone Company will determine the charges based on the number of data files or magnetic tapes prepared and on its count of the records processed. The number of records processed will be determined using the number of records input to or the number of records output from the programs that process information and load the magnetic tapes or data file, whichever number of records is higher.

8.1.7	Rates and Charges 🖁	ANGELLE		. -
	The rates and charges	are:UL 1 1986		
		10 c # 7 -	USOC	Rates
	Recording, Pl per IC message per Special Order		BARR+ BARR+	8130702753
	Assembling and Editing per IC message(1)		BARA+	.0035
8.2.		Processing Service as se d to the IC, except as s llowing.		
Issued:	DEC 2 9 1983	Effective: JAN	0 1 1984	

i

ļ

· · -----

	tariff except	plement to the will be issue for the purp- nceling this t	ed ose	ACCESS SERVICES	, R	1st	Services Tariff Section 8 Revised Sheet 8 Original Sheet 8
	8. BI	LLING AND COL	LECTION SERV	ICES-(Continued))		REGEIVED
	8.1	Recording Ser	vice-(Contin	ued)			APR 10 100
	8.1.	7 Rates and	Charges-(Con	tinued)			
		The rates	and charges	are:-(Continued))	ETD	MISSOURI Public Service Commission
		per recor		• • • • • • • •	• • •	PRMD	\$.0035
•		Data Trans per recor	nission to a d transmitte	n IC Location, d	NCELL	ED BOD :	DT ICB rates and charges
(NR)		Other tran per tape	smission to or data file	a customer locat	ipal 1	Commission Commission SOURI	ICB rates and charges
	8.2	Billing Servi		autr	BY Service	SOURI	apply.
				the Telephone Co ovide Billing So		th reason	able notice and
		tory. The mi Billing Servi supplies the input records	nimum territ ce is its st input record , the Teleph	ory for which the	ne Teleph erritory equest. L process	one Compa when the When the the inpu	Telephone Company IC supplies the t records sup-
		(1) Message B	illing Servi	ll provide two t ice as set forth set forth in Par	in Parag	raph 8.2.	1 and (2) Private
		Billing Servi	ce only on t	ll provide Bill : the condition th C as set forth i	at it pur	chase the	accounts receiv-
			delivery of	E telegrams, flo			iff for the pro- or other like
	8.2	.1 Message Bi	lling Servio	ce			
		Processing	Service. A				Service and Bill Service or Bill
		···					
-	Issue	0 1085				Y 20 198	MAY 2 0 1985
	Ark I	9 (985		RON, President- stern Bell Telep St. Louis, Miss	hone Comp		Public Service Commission

No supplement to this Access Services Tariff tariff will be issued Section 8 except for the purpose Original Sheet 8 of canceling this tariff. ACCESS SERVICES UEU 2 BILLING AND COLLECTION SERVICES-(Continued) 8.1 Recording Service-(Continued) 8.1.7 Rates and Charges-(Continued) The rates and charges are:-(Continued) FID Rates Provision of Message Detail, PRMD NOELLED per record processed \$.0035 per tape or date file. . . 58.00 MAY. 2.0.1985 Data Transmission to an IC Location, BOD DT per record transmitted ICB rates and charges PUBLIC SERVICE COMMISSION apply 8.2 Billing Service

At the request of an IC, the Telephone Company with reasonable notice and reasonable effort will provide Billing Service.

The Telephone Company will provide Billing Service in its operating territory. The minimum territory for which the Telephone Company will provide Billing Service is its state operating territory when the Telephone Company supplies the input records at the IC's request. When the IC supplies the input records, the Telephone Company will process the input records supplied by the IC as set forth in Paragraphs 8.2.1 and 8.2.2, following.

The Telephone Company will provide two types of Billing Service: (1) Message Billing Service as set forth in Paragraph 8.2.1 and (2) Private Line Billing Service as set forth in Paragraph 8.2.2, following.

The Telephone Company will provide Bill Processing Service, Private Line Billing Service only on the condition that it purchase the accounts receivable, if any, from the IC as set forth in Paragraph 8.2.3, following.

The Telephone Company will not render bills under this Tariff for the provision and/or delivery of telegrams, flowers, gifts, wine or other like JANTI 1 1934 83-253 services that an IC offers.

8.2.1 Message Billing Service

Message Billing Service consists of Message Processing Service and Bill Processing Service. An IC may order Message Processing Service or Bill Processing Service or both services.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - A. General Description
 - 1. Message Processing Service

Message Processing Service is the transforming of the recorded IC call details into rated messages in preparation for billing. Message Processing Service includes initial data entry and rating of messages.

Initial data entry is the assembly of recorded IC call details into IC messages. This function includes editing and verification of recorded details to assure that the data required for rating are present.

Rating of IC messages is the computing of applicable charges for each IC message based on the IC-provided schedule of.rates. Rating also includes the preparation of IC message detail for input to Bill Processing Service, the IC or other entities.

Further, rating is always performed and editing may be performed coincident with the implementation of a change in the IC's schedule of rates.

2. Bill Processing Service

DEC 2 9 1983

Bill Processing Service is the preparation of bills for messagebilled service and bulk-billed service, mailing of statements of the amounts due for service received from the IC and the collection of the deposits and monies due from the End Users. Bill Processing Service includes message-billed (when necessary) and bulk-billed account establishment, posting of rated messages and rate elements, rendering of bills, collection of deposits, receiving payments, maintenance of accounts, treatment of accounts, $|P_{i}|$ message investigation and inquiry (when ordered by the IC).



JAN 1 1988 BY 15+ R56

Issued:

Public Service Commission MISSQUELVe:

JAN 0 1 1994

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 9

DEC 20 12

J. . . **1** (92)

83 - 253

Access Services Tariff Section 8 Original Sheet 10

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- A. General Description-(Continued)
 - 2. Bill Processing Service-(Continued)

Bulk-billed service is a billing service for an End User's account with a Dedicated Access Line (DAL) Service to the End User's premises where individual IC messages are not posted to the account and are not listed on the bill rendered to the End User.

Message-billed service is a billing service for an End User's account with an End User's common line where individual IC messages are posted to the account and are listed on the bill rendered to the End User. Message-billed service is also a billing service for an IC credit card End User's account without an End User's common line or Dedicated Access Line Service where individual messages or groups of messages are posted to the account and listed on the bill rendered to the End User.

Account establishment is the preparation of an IC End User's record so that a bill can be sent to that End User.

Posting of rated messages is the processing for billing of rated messages. Posting also is the examination and identification of all the rateable elements specified by the IC to be billed to an End User. Application of appropriate IC rates and charges to all such rate elements is also included when requested by the IC. The rating may be performed by the Telephone Company, another entity or the IC. Editing and rating of rate elements is performed when IC services are established or changed. Rating is always performed and editing may be performed coincident with the implementation of a change in the IC's schedule of rates.

CANCELLED

JAN 1 1988 BY 167 RS6 Public Service Commission MISSOURI



Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

DEC 25 (11)

Falle Content of the Store

UNT1601 83-253

Access Services Tariff Section 8 Original Sheet 11

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

DEC 27 ()

J. . 1 T 1 198 1

83-253

8.2 Billing Service-(Continued)

- 8.2.1 Message Billing Service-(Continued)
- A. General Description-(Continued)
 - 2. Bill Processing Service-(Continued)

Rendering of bills is the preparation and mailing of statements of the deposits and amounts due from the End User for IC message-billed and bulk-billed services. These statements may, at the Telephone Company's choice, be included as part of the regular monthly bill for Local Telephone Exchange Service mailed to the End User.

Receiving payment and maintenance of accounts is the collecting of monies from End Users for services furnished by the IC and maintenance of records of all transactions.

Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting of credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's services and/or Local Telephone Exchange Services to an End User. Where Local Telephone Exchange Service access is denied, access to IC service will also be denied.

Message investigation is that activity undertaken by the Telephone Company to secure, or attempt to secure, proper billing information for IC messages.

Inquiry is the answering of End User's questions about charges billed for IC services and application of credits and adjustments to End User's accounts and review of IC messages removed from an End User's bill.

CANCELLED

JAN 1 1988 BY <u>JSA RS6</u> Public Service Commission MISSOURI

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1934

Access Services Tariff Section 8 Original Sheet 12

article to sea a

DEC 20112

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - Β. Undertaking of the Telephone Company
 - 1. Message Processing Service
 - When Message Processing is ordered by an IC, the Telephone Company a. will process all of the IC messages it possesses in a state as set forth in Paragraph 8.2.1, B.1.b.through 1., following, at rates and charges set forth in Paragraph 8.2.1, G., following.
 - b. The Telephone Company will provide Message Processing Service only for IC messages originating or recorded within the operating territory of the Telephone Company. The IC messages which the Telephone Company will process may be IC messages from Recording Service as set forth in Paragraph 8.1, preceding, or at the direction of the IC, other IC messages which are chargeable in accordance with the rate schedule furnished by the IC. Any sent-paid coin IC messages provided as input by the IC will be processed unless the IC specifies in writing that such IC messages are not to be processed. When such IC messages are processed, message processing charges will When such messages are not processed, they will not be apply. included in any IC message detail provided to the IC.
 - c. A record of IC call details is required to provide Message Processing Service. Where an IC subscribes to Recording Service as set forth in Paragraph 8.1, preceding, those recorded details may be used as the input. Where the IC provides the call details, the records must be in the standard format established by the Telephone Company and delivered to the location specified by the Telephone Company. The charges as set forth in Paragraph 8.2.1, G.12., following, will apply if the IC data transmits its call details to the Telephone Company. If the IC-provided records must be converted by the Telephone Company to the standard format, and the Telephone Company agrees to make the conversion, program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary programs. The assembling and editing charges as set forth in Paragraph 8.1.7, preceding, applies in addition to all other charges for all such details converted by the Telephone Company. The CANCELLED



Issued: DEC 2 9 1983

BY ETERECTIVE: BY R. D. BUDWONSERVICE COmmissionJAN 0 1 1984 Southwestern MISSelephone Company St. Louis, Missouri
Access Services Tariff Section 8 Original Sheet 13

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

DEC 25 (TT)

- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 1. Message Processing Service-(Continued)
 - c. (Continued)

Telephone Company will provide to the IC the precise details of the required standard format. If, in the course of Telephone Company business, it is necessary to change the standard format, the Telephone Company will provide notification to the involved IC's six months prior to the change. If the IC requests the IC-provided call details be reprocessed by the Telephone Company because of an IC error, the Telephone Company will reprocess the IC-provided call details and the appropriate charges as set forth in Paragraph 8.2.1, G., following, will apply.

- d. The Telephone Company will develop the IC's schedule of rates into a rating program. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary programs.
- e. Upon acceptance by the Telephone Company of a Special order for Message Processing Service from an IC, the Telephone Company will determine the period of time to implement such service on an individual-order basis.
- f. Changes in the rate levels of IC charges to be billed will normally be implemented within 30 days after receipt of a Special Order from the IC requesting such changes. Such changes will require modifications of the rating program. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary program changes. If any IC message detail must be reprocessed in order to apply the rate changes, the appropriate message processing charges as set forth in Paragraphs 8.2.1, G.1 and 2., following, apply for

CANCELLED all IC messages reprocessed. UNITISS 83-253 JAN 1 1988 BY 157 AS 6 Public Service Commission MISSOURI JAN 0 1 1984 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Issued:

d: DEC 2 9 1983

Access Services Tariff Section 8 Original Sheet 14

DEC 27 1

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - Undertaking of the Telephone Company-(Continued) Β.
 - 1. Message Processing Service-(Continued)
 - Changes in the rate structure for IC services to be billed also g٠ require a change in the rating program. When the Telephone Company determines that it can accommodate the changes, the conditions and the period of time required to make such changes will be determined on an individual-order basis. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary program changes. If any IC message detail must be reprocessed in order to apply the rate structure changes, the appropriate message processing charges as set forth in Paragraphs 8.2.1, G.1. and 2., following, apply for all IC messages reprocessed.
 - **h**. Where the Telephone Company has rated IC messages which are to be billed to an End User by another exchange telephone company, the Telephone Company will enter the IC messages on a magnetic tape or data file which can be used for data transmission of the details. When the IC has so arranged with an involved exchange telephone company, the Telephone Company will transmit the rated IC message details to such other exchange telephone companies for billing to End Users in their operating territories. When the IC does not have billing arrangements with an exchange telephone company, rated IC messages for such an exchange telephone company will be delivered to the IC. The charges as set forth in Paragraph 8.2.1, G.4., following, apply to rated IC messages that are data-transmitted to the other exchange telephone companies. The charges as set forth in Paragraph 8.2.1, G.5., following, apply to rated IC messages that are delivered to the IC. When the IC message details are datatransmitted to the IC location, the data transmission charges will be determined on an individual-case basis. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary programs.

CANCELLE JAN 1 1988 BY 157 PS6 JAM T 1 1981 à 83-253 Public Service Commission

JAN 0 1 1984

Issued:

DEC 2 9 1983

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

B. Undertaking of the Telephone Company-(Continued)

- 1. Message Processing Service-(Continued)
 - i. Where the rates for IC services have been implemented under an accounting order pending final approval from a regulatory agency, the Telephone Company will, upon written request from the IC, keep such records as may be required to make any adjustments to the End Users as may be ordered by the regulatory agency. The charges for such a service will be determined on an individual-case basis.
 - j. The Telephone Company will, upon request, provide the IC the rated IC message detail.

The rated IC message detail will be provided on a request-byrequest basis in a format similar to that used by the Telephone Company as input to Bill Processing Service. All rated IC message detail available to the Telephone Company will be provided to the IC. The rated IC message detail will not be sorted to furnish detail by specific End Users, groups of End Users, by office or by location.

The Telephone Company will provide the IC detail on a magnetic tape. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary programs. When a magnetic tape is provided, the charges as set forth in Paragraph 8.2.1, G.5., following, will apply.



Access Services Tariff

DEU 211

Original Sheet 15

Section 8

Issued:

1: DEC 2 9 1983

Effective: JAN 0 1 1284

Access Services Tariff Section 8 Ist Revised Sheet 16 Replacing Original Sheet 16

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- B. Undertaking of the Telephone Company-(Continued)
 - 1. Message Processing Service-(Continued)
 - j. (Continued)

Unless specified otherwise by the IC, the magnetic tapes will be sent to the IC using first-class U.S. Mail Service. However, the IC may pick up the magnetic tapes at a location designated by the Telephone Company or request the information on the magnetic tapes be data-transmitted to the IC. When the information is datatransmitted to an IC location, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis.

- k. If the IC makes a request within 30 days of the date the IC details were initially made available to the IC, the Telephone Company will make every reasonable effort to recover the IC detail and make it available again to the IC and the charges as set forth in Paragraph 8.2.1G.5., will apply for all such IC detail provided. When the IC details are data-transmitted to an IC location, the data transmission charges will be determined on an individual-case basis.
- 1. IC messages which the Telephone Company processes that cannot be rated in accordance with the IC rate schedule will be reviewed by Telephone Company message investigation groups. Upon completion of the review, rated IC messages will be delivered to the IC when the IC orders such service or to Bill Processing Service when the IC orders such service. Unrated messages will be handled in accordance with instructions that have been mutually determined by the Telephone Company and the IC. At the request of the IC, the unrated IC messages will be reviewed for unauthorized use of the IC service by Telephone Company message investigation groups for a period of up to 90 days after the IC message was processed. The appropriate charges as set forth in Paragraph 8.2.1G.5., following, or Paragraphs 8.2.1G.7. and 8., following, will apply.

	CANCELL 1988 - (0	
······	INT SC	导航危险
Issued: APR 19 1985	BY R. D. BABRONIC Preased Hissouri Division Southwestern Bell Telephone Company	MAY 201985
	Southwestern Bell Telephone Company St. Louis, Missouri	Public Service Commissio



(AT)

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

8.2 Billing Service-(Continued)

- 8.2.1 Message Billing Service-(Continued)
- В. Undertaking of the Telephone Company-(Continued)
 - 1. Message Processing Service-(Continued)

(Continued) i.

OF MISSOURI Unless specified otherwise by the IC, the magnetic tapes will be sent to the IC using first-class U.S. Mail Service. However, the IC may pick up the magnetic tapes at a location designated by the Telephone Company or request the information on the magnetic tapes be data-transmitted to the IC. When the information is datatransmitted to an IC location, the data transmission charges will be determined on an individual-case basis.

Access Services Tariff

GANCELLED

MAY 2 0 1985

PUBLIC SERVICE COMMISSION

Original Sheet 16

Section 8

DEC 20 COU

U/ANTT1 (SE) 83-253

- ĸ. If the IC makes a request within 30 days of the date the IC details were initially made available to the IC, the Telephone Company will make every reasonable effort to recover the IC detail and make it available again to the IC and the charges as set forth in Paragraph 8.2.1, G.5., will apply for all such IC detail provided. When the IC details are data-transmitted to an IC location, the data transmission charges will be determined on an individual-case basis.
- 1. IC messages which the Telephone Company processes that cannot be rated in accordance with the IC rate schedule will be reviewed by Telephone Company message investigation groups. Upon completion of the review, rated IC messages will be delivered to the IC when the IC orders such service or to Bill Processing Service when the IC orders such service. Unrated messages will be handled in accordance with instructions that have been mutually determined by the Telephone Company and the IC. At the request of the IC, the unrated IC messages will be reviewed for unauthorized use of the IC service by Telephone Company message investigation groups for a period of up to 90 days after the IC message was processed. The appropriate charges as set forth in Paragraph 8.2.1, G.5., following, or Paragraphs 8.2.1, G.7. and 8., following, will apply. 尼山居り

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 Original Sheet 17

DEC 20 (000

10.000 PTR SERVER

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service
 - a. When Bill Processing Service is ordered by an IC, the Telephone Company will establish and maintain End User's accounts and prepare and render bills for all IC messages, bulk-billed messages and related rate elements it possesses for a state as set forth in Paragraphs 8.2.1, B.2.b. through m., following, at rates and charges as set forth in Paragraph 8.2.1, G., following. The Telephone Company will not establish an End User's account with any IC balance due.

In addition, the Telephone Company will, in accordance with Telephone Company deposit regulations, determine and collect a deposit from the End User for the IC service. The Telephone Company will, when necessary, in accordance with the Telephone Company deposit regulations, determine and collect the service deposit for new or established End User's accounts or for established accounts when the first message is posted to the End User's account. The Telephone Company will, when necessary, in accordance with the Telephone Company deposit regulations, maintain a service deposit balance for each End User's account. Service Deposits will not be maintained by individual IC accounts, but will be maintained for the End User's account in general. The Telephone Company will provide the IC a copy of its deposit regulations, maintain (account) for means the IC.

- b. The Telephone Company will provide Bill Processing Service for Mission message-billed service, bulk-billed service and related of the plane elements which are posted to End User's accounts de Sterier Sthin the operating territory of the Telephone Company only. The Telephone Company will separate the rated IC messages into a message-billed group and a bulk-billed group for application of frates as set forth in Paragraph 8.2.1, G., following: 83-253
- c. At the request of the IC, the Telephone Company will prepare and distribute IC credit cards by first-class U.S. Mail Service. The Telephone Company will assign the credit card number and will mark its records and files to show that an End User has been issued an IC credit card. The Telephone Company will specify the information it requires to issue a credit card and the format to be used by the

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 18

DEC 2 0 1000

practic Contract and Mostica

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

B. Undertaking of the Telephone Company-(Continued)

2. Bill Processing Service-(Continued)

c. (Continued)

IC in furnishing such information. The charges as set forth in Paragraph 8.2.1, G.13., following, apply. Plastic-coated paper cards will be distributed unless the IC requests another type of card be provided. Charges to prepare and distribute other such cards will be developed on an individual-case basis. When it becomes necessary, as determined by the Telephone Company, to change the credit-card number or to discontinue the billing of credit-card calls to an End User's account because of nonpayment of charges or unauthorized use of Telephone Company and IC service offerings, the Telephone Company will notify the IC. The Telephone Company will provide the IC the credit-card number, associated End User's account name and billing address for the credit-card number change or billing discontinued. All charges for calls associated with such a discontinued after the IC has been notified will become the Tesponsibility of the IC. End User's questions concerning the issuing of IC credit cards will not be handled by the Telephone Company.

d. Rated IC messages are required to provide Bill Processing Vervice. If the IC subscribes to Message Processing Service as set forth in Paragraph 8.2.1, B.1., preceding, the rated IC destances may be used as the input. If the IC provides the rated IC destances may be used as the input. If the IC provides the rated IC destances, those IC messages must be in the standard format established by the Telephone Company and delivered to the location specified by the Telephone Company. The charges as set forth in Paragraph 8.2.1, G.12., following, apply if the IC data transmits its rated message data to the Telephone Company. Such IC-provided rated message data must identify the End User's account to be billed. If the IC-provided rated messages must be converted by the Telephone Company to the standard format and the Telephone Company agrees to make the conversion, program development charges as set forth Paragraph 8.2.2, G.3., Collowing, apply for the hours required to design, develop, test and maintain the necessary programs. The assembling and editing charge as set forth in Paragraph 8.1.7, preceding, applies in addition to all other charges

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 1st Revised Sheet 19 Replacing Original Sheet 19

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- Undertaking of the Telephone Company-(Continued) Β.
 - 2. Bill Processing Service-(Continued)
 - d. (Continued)



for all such related IC messages converted by the Telephone Company. The Telephone Company will provide to the IC the precise details of the required format. If, in the course of Telephone Company business, it is necessary to change the format, the Telephone Company will notify the involved IC's six months prior to the change. If the IC requests the IC-provided rated IC messages be reprocessed by the Telephone Company because of an IC error, the Telephone Company will reprocess the IC-provided rated IC messages, and the appropriate charges, as set forth in Paragraph 8.2.1, G., following, will apply

- e. For End User's accounts in its operating territory where the IC has ordered Bill Processing Service, the Telephone Company will bill all rated IC messages provided by the IC. The bill format will be determined by the Telephone Company. However, subject to availability of the necessary equipment, the IC may request up to five (5) lines of informational or promotional wording on each of the IC's End User bills based on written instructions provided by the IC. Program development charges, as set forth in 8.2.1, G., following, apply for the hours required to change, design, develop, test and maintain the necessary programs to include such lines in the bill format. In addition, charges for Bill Page Information, as set forth in 8.2.1, G., following, will also apply. Any informational or promotional wording submitted by an IC for inclusion on End User bills is subject to approval by the Telephone Company.
- Upon acceptance by the Telephone Company of a Special Order for Bill f. Processing Service from an IC, the Telephone Company will determine the conditions and the period of time to implement such service on an individual-order basis. Program development charges as set forth in Paragraph 8.2.1, G., following, apply for the hours required to design, develop, test and maintain the necessary programs including any programs to rate, change the rates of or change the rate structure of any rate elements associated with the IC services.

The Telephone Company will provide Bill Processing Service on g. the condition that it purchase the accounts receivable from the IC as set forth in CARAGE 8.2.3, following. SEP 29 1986

Issued: AUG 29 1986

(AT)

986 BY LAN 1 BY R. D. BARRON, SErvice Commission SouthWestern, RessOlutephone Company St. Louis, Missouri

JAN 1 1988

Access Services Tariff Section 8 Original Sheet 19

DEC 27 CTU

83-253

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - Β. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - d. (Continued)

for all such related IC messages converted by the Telephone Company. The Telephone Company will provide to the IC the precise details of the required format. If, in the course of Telephone Company business, it is necessary to change the format, the Telephone Company will notify the involved IC's six months prior to the change. If the IC requests the ICprovided rated IC messages be reprocessed by the Telephone Company because of an IC error, the Telephone Company will reprocess the IC-provided rated IC messages, and the appropriate charges as set forth in Paragraph 8.2.1, G., following, will apply.

- e. For End User's accounts in its operating territory where the IC has ordered Bill Processing Service, the Telephone Company will bill all rated IC messages provided by the IC. The bill format will be determined by the Telephone Company.
- f. Upon acceptance by the Telephone Company of a Special Order for Bill Processing Service from an IC, the Telephone Company will determine the conditions and the period of time to implement such service on an individual-order basis. Program development charges as set forth in Paragraph 8.2.1, G.3., following, apply for the hours required to design, develop, test and maintain the necessary programs including any programs to rate, change the rates of or change the rate structure of any rate elements associated with the IC services.
- g. The Telephone Company with provide Bill Processing Service only on the condition that it purchase the accounts receivable from the IC as set forth in Paragraph 8.2.3, following.

PUBLIC SERVICE COMMISSION

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(AT)

(FC)

(AT)

Access Services Tariff Section 8 1st Revised Sheet 20 Replacing Original Sheet 20

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - h. The Telephone Company will not provide any information related to Bill Processing Service accounts under this Section of the Tariff. Bill Processing Services information may be obtained as set forth in Paragraph 8.4, following.
 - i. The Telephone Company will, at the option of the IC, provide message-billed Bill Processing Service with or without inquiry and bulk-billed Bill Processing Service with or without inquiry. In addition, the Telephone Company will, at the option of the IC, investigate End User message bill charges. Such services will be provided as follows:
 - When the Telephone Company provides inquiry, the Telephone (1)Company will be responsible for contacts and arrangements with the IC's End Users concerning the billing, collecting crediting and adjusting of the IC service charges, except prior IC balances due from End Users, in accordance with written instructions furnished by the IC. At the request of the IC, when the IC has ordered inquiry, the billed IC messages which are removed from an End User's bill in accordance with IC inquiry instructions will be reviewed for unauthorized use of IC service by Telephone Company message investigation groups for a period of up to ninety (90) days after the billed IC message has been removed from an End User's bill. For any billed IC messages removed from an End User's bill in accordance with IC inquiry instructions, the Telephone Company will make apropriate adjustments to the IC's accounts receivable. Inquiry will only be provided when the IC is provided Bill Processing Service at the same time for the same state operating ...

area CANCELLED (MT)JAN 1 1988 SEP 29 1986 BY 12× RS6 Public Service Commission Public Service Commission MASSOURI Issued: AUG 29 1986 ^{ve:} SEP 29 1986 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff Section 8 Original Sheet 20

DEC 20 (20)

Fible S

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - h. The Telephone Company will not provide any information related to Bill Processing Service accounts under this Section of the Tariff. Bill Processing Services information may be obtained as set forth in Paragraph 8.4, following.
 - i. The Telephone Company will, at the option of the IC, provide message-billed Bill Processing Service with or without inquiry and bulk-billed Bill Processing Service with or without inquiry. When the Telephone Company provides inquiry, the Telephone Company will be responsible for contacts and arrangements with the IC's End Users concerning the billing, collecting, crediting and adjusting of the IC service charges, except prior IC balances due from End Users, in accordance with written instructions furnished by the IC. At the request of the IC, when the IC has ordered inquiry, the billed IC messages which are removed from an End User's bill in accordance with IC inquiry instructions will be reviewed for unauthorized use of IC service by Telephone Company message investigation groups for a period of up to 90 days after the billed IC message has been removed from an End User's bill. For any billed IC messages removed from an End User's bill in accordance with IC inquiry instructions, the Telephone Company will make appropriate adjustments to the IC's accounts receivable. When the Telephone Company provides Bill Processing Service without inquiry, all contacts from IC End Users concerning the ICbilled amounts will be referred to the IC, and the Telephone Com-pany will only be responsible for contacts with IC's End Users concerning the collection of IC service deposits and charges, except prior IC balances due from End Users. In Figy will only be provided when the IC has ordered BillyProcessing Service for the same state operating area. SEP
 - j. The Telephone Company will accept IC stitute of the set of the

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - i. (Continued)
 - (2) When the Telephone Company provides Bill Processing Service without inquiry, all contacts from IC End Users concerning the IC-billed amounts will be referred to the IC, and the Telephone Company will only be responsible for contacts with IC's End Users concerning the collection of IC service deposits and charges, except prior IC balances due from End Users. Inquiry will only be provided when the IC has ordered Bill Processing Service for the same state operating area.
 - (3) Investigation of End User message bill charges is a Telephone Company review of charges billed to an IC's End User which the IC proposes to remove from the End User bill or has removed from the End User bill. The review will be provided based on a written order from the IC. The review will include an investigation to determine the proper party to be billed for the IC designated messages.

Investigation also includes the review of a billed IC message removed from an End User bill at the IC's request for a period of up to ninety (90) days after the billed IC message has been removed from an End User's bill. The review is provided to identify suspected unauthorized use of the IC's service.

AUG 29 1986 Investigation to determine if unauthorized use has occurred is not provided under this section, but is provided as set FALED JAN 1 1988 BY 127 AS 6 BY 127 AS 6 BY 127 AS 6 BY 127 AS 6 Public Service Commission MISSOURI Effective: SEP 29 1986

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



(FC)

(AT)

Issued:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - i. (Continued)

(AT)

(MT)

(3) (Continued)

The Telephone Company will select the personnel to make the review, and the response will be provided by telephone to the IC. The rates as set forth in 8.2.1, G., following, will apply to each request received from the IC. If adjustments are to be made to the End User account, they will be made in accordance with the IC's order and the appropriate charges as set forth in 8.2.1, F., following, will apply.

j. The Telephone Company will accept IC gift certificates for payment from End Users if the IC agrees in writing to redeem all such gift certificates. The format of the gift certificate must be acceptable to the Telephone Company.

CANCELLED JAN 1 1988 Public Service Commission MISSOURI



SEP 29 1986

Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986



Access Services Tariff Section 8 Original Sheet 21

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

B. Undertaking of the Telephone Company-(Continued)

DEC 20 CO

- 2. Bill Processing Service-(Continued)
 - k. Rated IC messages input to Bill Processing Service which the Telephone Company cannot bill for any reason will be reviewed by the Telephone Company's message investigation groups. Upon completion of the review, the billable messages will be posted and the appropriate charges as set forth in Paragraphs 8.2.1, G.7. and 8., following, will apply. Unbillable messages will be handled in accordance with instructions that have been mutually determined by the Telephone Company and the IC. At the request of the IC, the rated IC messages which cannot be billed to an End User will be reviewed for unauthorized use of IC service by Telephone Company message investigation groups for a period of up to 90 day. After the rated IC message was processed.
 - I. The Telephone Company will post rated IC messages to the topprastrate End User's account when it identifies an IC message to be offlied to an End User and will mark the appropriate End User's viscout when an IC credit card is issued to an End User. The Telephone Company will bill to an End User other IC message-billed service charges, such as provision of a credit card, issuing of a credit card, blocking of third-number billing, time and rate charges and subscription charges when it receives an order for such services from an IC. Other IC message-related charges, such as Directory Assistance and DIAL-IT charges, will be billed to the End User based on IC message data received from Message Processing Service or from the IC. The Telephone Company will make adjustments to End User's balances due to account for application of credits authorized by IC inquiry instructions and IC-furnished statements.
 - 83-253
 m. The Telephone Company will establish an End User's account for IC bulk-billed service when it receives an order from an IC to per-view form such activity for a specific End User and will bill IC bulk-billed charges to the End User. The Telephone Company will bill other IC bulk-billed rate elements, such as provision of a bulk-billed service access line, installation of an access line and provision of an access line extension when it receives an order for such services from an IC. Other IC message-related charges

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Rv R. D. BARRON, Vice President-Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - m. (Continued)

for bulk-billed service, such as Directory Assistance and DIAL-IT charges, will be billed to the End User based on IC message data received from Message Processing Service or from the IC. The Telephone Company will make adjustments to End User's balances due to account for application of credits authorized by IC inquiry instructions and IC-furnished statements.

- n. Where the rates for IC services have been implemented under an accounting order pending final approval from a regulatory agency, the Telephone Company will, upon written request from the IC, keep such records as may be required to make any adjustments to End Users as may be ordered by the regulatory agency. The charges for such a service will be determined on an indificulal-case basis. sage Billing Service Ordering
- 3. Message Billing Service Ordering
- 1st RS10 The Telephone Company will provide Message Billing Services under а. a Special Order. For all Message Billing Service or Miller than establishment of or changes to End User's account data (including credit card data), establishment of or changes to End User's account rate elements, changes to End User's balances due and Bill Page Information, the Message Billing Service Special Order Charge as set forth in Paragraph 8.2.1, G., following, will apply to orders accepted by the Telephone Company. The format of this Special Order will be specified by the Telephone Company.
 - b. The Telephone Company will arrange with the IC to accept under a Special Order, End User's account information to establish and change End User's account data (including credit card data), establish and change an End User's account rate elements and change End User's balances due. The methods, procedures and manner in which the End User's account data and changes are forwarded to the Telephone Company must be agreeable to the Telephone Company. The charges to handle such Special words except Special Orders for application of credit adjustments IC charges that change End User balances due, will be determined on an individual-case basis.

Issued: AUG 29 1986

Effective: SEP 29 1989 blic Service Commission

1988

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Replacing-Original-Sheet REGEI AUG 28 1986 MISSOURI **Public Service Commission**

Section 8

Access Services Tariff

1st Revised Sheet 22



(AT)

(AT)

No supplement to this tariff will be issued. except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 22

DEC 2 CLU

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- B. Undertaking of the Telephone Company-(Continued)
 - 2. Bill Processing Service-(Continued)
 - (Continued) п.

for bulk-billed service, such as Directory Assistance and DIAL-IT charges, will be billed to the End User based on IC message data received from Message Processing Service or from the IC. The Telephone Company will make adjustments to End User's balances due to account for application of credits authorized by IC inquiry instructions and IC-furnished statements.

- n. Where the rates for IC services have been implemented under an accounting order pending final approval from a regulatory agency, The Telephone Company will, upon written request from the IC, keep such records as may be required to make any adjustments to End Users as may be ordered by the regulatory agency. The charges for such a service will be determined on an individual-case basis.
- 3. Message Billing Service Ordering
 - The Telephone Company will provide Message Billing Services under a. a Special Order. For all Message Billing Service, other than establishment of or changes to End User's account data (including credit card data), establishment of or changes to End User's account-rate elements and changes to the User's balances due, the Message Billing Service Special Order charge as set forth in Paragraph 8.2.1, G.14., following, will apply to orders accepted by the Telephone Company. The format of this Special Order will be specified by the Telephone Company. 83 - 253
 - b. The Telephone Company with Arrange Vista the IC to accept under a Special Order, End User's Paccount Information to establish and change End User's account data (including credit card data), establish and change an End User's account rate elements and change End User's balances due. The methods, procedures and manner in which the End User's account data and changes are forwarded to the Telephone Company must be agreeable to the Telephone Company. The charges to handle such Special Orders will be determined on an individual-case basis.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 23

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - C. Liability of the Telephone Company

n in the second se

> DEC 25 DEC

Notwithstanding Section 2, Paragraph 2.1.3, preceding, the Telephone Company liability for Message Billing Service is as follows:

- 1. If Bill Processing Service detail is not available because the Telephone Company lost or damaged records or incurred processing system outages, the Telephone Company will attempt to recover the lost IC detail. If the lost IC detail cannot be recovered and the Telephone Company recorded the details, the IC detail and the extent of the Telephone Company's liability for damages will be as set forth in Paragraph 8.1.3, A., preceding. If the lost IC detail cannot be recovered and the IC provided the detail, the IC will be requested to resupply the detail. If the IC cannot resupply the detail, the detail and the extent of the Telephone Company's liability for damages will be as set forth in Paragraph 8.1.3, A., preceding. This recovered detail will be included in message detail provided to the IC when the IC orders such service and any recovered IC messages will be billed.
- 2. When the Telephone Company is notified that, due to its error or omission, incomplete IC detail has been provided to an IC as set forth in Paragraph 8.2.1, B.1.j., preceding, and/or Paragraph 8.4, following, to an IC, the Telephone Company will make every reasonable effort to recover and provide the IC detail to the IC at no additional charge. Such request to recover the IC detail must be made within 30 days from the date the IC detail was initially made available to the IC. If the detail cannot be recovered, the extent of the Telephone Company's liability for damages shall be the total as set forth in Paragraph 8.1.3, A., preceding.
- A., preceding.
 3. If the Telephone Company finds of is notified of an error in billing to an End User, it will correct the error and bill the appropriate End User within the limits permitted by laws of the state in wigh ip provides the service. If the error is caused by the Telephone Company and the Telephone Company cannot timely bill the proper End User, the extent of the Telephone Company's liability for damages will be the known amount misbilled or when the amount misbilled is unknown, limited as set forth in Paragraph 8.1.3, A., preceding.
- 4. In the absence of willful misconduct, no liability for damages to the IC or other person other than that as set forth in Paragraphs 8.2.1, C.1., 2. and 3., preceding, shall attach to the Telephone Company for its action or the conduct of its employees in providing Message Billing Service.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 24

and the second

UEU 2. Las

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

D. Obligations of the IC

> The IC shall order Message Billing Services under a Special Order for 1. each state where service is desired. The IC shall be responsible for all balances due from End Users that exist prior to ordering Bill Processing Service.

At the time Message Processing Service and/or Bill Processing Service is initially ordered, the IC shall order the service for one, three or five years. Thereafter, upon six months' written notice, additional service may be ordered for one, three or five years at the rates and charges as set forth in Paragraph 8.2.1, G., following. The IC may order inquiry for a service period that is different from that for Bill Processing. However, the IC shall not order inquiry unless it also has ordered Bill Processing Service for the same period and the same state operating area. Not later than six months prior to the end of an order period, the IC shall notify the Telephone Company in writing if service is to be discontinued at the end of the period. If no notice is received from the IC, the Telephone Company will automatically extend the services for another year, using the most recent 12 months of capacity provided. The IC will be notified by the Telephone Company when such an extension is made. All appropriate charges as set forth in Paragraph 8.2.1, G., following, for another year will apply, and the minimute Darges will be based on the most recent 12 months of message papacity and/or bill capacity provided. 1988

2. When Message Processing Service jalordered theeld shall furnish the Telephone Cormany for each state Telephone Company, for each state and for each year in the order, an estimate of the number of messages (messages apacity), including those messages which will be bulk-billed to be processed. 1123 112

In addition, when Bill Processing Service is ordered, the IC shall furnish the Telephone Company, for each state and for each year in the order, an estimate of the number of message-billed and/or bulkbilled messages (bill capacity) for which billing is to be provided. Separate estimates shall be furnished by the IC for message-billed messages and bulk-billed messages. The capacity estimate for inquiry for each state and each year shall be the same as that for messagebilled and/or bulk-billed messages.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 25

DEC 20 (TO)

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - D. Obligations of the IC-(Continued)
 - 3. The IC shall furnish all information necessary for the Telephone Company to provide the Message Billing Service, including any per-month service charges applicable to an End User, and an affidavit that states whether the IC service is subject to any federal and/or state taxes. When IC messages are to be billed by an entity other than the Telephone Company, the IC shall furnish written instructions as to how the rated IC messages are to be provided to that other entity. If the IC does not furnish complete instructions, all resulting unbillable messages will be delivered to the IC. The information shall be furnished by the IC in a timely manner.
 - 4. The IC shall furnish to the Telephone Company a written schedule of its rates and charges in sufficient time to allow the Telephone Company to establish a rating program. The IC's rate structure must be consistent with established Telephone Company rating methodologies. The interval required to establish a rating program must be mutually agreeable to the Telephone Company and the IC.
 - 5. When the IC orders Bill Processing Service, it shall authorize the Telephone Company in writing to deny service to End Users for non-payment. If that authorization is not received, Bill Processing Service will not be provided.
 - 6. The IC shall be responsible for all contacts and arrangements, including prior IC balances due from End Users, with its End Users concerning the provision and maintenance of the IC's service.
 - 7. When the IC orders message-billed or bulk-billed Bill Processing Service with inquiry, the IC shall furnish to the Telephone Company written instructions, which are agreeable to the Telephone Company, for the handling of End User's questions about bills.

When the IC orders message-billed or bulk-billed Bill 8 Boces 3 3 Service without inquiry, the IC shall furnish the Telephone Company with written instructions as to where inquiries are to be referred.

JAIN 1 1988

83 BYEFTEE South JAN 0 1 1984 By R. D. EARRONIDUCE NATSIDENT Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Issued: DEC 2 9 1983

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - D. Obligations of the IC-(Continued)
 - 7. (Continued)

Access Services Tariff Section 8 Original Sheet 26

DEU 2. WWW

When the IC does not order inquiry service and desires credit adjustments be made to the End User's balance due, the IC shall furnish a statement for each End User's account where the credit is desired. These statements shall show the IC message, the date the IC message was billed and the amount of the credit. These statements shall be furnished to the Telephone Company under a Special Order as set forth in Paragraph 8.2.1, B.3.b., preceding.

The IC shall notify its End Users through its tariff or other appropriate means when the IC handles the bill inquiries. The IC shall furnish the Telephone Company in writing all End User's bill adjustment statements.

- 8. The IC will immediately redeem all IC gift certificates the Telephone Company receives in payment for any End User's charges. The IC agrees to use a gift certificate format which is agreeable to the Telephone Company.
- 9. The IC agrees to permit the Telephone Company to determine in accordance with Telephone Company regulations, bill and collect IC service deposits from all End Users of the IC's services for which the Telephone Company provides billing for the IC. The IC will notify its End Users through its tariffs or other means that the Telephone Company will, when necessary, in accordance with Telephone Company deposit regulations, determine and collect IC service deposits. The IC will also include in its tariff and service Arrangements and obtain regulatory concurrence for the Telephone Company deposit regulations that the Telephone Company will use Conducter and collect End User's service deposits.
- deposits.
 10. When the IC desires that a credit card bendssued by the Telephone Company for an End User, the BC shate furnish the credit card information as specified by the Telephone Company. The information shall include a statement from the IC that the End User has requested the credit card. When the IC is notified by the Telephone Company that an IC credit card billing is discontinued, the IC shall notify the appropriate End User. The IC also agrees to be responsible for all charges to the discontinued credit card after receipt of notice of discontinuance, and all charges to an IC credit card where the End User states in writing to the Telephone Company that the End User did not request the credit card.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 27

H. Herring

DEC 20 1000

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- D. Obligations of the IC-(Continued)
 - 11. When the IC furnishes recorded IC detail for Message Processing Service and/or rated IC message detail for Bill Processing Service, it shall be responsible to deliver the detail to the location specified by the Telephone Company, and it shall retain a copy of the detail furnished for at least 90 days.
- E. Payment Arrangements and Audit Provision
 - 1. Audit Provision

Upon written notice by the IC to the Telephone Company, the IC shall have the right, through its authorized representative to examine and audit during normal business hours and at reasonable intervals determined by the Telephone Company, all such records and accounts as may under recognized accounting practices contain information bearing upon the amount payable to the IC. Adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit. Neither such right to examine and audit nor the right to receive such adjustment shall be affected by any statement to the contrary, appearing on checks or otherwise, unless such statement expressly waiving such right appears in a letter signed by the authorized representative of the party having such right and delivered to the other party.

Any information received or reviewed by the IC or its authorized representative during the audit is to be considered confidential and not to be distributed, provided or disclosed in any form to anyone not involved in the audit, non ERsuch information to be CANCEL 用于正常。 used for any other purpose.

2.

Minimum Period The minimum period for which Message Hilling Service is provided and for which charges apply is one Wear, cell the IC orders Message Proc-essing Service and/or Bill Processing Service for three or five years, then the minimum period and the period for which charges apply is that period ordered by the IC. A minimum period of one, three or five years, as ordered by the IC, applies for each additional period of service ordered.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- E. Payment Arrangements and Audit Provision-(Continued)
 - 2. Minimum Period-(Continued)

If the service is discontinued prior to the end of the period ordered, monthly charges apply for each remaining month and fraction of a month. The monthly charge will be 1/12 of the minimum yearly charge.

3. Minimum Order Capacities and Minimum Yearly Charges

Message Billing Service is subject to minimum order capacities and minimum yearly charges. The minimum order capacities are the message capacity and bill capacity as specified in Paragraph 8.2.1, E.3.a., following. The minimum yearly charge as specified in Paragraphs 8.2.1, E.3.b. or c., following, whichever is higher, shall apply.

a. The message capacity ordered by the IC for any year in a threeto five-year order for Message Processing Service shall not be lower than 25 percent of the largest message capacity ordered for any year in a three- or five-year order. The bill capacity ordered by the IC for any year in a three- or five-year order for Bill Processing Service shall not be lower than 25 percent of the largest bill capacity ordered for any year in a threeor five-year order. The minimum yearly bill capacity for messagebilled messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth in Paragraph 8.2.1, D.2., preceding. The minimum yearly capacity ordered for each year for inquiry shall be the same as that for message-billed and/or bulk-billed messages.

b. For Message Processing Service, the minimum yearly charge is the product of the message capacity per year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a, preceding, and the appropriate Message Processing Service rate times 0.7. The appropriate Message Processing Service rate as set forth in Paragraph 8.2.1, G.1., following, for the specific year period or define.

CANCE SEP 29 1986 Bublic Service Commission BY Issued: AUG 29 1986 By R. D. BARRON, President descuri Division Southwestern Bell Terephone Company

St. Louis, Missouri

(CP)

aug 2/8/1980 Missouri

Section 8

Access Services Tariff

Replacing DE T

1st_Revised-Sheet-28

Public Service Commission

Access Services Tariff Section 8 Original Sheet 28

DEC 28 CON

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provision-(Continued)
 - 2. Minimum Period-(Continued)

If the service is discontinued prior to the end of the period ordered, monthly charges apply for each remaining month and fraction of a month. The monthly charge will be 1/12 of the minimum yearly charge.

3. Minimum Order Capacities and Minimum Yearly Charges

Message Billing Service is subject to minimum order capacities and minimum yearly charges. The minimum order capacities are the message capacity and bill capacity as specified in Paragraph 8.2.1, E.3.a., following. The minimum yearly charge as specified in Paragraphs 8.2.1, E.3.b. or c., following, whichever is higher, shall apply.

- a. The message capacity ordered by the IC for any year in a threeto five-year order for Message Processing Service shall not be lower than 25 percent of the largest message capacity ordered for any year in a three- or five-year order. The bill capacity ordered by the IC for any year in a three- or five-year order for Bill Processing Service shall not be lower than 25 percent of the largest bill capacity ordered for any year in a threeor five-year order. The minimum yearly bill capacity for messagebilled messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth in Paragraph 8.2.1, D.2., preceding. The minimum yearly capacity ordered for each year for inquiry shall be the same as that for message-billed and/or bulk-billed messages.
- b. For Message Processing Service, the minimum yearly charge is the product of the message rapacity per year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a, preceding S and the appropriate Message Processing Service/rate times 0.9253 appropriate Message Processing Service rate as set forth in Paragraph 3.2.1, G.1., following, for the specific year-period ordered.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff 1st Reg: Replacing Guigina AUG 28 1986 MISSOURI

Public Service Commission

- ACCESS SERVICES
- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provision-(Continued)
 - 3. Minimum Order Capacities and Minimum Yearly Charges-(Continued)
 - (Continued) b.

(CP)

(CP)

(CP)

For Bill Processing Service, the minimum yearly charge is the product of the message-billed bill capacity and/or bulk-billed bill capacity per year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the appropriate Bill Processing Service message-billed and/or bulk-billed rate times 0.7. The minimum yearly charge for message-billed messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding. The appropriate Bill Processing Service message-billed rate is the average of the messagebilled rates as set forth in Paragraph 8.2.1, G.7., following, for the specific message-billed service year-period ordered. The appropriate Bill Processing Service bulk-billed rate is the bulkbilled rate as set forth in Paragraph 8.2.1, G.7., following, for the specific bulk-billed service year-period ordered.

For inquiry, the minimum yearly charge for message-billed inquiry and bulk-billed inquiry will be determined separately. For messageinquiry, the minimum yearly charge is the product of the messagebilled bill capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a, preceding, and the appropriate message-billed inquiry rate times 0.7. For bulk-billed inquiry, the minimum yearly charge is the product of the bulk-billed bill capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a, preceding, and the appropriate bulk-billed inquiry rate times 0.7. The appropriate message-billed inquiry rate is the average of the message-billed inquiry rates as set forth in Paragraph 8.2.1, G.7., following, for the year-period ordered. The appropriate bulk-billed inquiry pte is the bulk-billed inquiry rate as set forth in Paragraph E.2.1, G.7., following, fdp the year-period ordered. SEP 29 1986 Commission

SEP 29 1986

Public Service Commission

Issued: AUG 29 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

BI Service Com

Access Services Tariff Section 8 Original Sheet 29

UEU 2.7 KCO

Testion

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provision-(Continued)
 - 3. Minimum Order Capacities and Minimum Yearly Charges-(Continued)
 - b. (Continued)

For Bill Processing Service, the minimum yearly charge is the product of the message-billed bill capacity and/or bulk-billed bill capacity per year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the appropriate Bill Processing Service message-billed and/or bulk-billed rate times 0.9. The minimum yearly charge for message-billed messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth Paragraphs 8.2.1, D.2. and E.3.a., preceding. The appropriate Bill Processing Service message-billed rate is the average of the messagebilled rates as set forth in Paragraph 8.2.1, G.7., following, for the specific message-billed service year-period ordered. The appropriate Bill Processing Service bulk-billed rate is the bulkbilled rate as set forth in Paragraph 8.2.1, G.7., following, for the specific bilk-billed service year-period ordered.

For inquiry, the minimum yearly charge for message-billed inquiry and bulk-billed inquiry will be determined separately. For messageinquiry, the minimum yearly charge is the product of the messagebilled bill capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a, preceding, and the appropriate message-billed inquiry rate times 0.9. For bulk-billed inquiry, the minimum yearly charge is the product of the bulk-billed bill capacity for the year furnished by the IC as set forth is garage 8.2.1, D.2. and E.3.a, preceding, and the appropriate bulk-billed inquiry rate times 0.9. The appropriate message-billed inquiry rate is the average of the message-billed inquiry rates as set forth in Paragraph 8.2.1, G.7., following for the year period ordered. The appropriate bulk-billed inquiry rate is the bulk-billed inquiry rate as set forth in Paragraph 8.2.1, G.7.966 following, for the year-period ordered.

PUBLIC SERVICE COMMISSION OF MISSOURI

Issued:

NFC 2 9 1983

Effective: JAN 0 1 1984

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - Ε. Payment Arrangements and Audit Provision-(Continued)
 - 3. Minimum Order Capacities and Minimum Yearly Charges-(Continued)
 - c. For Message Processing Service, the minimum yearly charge is the difference of the message capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the year allowance times the appropriate Message Processing Service rate. The year allowance will be determined by the Telephone Company and will be the total IC messages for all IC's processed by the Telephone Company in the calendar year prior to the year for the minimum yearly charge times 0.25. The appropriate Message Processing Service rate is the Message Processing Service rate as set forth in Paragraph 8.2.1, G.1., following, for the specific year-period ordered.

For Bill Processing Service, the minimum yearly charge is the difference of the message-billed bill capacity plus the bulkbilled bill capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the year allowance times the appropriate Bill Processing Service message-billed and/or bulk-billed rate. The minimum yearly charge for message-billed messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and a prorate of the allowance. The allowance will be prorated using the ratio of the message-billed bill capacity furnished by the IC to the bulk-billed bill capacity furnished by the IC for the year for which the minimum yearly charge is determined. The year allowance will be determined by the Telephone Company and will be the total IC messages for all IC's billed by the Telephone Company in the calendar year prior to the year for the minimum yearly charge times 0.25. The appropriate Bill Processing Service message-billed rate is the average of the message-billed rate as set forth in Paragraph 8.2.1, G.7., following, for the year-period ordered. The appropriate Bill-Processing Service bulk-billed rate is the bulk-billed rate as set Edith in Paragraph 8.2.1, G.7., folloging, for the year-period ordered. CANCE SEP 29 1986

Access Services Tariff

Replacing Original Sheet REC

1st Revised Sheet 30

E

AUG 28 1986

MISSOURI

Public Service Commission

Section 8

Public Service Commission

BY R. D. BARRON: CRESiden Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

1988

(CP)

(CP)

Issued: AUG 29 1986

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

Public Seattle

DEC 20 KCCU

Access Services Tariff

Original Sheet 30

Section 8

- E. Payment Arrangements and Audit Provision-(Continued)
 - 3. Minimum Order Capacities and Minimum Yearly Charges-(Continued)
 - c. For Message Processing Service, the minimum yearly charge is the difference of the message capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the year allowance times the appropriate Message Processing Service rate. The year allowance will be determined by the Telephone Company and will be the total IC messages for all IC's processed by the Telephone Company in the calendar year prior to the year for the minimum yearly charge times 0.05. The appropriate Message Processing Service rate is the Message Processing Service rate as set forth in Paragraph 8.2.1, G.1., following, for the specific year-period ordered.

For Bill Processing Service, the minimum yearly charge is the difference of the message-billed bill capacity plus the bulkbilled bill capacity for the year furnished by the IC as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and the year allowance times the appropriate Bill Processing Service message-billed and/or bulk-billed rate. The minimum yearly charge for message-billed messages and bulk-billed messages will be determined separately based on the estimates the IC furnishes as set forth in Paragraphs 8.2.1, D.2. and E.3.a., preceding, and a prorate of the allowance. The allowance will be prorated using the ratio of the message-billed bill capacity furnished by the IC to the bulk-billed bill capacity furnished by the IC for the year for which the minimum yearly charge is determined. The year allowance will be determined by the Telephone Company and will be the total IC messages for all IC's billed by the Telephone Company in the calendar year prior to the year for the minimum yearly charge times 0.05. The appropriate Bill Processing Service message-billed rate is the average of the message-billed rate as set forthin Paragraph 8.2.1, G.7., follow-ing, for the year-period ordered. The appropriate Bill Processing Service bulk-billed rate is the bulk-billed rate as-set forth in Paragraph 8.2.1, GST.P. 2following, for the years giriog 5rgered.

Issued:

DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

BY 124 B 5 # 3.0 PUDVIC SERVICE COMMISSION OF MISSOURI

Access Services Tariff Section 8 1st Revised Sheet 31 Replacing-Original Sheet Ê **위**도 AUG 28 1986 MISSOURI Public Service Commission

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- E. Payment Arrangements and Audit Provision-(Continued)
 - Minimum Order Capacities and Minimum Yearly Charges-(Continued) 3.
 - c. (Continued)

For inquiry, the minimum yearly charge will be determined for the year by using the message-billed bill capacity or the bulk-billed capacity less a prorate of the year allowance times the appropriate inquiry message-billed or bulk-billed inquiry rate. The minimum yearly charge for message-billed inquiry and the bulk-billed inquiry will be determined separately. The prorate of the allowance will be the same as that set forth in the preceding paragraph.

4. Cancellation of a Special Order

An IC may cancel a Special Order for Message Billing Service or Bill Page Information on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for Message Billing Service or Bill Page Information is the date the IC and the Telephone Company mutually agree the service is to start.

- (AT) When an IC cancels a Special Order for Message Billing Service or Bill Page Information after the order date but prior to the start of service, a charge equal to the Special Order Charges, Elogram development costs and any nonrecoverable capital model incurred by the Telephone Company will apply to the IC. CAN and
 - 5. Change to Special Orders

of the change will apply.



SFP 29 1986

Public Service Commission

(AT)

(AT)

(AT)

Issued: AUG 29 1986

Effective: SEP 2

Access Services Tariff Section 8 Original Sheet 31

DEC 20 1000

Contractor and

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provision-(Continued)
 - 3. Minimum Order Capacities and Minimum Yearly Charges-(Continued)
 - (Continued) c.

For inquiry, the minimum yearly charge will be determined for the year by using the message-billed bill capacity or the bulk-billed capacity less a prorate of the year allowance times the appropriate inquiry message-billed or bulk-billed inquiry rate. The minimum yearly charge for message-billed inquiry and the bulk-billed inquiry will be determined separately. The prorate of the allowance will be the same as that set forth in the preceding paragraph.

4. Cancellation of a Special Order

An IC may cancel a Special Order for Message Billing Service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for Message Billing Service is the date the IC and the Telephone mutually agree the service is to start.

When an IC cancels a Special Order for Message Billing Service after the order date but prior to the start of service, a charge equal to the Special Order Charges, program development, costs and any nonrecoverable capital costs incurred by the Telephone Company will apply to the IC. JAN = 1 (Sa) 83 - 253

5. Change to Special Orders

When an IC requests changes to a pending Special Wher for Message Billing Service, such changes will be hindeftaken if they can be accommodated by the Telephone Company. W A charge equal to any costs incurred by the Telephone Company because of the change will apply. SEP 2.9 1300

PUBLIC SERVICE COMMISSION OF MISSOUR

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations

(CP)

(CP)

(CP)

REGEIVED AUG 28 1986 MISSOURI Public Service Commission

- 1. The Message Processing and Bill Processing Service message charges apply during the yearly periods ordered by the IC. The Telephone Company will not initiate rate changes to the Message Processing Service charges as set forth in Paragraph 8.2.1, G.1., following, or the Bill Processing Service charges as set forth in Paragraph 8.2.1, G.7., following, that apply to the IC order during the yearly periods for that specific order. For the purpose of determining the charges applicable to bulk-billed service for Bill Processing Service, a bulk-billed message is an IC message which is used by the Telephone Company to develop the IC bulk-billed charge.
- 2. During any yearly period in which the actual IC messages processed and/or billed exceeds the message capacity as stated following, additional charges apply. For Message Processing Service, for each IC message processed that exceeds (a) 130 percent of the message capacity ordered for the year or (b) the message capacity ordered plus the year allowance, whichever of (a) or (b) is lower, the appropriate charges for the specific service ordered as set forth in Paragraphs 8.2.1, G.1. and 2., following, apply. The year allowance is the quantity determined as set forth in Paragraphy Azer. 1, E.3.c., preceding.
- For Bill Processing Service, for each message-billed IC message processed that exceeds (a) 130 percent of the message-billed privious capacity ordered or (b) the message-billed bill capacity plusicate provated allowance, whichever of (a) or (b) is lower, the appropriate Charges for the specific service ordered as set forth in Paragraphs 8.2.1, G.7. and 8., following, apply. The prorated allowance is the quantity determined as set forth in Paragraphs 8.2.1, E.3.c., preceding.

For Bill Processing Service, for each bulk-billed IC message processed that exceeds (a) 130 percent of the bulk-billed bill capacity ordered or (b) the bulk-billed bill capacity plus the prorated allowance, whichever of (a) or (b) is lower, the appropriate charges for the specific service ordered as set forth in Paragraph 8.2.1.7777. and 8., following, apply. The prorated allowance is unellar protated quantity determined as set forth in Paragraph 8.2.1, E.3.c., preceding.

Issued: AUG 29 1986

Effective: SEP 29-1986

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 32

DEC 201 3

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.1 Message Billing Service-(Continued)

- F. Rate Regulations
 - The Message Processing and Bill Processing Service message charges 1. apply during the yearly periods ordered by the IC. The Telephone Company will not initiate rate changes to the Message Processing Sarvice charges as set forth in Paragraph 8.2.1, G.l., following, or the Bill Processing Service charges as set forth in Paragraph 8.2.1, G.7., following, that apply to the IC order during the yearly periods for that specific order. For the purpose of determining the charges applicable to bulk-billed service for Bill Processing Service, a bulk-billed message is an IC message which is used by the Telephone Company to develop the IC bulk-billed charge.
 - 2. During any yearly period in which the actual IC messages processed and/or billed exceeds the message capacity as stated following, additional charges apply. For Message Processing Service, for each IC message processed that exceeds (a) 110 percent of the message capacity ordered for the year or (b) the message capacity ordered plus the year allowance, whichever of (a) or (b) is lower, the appropriate charges for the specific service ordered as set forth in Paragraphs 8.2.1, G.1. and 2., following, apply. The year allowance is the quantity determined as set forth in Paragraphs 8.2.1, E.3.c., preceding.

For Bill Processing Service, for each message-billed IC message processed that exceeds (a) 110 percent of the message-billed bill capacity ordered or (b) the message-billed bill capacity plus the prorated allowance, whichever of (a) or (b) is lower, the appropriate charges for the specific service ordered as set forth in Paragraphs 8.2.1, G.7. and 8., following, apply. The prorated allowance is the quantity determined as set forth in Paragraphs 8.2.1, E.3.c., preceding 3 - 253

For Bill Processing Service, for each bulk-billed IG message proc-essed that exceeds (a) 110 percent of the Bulk-billed bill capacity ordered or (b) the bulk-billed bill capacity plus the prorated allowance, whichever of (a) or (b) is lower, the appropriate charges for the specific service ordered as set forth in Paragraph 8.2.1, G.7.

OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 1st Revised Sheet 33 Replacing Original Sheet 33

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - 2. (Continued)

(CP)



For each year for message-billed inquiry, the additional charges will apply for the same capacity that additional charges apply for messagebilled messages. The appropriate charges for the specific inquiry ordered as set forth in Paragraphs 8.2.1, G.7. and 8. apply.

For each year for bulk-billed inquiry, the additional charges will apply for the same capacity that additional charges apply for bulkbilled messages. The appropriate charges for the specific inquiry ordered as set forth in Paragraphs 8.2.1, G.7. and 8. apply.

For each IC message processed and/or billed between the message capacity ordered and the allowance (i.e., 130 percent level or order plus allowance level, whichever applies) specified in the preceding paragraphs, the appropriate rate as set forth in Paragraphs 8.2.1, G.1. or 7. will apply.

The IC, at its option, may order additional capacity for inquiry and capacity required to process and/or bill the additional messages at the rates as set forth in Paragraph 8.2.1, G., following, subject to the minimum period and minimum yearly charges as set forth in Paragraphs 8.2.1, E.2. and 3., preceding. When more than one order exists at the same time for an IC, the rates for inquiry, IC messages processed and/or billed that exceed the total inquiry, message and/or bill capacity, respectively, for all existing orders will be the appropriate rates for the most recent order, plus the appropriate additional inquiry message and/or bill capacity, G.2. and S., following, for the most recent order.

MUSL recent order.
3. The Message-Billed Service Charge applies each most cost one or more messages or related rate elements are billed to an information one company to the End User on the same bill for the IC, the Message-Billed Service Charge times 0.5 applies for each month. When more than one copy of the End User's bill is provided to the End User, the Message-Billed Service Charge applies for each additional copy of the End User's bill provided.

Issued: AUG 29 1986

Effective: SEP 29 986blic Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 Original Sheet 33

DEC 20 (20)

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
- F. Rate Regulations-(Continued)
 - 2. (Continued)

For each year for message-billed inquiry, the additional charges will apply for the same capacity that additional charges apply for messagebilled messages. The appropriate charges for the specific inquiry ordered as set forth in Paragraphs 8.2.1, G.7. and 8. apply.

For each year for bulk-billed inquiry, the additional charges will apply for the same capacity that additional charges apply for bulkbilled messages. The appropriate charges for the specific inquiry ordered as set forth in Paragraphs 8.2.1, G.7. and 8. apply.

For each IC message processed and/or billed between the message capacity ordered and the allowance (i.e., 110 percent level or order plus allowance level, whichever applies) specified in the preceding paragraphs, the appropriate rate as set forth in Paragraphs 8.2.1, G.1. or 7. will apply.

The IC, at its option, may order additional capacity for inquiry and capacity required to process and/or bill the additional messages at the rates as set forth in Paragraph 8.2.1, G., following, subject to the minimum period and minimum yearly charges as set forth in Paragraphs 8.2.1, E.2. and 3., preceding. When more than one order exists at the same time for an IC, the rates for inquiry, IC messages processed and/or billed that exceed the total inquiry, message and/or bill capacity, irrespectively, for all existing orders will be the appropriate rates for the most recent order, plus the appropriate additional inquiry message and/or bill processing charge as set forth in Paragraphs 8.2.1, G.2. and 8., following, for the most recent order. RR - 959

3. The Message-Billed Service COMMISSION
83-253
3. The Message-Billed Service Charge applies each month that one or more messages or related rate elements are billed to an End User. When both interstate and state IC messages are billed by the Telephone Company to the End User on the same bill for the IC, the Message-Billed Service Charge times 0.5 applies for each month. When more than one copy of the End User's bill is provided to the End User, the Message-Billed Service Charge applies for each additional copy of the End User's bill provided.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 1st Revised Sheet 34 Replacing Original Sheet 34

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - 4. A Bulk-Billed Service Charge applies each month that one or more charges are billed to an End User. When both interstate and state IC bulkbilled charges are billed by the Telephone Company to the End User on the same bill for the IC, the Bulk-Billed Service Charges times 0.5 applies for each month. When more than one copy of the End User's bill is provided to the End User, the Bulk-Billed Service Charge applies for each additional copy of the End User's bill provided.
 - 5. When message detail is data-transmitted to or received from an exchange telephone company location by the Telephone Company, a charge as set forth in Paragraph 8.2.1, G.4., following, on a per-record basis will apply. A record is a logical grouping of information as described in the program that processes the information and loads the magnetic tape or data file used to supply the message detail which is data-transmitted. The Telephone Company will determine this charge based on its count of the records transmitted.
 - 6. When message detail is data-transmitted to or received from an IC location by the Telephone Company, a charge as set forth in Paragraphs 8.2.1, G.6. and/or 12., following, on a per-record basis will apply. A record is a logical grouping of information as described if the program that processes the information and loads the maximum table or data file used to supply the message detail which is tata-transmitted. The Telephone Company will determine this charge based on its leaunt of the records received.
 - The Message Billing Service Special Order Charge applies for each Special Order for Message Processing Service and or each Processing Service, other than establishment of or changes to End User's account data (including credit card data), establishment of or changes to End User's account rate elements, changes to End User's balances due and Bill Page Information accepted by the Telephone Company.

The End User's account activity charges apply whenever an IC Special Order requests End User's account data be established or changed, nonrecurring or recurring IC rate element be added or changed in an End User's account and/or an End User's balance due be changed.

SEP 29 1986

Issued: AUG 29 1986

(AT)

Effective: SEP 2 9/1986: Service Commission

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

REGEIVED AUG 2 8 1986 MISSOURI Public Service Commission

Access Services Tariff

Original Sheet 34

DEC 20 JUD

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - 4. A Bulk-Billed Service Charge applies each month that one or more charges are billed to an End User. When both interstate and state IC bulkbilled charges are billed by the Telephone Company to the End User on the same bill for the IC, the Bulk-Billed Service Charges times 0.5 applies for each month. When more than one copy of the End User's bill is provided to the End User, the Bulk-Billed Service Charge applies for each additional copy of the End User's bill provided.
 - 5. When message detail is data-transmitted to or received from an exchange telephone company location by the Telephone Company, a charge as set forth in Paragraph 8.2.1, G.4., following, on a per-record basis will apply. A record is a logical grouping of information as described in the program that processes the information and loads the magnetic tape or data file used to supply the message detail which is data-transmitted. The Telephone Company will determine this charge based on its count of the records transmitted.
 - 6. When message detail is data-transmitted to or received from an IC location by the Telephone Company, a charge as set forth in Paragraphs 8.2.1, G.6. and/or 12., following, on a per-record basis will apply. A record is a logical grouping of information as described in the program that processes the information and loads the magnetic tape or data file used to supply the message detail which is data-transmitted. The Telephone Company will determine this charge based on its count of the records received.
 - 7. The Message Billing Service Special Order Sharge applies for each Special Order for Message Brocessing Service and/or Bill Processing Service, other than establishment of or changes to End User's account data (including credit card data), lestablishment of or changes to End User's balances due user's account rate elements and changes to End User's balances due accepted by the Telephone Company. Statute Commission 83-253 BY Company of the company of the End User's account activity charges apply whenever an IC Special

The End User's account activity charges apply whenever an -iC Special Order requests End User's account data be established or changed, nonrecurring or recurring IC rate element be added or changed in an End User's account and/or an End User's balance due be changed.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.2 Billing Service-(Continued) 8.2.1 Message Billing Service-(Continued) F. Rate Regulations-(Continued) (Continued) 7. An End User's account is a record for message-billed service or bulkbilled service which has a unique name, address and billing identification number assigned by the Telephone Company to which a bill is rendered. (CT) An End User's account activity Special Order Charge always applies, and the End User's account establishment and change charge, End User's account rate element rate level change charge or the End User's account rate element rate structure change charge may apply depending on the activity ordered by the IC as set forth in Paragraphs 8.2.1, F.7.a., b., c. and d., following. (CT) An End User's account activity Special Order Charge applies whena. ever the IC furnishes to the Telephone Company End User's account information that establishes or changes the information, rates or (AT) balance due associated with an End User's account. The End User's account activity Special Order Charge applies whenever the IC orders Investigation of End User message bill charges. The End User's account establishment and change charge applies b. whenever IC-furnished information is used by the Telephone Company to establish or change End User's account data or rate elements or balances due, except for information to change End User's account rate element rate levels or rate structure. End User's account rate element rate level and rate structure change charges are applied as set forth in Paragraphs 20.1, F.7.c. and d., following. JAN 1 13× RS6 In addition, the End User's account establishment and istange charge does not apply when rated IC messages are posted to a message billed account associated with an 51% User's common line. The End User's account establishment and change charge does not apply when an IC credit card is listed on a messagebilled account associated with an End User's common line and does not apply for any subsequent preparation or distribution of an IC credit card. The End User's account establishment and change charge does apply when the Telephone Company the request of an IC, establishes or changes a message billed account with a credit card but without an associated End User's common line. The End User's account establistiment 1986 Issued: AUG 29 1986 **Effective:** SEP 29 986 Dic Service Commission By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

AUG 2 8 1986 MISSOURI

Section 8

Public Service Commission

REGEIVED

1st Revised Sheet 35 Replacing Original Sheet 35
No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 35

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

- 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)

F. Rate Regulations-(Continued)

DEC 2 ()

7. (Continued)

An End User's account is a record for message-billed service or bulkbilled service which has a unique name, address and billing identification number assigned by the Telephone Company to which a bill is rendered.

The End User's account activity Special Order Charge always applies, and the End User's account establishment and change charge, End User's account rate element rate level change charge or the End User's account rate element rate structure change charge may apply depending on the activity ordered by the IC as set forth in Paragraphs 8.2.1, F.7.a., b., c. and d., following.

- The End User's account activity Special Order Charge applies whenа. ever the IC furnishes to the Telephone Company End User's account information that establishes or changes the information, rates or balance due associated with an End User's account.
- b. The End User's account establishment and change charge applies whenever IC-furnished information is used by the Telephone Com-pany to establish or change End User's account data or rate elements or balances due, except for information to change End User's account rate element rate levels or rate structure. End User's account rate element rate level and rate structure change charges are applied as set forth in Paragraphs 8.2.1, F.7.c. and d., following. BY following. BY PUBLIC SERVICE COMMISSION

In addition, the End User's account establishment and change charge does not apply when rated IC messages are posted to a message-billed account associated with an End User's common line. The End User's account establishment and change charge does not apply when an IC credit card is listed on a messagebilled account associated with an End User's common line and does not apply for any subsequent preparation or distribution of an IC credit card. The End User's account establishment and change charge does apply when the Telephone Company, at the request of an IC, establishes or changes a message-billed account with a credit card but without an appociated End User's common line. The End User's account establishment

JAN (1 1984AN - 1 1984

Issued: DEC 2 9 1983

83-253 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri

Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 36

DEC 20 (LL)

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - (Continued) 7.
 - b. (Continued)

and change charge applies for each account established, rate element established, account changed (except for credit card changes), rate element changed and balance due changed.

- c. The End User's account rate element rate level change charge applies whenever IC-furnished information is used by the Telephone Company to change an End User's account rate element rate level. The charge applies for each End User's account rate element rate level changed.
- d. The End User's account rate element rate structure change charge applies whenever IC-furnished information is used by the Telephone Company to change an End User's account rate element rate structure. The charges to make the End User's account rate element rate structure changes will be determined on an individual-case basis.
- 8. When message detail is entered on a data file or magnetic tape to be provided to an IC, the per-tape charge applies for each data file or tape prepared, and the per-record charge applies for each record processed. A record is a logical grouping of information as described in the programs that process the information and load the magnetic tapes or data file. The Telephone Company will determine the charges based on the number of data files or tapes prepared and on its count of the records processed. The number of records processed will be determined using the numbered records input to or the number of records output from the programs that process the in-JAN 1 1988 FILLED JAN 1 ASG Bylzt ASG Bylzt Commission July 1 1921 Bylzt Service Commission 89 formation and load the magnetic tapesED data file, whichever number of records is higher.

Issued: DEC 29 1983

Effective: JAN 0 1 1984

83 - 253

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

8.

(CT)

(CT)

ACCESS SERVICES

FEB 2-1 1 BILLING AND COLLECTION SERVICES-(Continued) MISSOURI **Public Service Commission**

Replacing

Access Services Tariff

Section 8

Sheet

- - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - Rate Regulations-(Continued) F.
 - 9. The rates as set forth in Paragraph in 8.2.1, G.7., following, apply for Bill Processing Service for an IC message-billed service depending on the total (i.e., sum of interstate and intrastate IC messages) number of messages billed for an End User's account per month. The rate groups are 1 to 10 messages per month, 11 to 100 messages per month, 101 to 600 messages per month and over 600 messages per month. The rate for the largest number of IC messages billed for the End User's account in a month will be used to determine the bill processing service charges for that End User's account for the month. The Telephone Company will determine the charges for each IC based on its count of IC messages billed each bill day to that IC's End User account.
 - 10. When the Telephone Company receives an order from the IC to issue one or more IC credit card(s) to an End User, Special Order and Credit-Card Issuance preparation and distribution charges apply. If the IC requests another card to be issued for any reason, the preparation and distribution charges apply. If for any reason the IC requests a change to be made in the credit card information maintained on an End User's account by the Telephone Company, including the marking of the account to show issuance of a credit card by the IC or discontinuance of an existing card, and the IC does not request the distribution of a credit card, only the preparation charge applies. The Telephone Company will determine the charges based on its record of the Special Orders to prepare and distribute IC credit cards received from the IC.

CANCELLED JAN 1 1988 BY 15T AS 6 BY 15T AS FILLED MAR 2 8 1984 Public Service Commission Issued: FEB 27 1984 MAR 28 1984 Effective: By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 37

DEC 20 (TO)

R. M. C. A. S. MANDA

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - 9. The rates as set forth in Paragraph in 8.2.1, G.7., following, apply for Bill Processing Service for an IC message-billed service depending on the total (i.e., sum of interstate and intrastate IC messages) number of messages billed for an End User's account per month. The rate groups are 1 to 10 messages per month, 11 to 100 messages per month, 101 to 600 messages per month and over 600 messages per month. The rate for the largest number of IC messages billed for the End User's account in a month will be used to determine the bill processing service charges for that End User's account for the month. The Telephone Company will determine the charges based on its count of IC messages billed each bill day to an End's User's account.
 - 10. When the Telephone Company receives an order from the IC to issue one or more IC credit card(s) to an End User, Special Order and Credit-Card Issuance preparation and distribution charges apply. If the IC requests another card to be issued for any reason, the preparation and distribution charges apply. If for any reason the IC requests a change to be made in the credit-card information maintained on an End User's account by the Telephone Company, including the marking of the account to show issuance of a credit card by the IC or discontinuance of an existing card, and the IC does not request the distribution of a credit card, only the preparation charge applies. The Telephone Company will determine the charges based on its record of the Special Orders to prepare and distribute IC credit cards received from the IC.

GANGELLED 日前に民国 UNI - 1 (SE) 8 3 - 2 5 3 MAR 2 8 1984 stR5#37 PUBLIC SERVICE COMMISSION OF MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(MT)

(MT)

(AT)

(AT)

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)
 - 11. The basic per-hour rate and the premium per-hour rate for program development is for the use of one hour of one Telephone Company programmer's time.
- 12. The Telephone Company will keep a count of the hours and fraction thereof used by Telephone Company personnel to provide program development and will bill the IC in accordance with these records. The hours for each service ordered will be summed and then rounded to the nearest hour, except that when the total is less than one hour, one hour will be used to determine the charges.
 - Bill Page Information per bill rendered charge applies each month 13. that the informational or promotional wording appears on an End User's bill containing one or more messages or related rate elements of the IC.

The Special Order Charge for Bill Page Information applies for the establishment or change of the informational or promotional wording on the End User's bill.

14. The Telephone Company will keep a count of the number of orders from the IC for Investigation of End User message bill charges. Each End User message shown in the IC order where investigation is requested, will be considered a request for which charges, as set forth in Paragraph 8.2.1, G., following, will apply. If the Investigation requested is review of a message to identify suspected unauthorized use of the IC's service and another Investigation is requested for the same End User message, an additional request per message will be counted. The charge per request is in addition to the End User account activity Special Order Charge. The End User account activity Special Order Charge applies for each order where CANCELLED JAIN 1 1988 BY 15T RS6 Investigation is requested.



Access Services Tariff

Original Sheet 37.01

SEP 29 1986

Public Service Commission

Section 8

AUG 29 1800 Issued:

> By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Public Service Commission

Enfective: SEP 29 1000

	tariff wi except fo	ement to this ill be issued or the purpose ling this tariff.				Se ised S	Tariff ction 8 heet 38 heet-38
			ACCESS SERVICES		RE(BEIII	
}	8. BILL	ING AND COLLECTION	SERVICES-(Continued)			IG 2 8	1986
	8.2 Bil	lling Service-(Cont	inued)			MISSU	
	8.2.1	Message Billing Se	rvice-(Continued)		Public S	ervice (Commission
	F. Ra	ate Regulations-(Co		<u> </u>		{	
(MT) (MT)							(
	G.Ra	ates and Charges					
	Tì	he rates and charge	s are:				
				USOC	Ra	tes	
	1	3-year period, p	ng Service er message er message er message	BABM+ BABM+ BABM+	+ .	.011 .009 .007	
	2					.002	
	3	(applicable to within the Tele	work performed phone Company's edule and using the		10	4.00	
•					12	8.00	
		requires additi '	ons to the work force)		淂	ALE	
			CANCELLED		SE	P 2 9 1	986
			ons to the work force) CANCELLED JAN 1 988 JAN 1 056 BY 155 COMM PUBLIC SET SCOVE: MISCOVE: RON, President-Missouri estern Bell Telephone Com	mission	Public Se	ervice Co	mission
•	Issued:	AUG 29 1986 By R D BAR	RON, President-Missouri	SEP 29	1986	_	
		Southwe	stern Bell Telephone Co St. Louis, Missouri	mpany			

ł

j

1

- į

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 38

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.1 Message Billing Service-(Continued)
 - F. Rate Regulations-(Continued)

DEC 20 000

E1114 13.002

JAN = 1 1934 83 - 250

128.00

CANDELLED

SEP 2 9 1986

OF MISSOURI

- The basic per-hour rate and the premium per-hour rate for program 11. development is for the use of one hour of one Telephone Company programmer's time.
- 12. The Telephone Company will keep a count of the hours and fraction thereof used by Telephone Company personnel to provide program development and will bill the IC in accordance with these records. The hours for each service ordered will be summed and then rounded to the nearest hour, except that when the total is less than one hour, one hour will be used to determine the charges.
- G. Rates and Charges

The rates and charges are: -

	USOC	Rates	
 Message Processing Service year period, per message year period, per message year period, per message 	BABM+ BABM+ BABM+	\$.011 .009 .007	

- 2. Additional Message Processing, per message above the message capacity ordered and allowances specified
- 3. Program Development, Basic, per hour (applicable to work performed within the Telephone Company's normal work schedule and using the normal work force)

Premium, per hour (applicable to work performed PUBLIC SERVICE COMMISSION outside the Telephone Company's normal work schedule and/or which requires additions to the work force)

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

tariff wil except for	ent to this l be issued the purpose ng this tariff.		vices Tariff Section 8 nal Sheet 39
	ACCESS SERVICES		
8. BILLIN	G AND COLLECTION SERVICES-(Continued)	D	
8.2 Bill	ing Service-(Continued)		
`8.2.1 M	essage Billing Service-(Continued)		n an an an an an an an an an an an an an
G. Rat	es and Charges-(Continued)		
		FID	Rates
4.	Data transmission of rated IC messages details between other exchange telephone company locations, - per record transmitted	TRMD A TRMD B	\$.003 .0025
5.	Provision of rated IC message detail, - per record processed	PRMD	.003 40.00
. 6.	Data transmission to an IC location of rated IC message details, per record transmitted.	BODDT	ICB rates and charges apply.
	JAN 1 RS C BY 1ST Commission	Rat Messages per Account F to 11 to	End User's Per Month 101 to Over
7.	Bill Processing Service blic Service -	<u>10</u> <u>100</u>	600 600
_	Message-billed processing,		山自日
		255 \$.0215, 21 .0175 16 .0135	\$ 10195 \$.018 .016 .0145 .012 .011
	Message-billed inquiry,		
		17 .117 035 .1035 9 .09	.039 .039 .0345 .0345 .03 .03

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

83-253

wil: for	l be issued the purpose		ices Tariff Section 8 al Sheet 40
~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	ACCESS SERVICES		
	G AND COLLECTION SERVICES-(Continued)	DEC 2	
Bill	ing Service-(Continued)	-	
.1 M	essage Billing Service-(Continued)		
Rat	es and Charges-(Continued)	USOC	Rates
7.	Bill Processing Service-(Continued)		
	Bulk-billed processing,		
	1-year period, per message	BABU+ BABU+ BABU+	\$.003 .0025 .002
	Bulk-billed inquiry,		
	1-year period, per message 3-year period, per message 5-year period, per message	BABY+ BABY+ BABY+	.002 .0015 .001
8.	Additional Bill Processing, per message above the bill capacity ordered and allowance specified,		
	Message-billed processing, each	FILED	.005
	Message-billed inquiry, each	1988	.0135
ı	Bulk-billed processing,	ST RSb St RSb	n .0005
	Bulk-billed inquiry, each	ANISSOLIAI MISSOLIAI	.0005
9.	Message-Billed Service, in which one or more messages or message service related rate elements are billed,		an an an an an an an an an an an an an a
	- per bill rendered for an End User's account		113.00
10.	· · ·	AL I	M - 1 1984
	billed,	‡ satu 8	3 - 2 5 3
	- per bill rendered for an End User's account.		.80
	wil. for celi ILLIN Bill 1 M Rat 7. 8.	 LLING AND COLLECTION SERVICES-(Continued) Billing Service-(Continued) 1 Message Billing Service-(Continued) Rates and Charges-(Continued) 7. Bill Processing Service-(Continued) Bulk-billed processing, year period, per message year period, per message year period, per message year period, per message year period, per message year period, per message year period, per message year period, per message year period, per message year period, per message 8. Additional Bill Processing, per message above the bill capacity ordered and allowance specified, 8. Message-billed processing, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed inquiry, each. message-billed service, in which one or more messages or message service related rate elements are billed, per bill rendered for an End User's account. Sulk-Billed Service, in which a charge associated with a bulk-billed service is 	will be issued for the purpose Origin ACCESS SERVICES LLING AND COLLECTION SERVICES-(Continued) Billing Service-(Continued) DEC 2 Billing Service-(Continued) USOC 1 Message Billing Service-(Continued) USOC 7. Bill Processing Service-(Continued) BaBU+ Bulk-billed processing, BABU+ 3-year period, per message BABU+ S-year period, per message BABU+ Bulk-billed inquiry, L-year period, per message Bulk-billed inquiry, BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year period, per message BABY+ S-year perio

.

.

Issued: DEC 2 9 1983

1

Effective: JAN 0 1 1984

		2nd	s Services Tariff Section 8 Revised Sheet 41 Revised Sheet 41
	ACCESS SERVICE.	s	REGEIVED
8. BILLING	AND COLLECTION SERVICES-(Continue	ed)	AUG 2 8 1986
8.2 Billi	ng Service-(Continued)		MISSOURI
8.2.1 Me	essage Billing Service-(Continued)	Pi	iblic Service Commission
G. Rate	es and Charges-(Continued)	ę. <u></u>	Rates
11.	End User's Account Activity		
(AT)	Special Order Charge to receive account data not involving adjus Charges to End Users	tments of IC	ICB rates and charges apply.
(NR)	Special Order Charge to receive count data to adjust IC charges per End User account	to End Users,	\$ 2.42
	 End User's Account Establishment except Rate Element Rate Level C Rate Structure Charge, per End User's account establi and per recurring or nonrecurr established or changed, each . End User's account Rate Element Change Charge, per rate element changed, each End User's Account Rate Element Change Charge, per rate element changed, each 	hanges and shed or charge EF ing radio dement 	.90
12.		ation of Message ation,	ICB rates and charges apply.
13.	Credit Card Issuance, - preparation, per End User's ac - distribution, per card		\$.75
(MT)	Message Billing Service Special - per Special Order		SEP 29 ^{5,00} SEP 29 ^{5,00}
Issued: AU	G 29 1986 Effe	ective: SEP 29 198	
	By R. D. BARRON, President Southwestern Bell Tel St. Louis, Mi	lephone Company	,

تعشرى فنحد

i

tariff wil except for	ent to this l be issued the purpose ng this tariff.		ervices Tariff Section 8 vised Sheet 41 ginal Sheet 41
	ACCESS SERVICES		REGEIVED
8. BILLIN	G AND COLLECTION SERVICES-(Continued)		
8.2 Bill	ing Service-(Continued)		APR 15 1995
8.2.1 M	essage Billing Service-(Continued)		MISSOURI Public Service Commission
G. Rat	es and Charges-(Continued)		Rates
11.	End User's Account Activity		
	Special Order Charge to receive End account data		ICB rates and charges apply.
	End User's Account Establishment and except Rate Element Rate Level Chang Rate Structure Charge, - per End User's account established and per recurring or nonrecurring	es and or changed	
_	established or changed, each		\$.90
	End User's account Rate Element Rate Change Charge, - per rate element changed, each		ICB rates and charges apply.
	End User's Account Rate Element Rate Change Charge, - per rate element changed, each		ICB rates and charges apply.
12	Data transmission from an IC location Billing Service detail or information - per record received	GANGELLE	ICB rates and charges apply.
13.	 Credit Card Issuance, preparation, per End User's accour distribution, per card 	nt SEP ? 9 1986	\$.75
14	 Message Billing Service Special Orde per Special Order	PUBLICE SEPVICE COMMUNIC	95.00
15	. Retention of Records Under Accountin - per order, per month		ICB rates and charges apply.
(NR) 16	. Other transmission to a customer loc - per tape or data file	-	ICB rates and charges apply.
Issued: APR 1919	Effectiv By R. D. BARRON, President-Mis Southwestern Bell Telepho St. Louis, Missou	ssouri Division one Company	「写印LED」 MAY 2 O 1985 Iic Service Commission

I

3

i

.. _

No supplement to this

Access Services Tariff Section 8

.

tariff will be issued except for the purpose	Section 8 Original Sheet 41
of canceling this tariff.	
ACCESS SERVICES	,
8. BILLING AND COLLECTION SERVICES-(Continued)	DEParton
8. BILLING AND COLLECTION SERVICES-(Continued) 8.2 Billing Service-(Continued) GANGELLED	DEC 2 C (CC)
8.2.1 Message Billing Service-(Continued)	
G. Rates and Charges-(Continued)	ON <u>Rates</u>
11. End User's Account Activity	
Special Order Charge to receive End User's account data	ICB rates and . charges apply
End User's Account Establishment and Change, except Rate Element Rate Level Changes and Rate Structure Charge, - per End User's account established or changed and per recurring or nonrecurring rate elemen	
established or changed, each	
End User's account Rate Element Rate Level	· · ·
Change Charge, - per rate element changed, each	ICB rates and charges apply
End User's Account Rate Element Rate Structure	ICB rates and
Change Charge, - per rate element changed, each	
12. Data transmission from an IC location of Messag	ge
Billing Service detail or information, - per record received	ICB rates and charges apply
	. charges apply
13. Credit Card Issuance,	A 75
 preparation, per End User's account. distribution, per card 	. \$.75 60
14. Message Billing Service Special Order Charge,	11114月到
14. Message Billing Service Special Order Charge, - per Special Order	JAN -1 198
15. Retention of Records Under Accounting Orders, - per order, per month	83-253 ^{ICB} rates and
- per order, per month	-cnarges apply
	1

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

				ess Services Tariff Section 8 riginal Sheet 41.01 REGEIVED
	8. BILLING	S AND COLLECTION SERVICES-(Continued)		NUG 2 8 1986
-	8.2 Billi	ng Service-(Continued)		MISSOURI
	8.2.1 Me	essage Billing Service-(Continued)		Public Service Commission
	G. Rate	es and Charges-(Continued)		Rates
) (M	ſ) 15.	Retention of Records Under Accounting Orders, - per order, per month		ICB rates and . charges apply.
(M	ſ) 16.	Other transmission to a customer location, - per tape or data file		ICB rates and . charges apply.
(N	R) 17.	 Bill Page Information, Special Order Charge, per request per bill rendered 	•••	. \$722.08 04
(N	R) 18.	Investigation of End User Message Bill Charges, - per request		. 2.98

JAN 1 1988 BY Et CS 6 BY DE LOS Public Service Commission MISSOURI

FALED

SEP 29 1986

Public Service Commission

. •

Issued: AUG 29 1986

ł

Effective: SEP 29 1986

Access Services Tariff Section 8 Original Sheet 42

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service

S. Ash Public Service Commission

DEC 2 9 (203)

The Telephone Company will provide Private Line Billing Service only for those IC Private Line Services for which the Telephone Company is providing Special Access Service. The Telephone Company will not render bills for IC Private Line Services for which it does not provide Special Access Services that are connected at End User's premises to IC Private Line Services for which it does render bills.

The Telephone Company will not render bills for usage-based rate elements under this Section of the Tariff. Usage-based rate elements will be billed as set forth in Paragraph 8.2.1, preceding.

A. General Description

Private Line Billing Service includes editing and rating, account establishment, rendering of bills, receiving payments, maintenance of accounts, treatment of accounts and inquiry (when ordered by the IC).

Editing and rating is the examination and identification of all the rateable elements of an IC Private Line Service and the application of the appropriate IC rates and charges to the service. These functions are performed when Private Line Services for an IC's End User are established or changed. Rating is always performed, and conting may be performed, coincident with the implementation of eEdbarge in the IC's schedule of rates.

Account establishment is the preparation of an ICJENU¹User's record so that a bill can be sent to that End User. Rendering of bills is the preparation and mailing GetVice Commission amounts due from End Users for service received from the IC statements may, at Telephone Company's choice, be included as part of the regular monthly bill for Local Telephone Exchange Service mailed to the End User.

Receiving payment and maintenance of accounts is the collecting of deposits and monies from End Users for services furnished by the IC and maintenance of records of all transactions. 83 - 253

Issued: DEC 2 9 1983

JAN 0 1 1984 Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

A. General Description-(Continued)

Treatment of accounts is the forwarding of notices of delinquent or unpaid End User's accounts, posting credits and adjustments, and when necessary as determined by the Telephone Company, denial of the IC's Special Access Service.

Inquiry is the answering of End User's questions about charges for IC services and negotiating of credits and adjustments to End User's accounts and review of IC charges which are removed from an End User's bill.

- Β. Undertaking of the Telephone Company
 - When Private Line Billing Service is ordered by an IC, the Telephone 1. Company will establish a Private Line account, edit and rate the billing detail, bill the End User and maintain and treat the Private Line account (based on the rate and End User's data supplied by the IC) at the rates and charges set forth in Paragraph 8.2.2, G., following, as set forth in Paragraph 8.2.2, B.2. through 14., following. The Telephone Company will not establish a Private Line End User's account with any IC balance due. In addition, the Telephone Company will, in accordance with Telephone Company deposit regulations, determine and collect a deposit from the End User for the IC service. The Telephone Company will, when necessary, in accordance with the Telephone Company deposit regulations, determine and collect the service deposit when an End User's account is established. The Telephone Company will, when necessary, in accordance with Telephone Company deposit regulations, maintain a service deposit for each End User's account. The Telephone Company will provide the IC a copy of its deposit regulations upon request from the IC.

The Telephone Company will make adjustments to End User's balances due to account for application of creeds authorized by IC-furnished statements. JAN 1 1988 BY 121 RS6

Issued: DEC 2 9 1983

BY 121 Commission Public Service Commission ANSSOURI EF JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

DEC 2	3 9 4000
-------	-----------------

Public S

Access Services Tariff

Original Sheet 43

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 44

DEC 20 (203

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.2 Private Line Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 2. Private Line Billing Service records and End User's accounts will be maintained by the Telephone Company in a standard format in order to identify the End User and bill the rateable elements. The Telephone Company will establish this format and provide it to the IC. The Telephone Company will also establish the format it will use to bill Private Line Services and provide it to the IC. If, in the course of Telephone Company business, it is necessary to change these formats, the Telephone Company will notify the involved IC's six months prior to the change.
 - 3. The Telephone Company will develop the IC's schedule of rates and charges into a rating program. Program development charges as set forth in Paragraph 8.2.2, G.3., following, apply for the hours re-• quired to design, develop, test and maintain the necessary programs.
 - 4. Upon acceptance by the Telephone Company of a Special Order for Private Line Billing Service from an IC, the Telephone Company will determine the period of time to implement such service on an individual-case basis.
 - 5. Changes in the rate levels of IC services to be billed will normally be implemented within 60 days after receipt of a Special Order from the IC requesting such changes. Such changes will require modifications of the rating program. Program development charges as set forth in Paragraph 8.2.2, G.3., following, apply for the hours required to design, develop, test and maintain the necessary program changes.
 - 6. Changes in the rate structure of the IC services to be billed also require a change in the rating programs. When the Telephone Company determines that it can accommodate the changes, the conditions and the period of time required to make such changes will be determined on an individual-case basis. Program revelopment charges 25 set forth in Paragraph 8.2.2, G.3., following, apply for the hours re-quired to design, develop, test and maintain the necessary program changes. CAIN 1 1988 JAN 1 ASG Commission BY ECCOMMISSION BY ECCOMMISSI

JAN 0 1 1984

Issued:

DEC 2 9 1983

Access Services Tariff Section 8 Original Sheet 45

DEC 21 (13)

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.2 Private Line Billing Service-(Continued)
 - B. Undertaking of the Telephone Company-(Continued)
 - 7. The Telephone Company will not provide any information related to Private Line Billing Service accounts under this Section of the Tariff. Private Line Billing Service information may be obtained as set forth in Paragraph 8.4, following.
 - The Telephone Company will, at the option of the IC, provide Private 8. Line Billing with or without inquiry. When the Telephone Company provides Private Line Billing with inquiry, the Telephone Company will be responsible for contacts and arrangements with the IC's End Users concerning the billing, collecting, crediting and adjusting of the IC service charges, except prior IC balances due from End Users in accordance with written instructions furnished by the IC. At the request of the IC, the billed IC charges which are removed from the End User's bill in accordance with the IC inquiry instructions will be reviewed for unauthorized use of the IC service by Telephone Company message investigation groups for a period of up to 90 days after the billed IC charges have been removed from the End User's bill. For any billed IC charges which are removed from an End User's balance due in accordance with the IC's instructions, the Telephone Company will make an appropriate adjustment to the IC's accounts receivable. When the Telephone Company provides Private Line Billing without inquiry, all contacts from IC End Users concerning the IC's billed amounts will be referred to the IC, and the Telephone Company will only be responsible for contacts with IC's End Users concerning the collection of IC service charges, except prior IC balances (The from End Users. Inquiry will only be provided for those End User's accounts for which the IC has ordered JAN 1 157056 Private Line Billing.
 - 9. When the Telephone Company encounters pitts which are to be rendered to End Users or End User's billing addresses not located in the Telephone Company's operating tennitory of in a state where Private Line Billing Service has not been ordered, such bills will be handled as follows:
 - a. If the bill to the End User is for a service for which the Telephone Company provides a Special Access Service to the IC and the IC has ordered the appropriate Private Line Billing Service, the Telephone Company will bill the End User.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 46

DEC 20 (20)

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

- B. Undertaking of the Telephone Company-(Continued)
 - 9. (Continued)
 - b. In all other situations, the bill will be delivered to the IC, and the IC shall be responsible to furnish an accounts receivable adjustment to the Telephone Company as set forth in Paragraph 8.2.3, following.
 - 10. The Telephone Company will accept IC gift certificates for payment from End Users if the IC agrees in writing to redeem all such gift certificates. The format of the gift certificates must be acceptable to the Telephone Company.
 - 11. Where the rates for IC services have been implemented under an accounting order pending final approval from a regulatory agency, The Telephone Company will, upon written request from the IC, keep such records as may be required to make any adjustments to the End Users as may be ordered by the regulatory agency. The charges for such a service will be determined on an individual-case basis.
 - 12. The Telephone Company will provide Private Line Billing Services under a Special Order. For all Private Line Billing Services other than establishment of or changes to End User's account data, establishment of or changes to End User's account rate elements and changes to End User's balance due, the Private Line Billing Service Special Order charge as set forth in Haragraph 8.2.2, G.5., following, will apply to orders accepted by the Telephone Company. The format of this Special Order will be specified by the Telephone Company.
 - 13. The Telephone Company will arrange with the Telephone Company.
 13. The Telephone Company will arrange with the Telephone Company in the Telephone Company is account information; establish and change?
 End User's account data, establish and change? End User's account data, establish and change?
 rate element and change End User's balances due. The methods, procedures and manner in which the End User's accounts data and changes are forwarded to the Telephone Company must be agreeable to the Telephone Company. The charges to handle such Special Orders will be determined on an individual-case basis.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 47

DEC 20 JUL

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

- B. Undertaking of the Telephone Company-(Continued)
 - 14. If the IC requests the private line bills be reprocessed by the Telephone Company because of an IC error, the Telephone Company will treat the reprocessing as a rate level or rate structure change. Determination of whether the reprocessing is a rate level change or a rate structure change will be made by the Telephone Company based on the Special Order the Telephone Company receives from the IC. All appropriate charges as set forth in Paragraph 8.2.2, G., following, will apply.
- C. Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, the Telephone Company's liability for Private Line Billing Service is as follows:

- 1. If Private Line Billing Service detail is not available because the Telephone Company lost or damaged records or incurred processing system outages, the Telephone Company will recover the lost detail based on previously received information. This recovered detail will be provided to the IC if the IC has ordered the appropriate Billing Information Service as set forth in Paragraph 8.4, following. If the detail cannot be recovered, the extent of the Telephone Company's liability for damages will be the known amount not billed for the mount not billed is unknown, no more than three months' charges for the services not billed.

Issued: DEC 2 9 1983

Access Services Tariff

Original Sheet 48

Section 8

DEC 20 CCC

eej0n

No supplement to this. tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

Obligations of the IC

- 8.2.2 Private Line Billing Service-(Continued)
- С. Liability of the Telephone Company-(Continued)
 - If the Telephone Company finds, or is notified of, an error in billing 3. to an End User, it will make a reasonable effort to correct the error and bill the appropriate End User within the limits permitted by laws of the state in which it provides the service. If the error is caused by the Telephone Company and the Telephone Company cannot timely bill the proper End User, the extent of the Telephone Company's liability for damages will be the known amount misbilled or when the amount misbilled is unknown, no more than three months' charges for the services misbilled.
 - 4. In the absence of willful misconduct, no liability for damages to the IC or other person or entity other than that as set forth in Paragraphs 8.2.2, C.1., 2. and 3., preceding, shall attach to the Telephone Company for its action or the conduct of its employeds in providing Private Line Billing Service.

D.

JAN 1 1988 AS 6 The IC shall order Private Line Billing Service under a Special Order for each state where service is desired. When Private Line Billing Service is ordered inftally, the IC shall 1.

order the service for at least one year. Thereafter, upon six months' written notice, additional service may be ordered for a minimum of one year, and the rates and charges as set forth in Paragraph 8.2.2, G., following, will apply. Not later than six months prior to the end of an order period, the IC shall notify the Telephone Company in writing if service is to be discontinued at the end of the period.

If no notice is received from the IC, the Telephone Company will automatically extend the service for an additional year, using the most recent 12 months of bill capacity provided. All appropriate rates and charges as set forth in Paragraph 8.2.2, G., following, for another year will apply, and the minimum monthly charges will be based on the most recent 12 months of bill capacity provided. The IC will-be notified by the Telephone Company when such an extension is made. 83-253

2. When Private Line Billing Service is ordered, the IC shall furnish the Telephone Company for a state and for each year in the order an estimate of the average number of bills (bill capacity) to be rendered each year.

DEC 2 9 1983 Issued:

Effective: JAN 0 1 1984

Access Services Tariff

Original Sheet 49

DEC 2

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

- D. Obligations of the IC-(Continued)
 - 3. The IC shall furnish in the format specified by the Telephone Company all information necessary for the Telephone Company to provide the Private Line Billing Service, including an affidavit that states whether the IC Private Line Billing Service is subject to any federal and/or state taxes. When IC bills are to be rendered to locations not in the operating territory of the Telephone Company, the IC shall furnish an address where the bills are to be sent. If the IC does not furnish an address, all unaddressable bills will be delivered to the IC. The information shall be furnished by the IC in a timely manner.
 - 4. The IC shall furnish to the Telephone Company a written schedule of its service rates and charges in sufficient time to allow the Telephone Company to establish a rating program. The interval required to establish a rating program must be mutually agreed to by the Telephone Company and the IC.
 - 5. When the IC orders Private Line Billing Service, it shall approrize the Telephone Company in writing to deny service to End Viers for nonpayment. If that authorization is not received. Norivate Line Billing Service will not be provided.
 6. The IC chall is
 - The IC shall be responsible for all contacts and arrangements, infillding prior IC balances due from End Users, with its and Users Oconcerning the provision and maintenance of the IC's services. Service public Services Services and Services Services and Services
 - 7. When the IC orders Private Line Billing Service with inquiry, the IC shall furnish the Telephone Company written instructions which are agreeable to the Telephone Company for the handling of End User's guestions about bills.

83-253 When the IC orders Private Line Billing Service without inquiry, the IC shall furnish the Telephone Company with written instructions as _ to where inquiries are to be referred. When the IC does not order Telephone Company Inquiry Service and desires credit adjustments be made to the balances due from an End User, the IC shall furnish a statement for each End User's account where the credit is desired. These statements shall show the rate element to be credited, the date the rate element was billed and the amount of the credit.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 50

DEC 20 LL

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

- D. Obligations of the IC-(Continued)
 - 7. (Continued)

The IC shall notify its End Users through its tariff or other appropriate means when the IC handles the bill inquiries. The IC shall furnish the Telephone Company in writing all bill adjustments as set forth in Paragraph 8.2.3, following.

- 8. The IC will immediately redeem all IC gift certificates that the Telephone Company receives in payment for End User's charges. The IC agrees to use a gift certificate format which is agreeable to the Telephone Company.
- 9. The IC agrees to permit the Telephone Company to, when necessary, in accordance with Telephone Company deposit regulations, determine and collect IC service deposits from all End Users of the IC's services for which the Telephone Company provides billing for the IC. The IC will notify its End Users through its tariffs or other means that the Telephone Company will, when necessary, in accordance with Telephone Company deposit regulations, determine and collect IC service deposits. The IC will also include in its tariff and service arrangements and obtain regulatory concurrence for the Telephone Company deposit regulations that the prephone Company will use to collect End User's service deposite
- Ε.

Payment Arrangements and Audit Provisions⁹⁸⁸ 1. Audit Provisions Upon reasonable written notice SP, the IC to the Telephone Company, the IC shall have the right through its authorized representative to examine and audit, during normal business hours and at reasonable intervals as determined by the Telephone Company, all such records and accounts as may under recognized accounting practices contain information bearing upon the amount payable to the IC. Adjustment shall be made by the proper party to compensate for any errors or omissions disclosed by such examination or audit. Neither such right to examine and audit nor the right to receive such adjustment

Issued: DEC 2 9 1983

JAN 0 1 1931 Effective:

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 51

DEC 2

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
 - 8.2 Billing Service-(Continued)
 - 8.2.2 Private Line Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provisions-(Continued)
 - 1. (Continued)

shall be affected by any statement to the contrary, appearing on checks or otherwise, unless such statement expressly waiving such right appears in a letter signed by the authorized representative of the party having such right and delivered to the other party.

All information received or reviewed by the IC or its authorized representative is to be considered confidential and not to be distributed, provided or disclosed in any form to anyone not involved in the audit, nor is such information to be used for any other purpose.

2. Minimum Periods

The minimum period for which Private Line Billing Service is provided and for which charges apply is one year.

A minimum period of one year applies for each additional period of service ordered.

When service is discontinued prior to the expiration of a minimum period, the minimum monthly charge is applicable for each month and fraction of month remaining in the minimum period.

3. Minimum Monthly Charges

During the initial-year minimum period, there is a minimum montha. ly charge based on the IC's estimate of the number of bills to be rendered during that period. The minimum monthly charge is the charge for Bill Rendering as set forth in Paragraph 8.2.2, G.2., following, for 75 percent of the monthly volume. The monthly volume is 1/12 of the bill capacity ordered during the initial year. CANCELLED ordered during the initial (2, 1)JAIN 1 1988 BY 121 - 1988

DEC 2 9 1983 Issued:

Public Service Commissio JAN 0 1 1984

83 - 253

Access Services Tariff Section 8 Original Sheet 52

DEC 20 1000

i esseuri Peter Service Commission

JAN-1 (SO)

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
 - 8.2.2 Private Line Billing Service-(Continued)
 - E. Payment Arrangements and Audit Provisions-(Continued)
 - 3. Minimum Monthly Charges-(Continued)
 - a. (Continued)

If the actual monthly volume during any consecutive three month period exceeds 1/12 of the bill capacity ordered by 50 percent or more per month, a new minimum monthly charge will be established. The new charge is the charge for Bill Rendering as set forth in Paragraph 8.2.2, G.2., following, for the monthly average of the actual volume used during those three months.

b. During each additional yearly minimum period, the minimum monthly charge is the charge for Bill Rendering as set forth in Paragraph 8.2.2, G.2., following, for the monthly average of the actual volume of bills rendered during the previous 12 months.

If the actual monthly volume during any consecutive three-month period exceeds the monthly average by 50 percent or more per month, a new minimum monthly charge will be established. The new charge is the charge for Bill Rendering as set forth in Paragraph 8.2.2, G.2., following, for the monthly average of the actual volume used during those three months.

4. Cancellation of a Special Order

83-253 An IC may cancel a Special Order for Private Line Billing Service on any date prior to the service date. The cancellation date is, the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for Private Line Billing Service is the date that the IC and the Telephone Company mutually agree service is to start.

When an IC cancels a Special Order for Private Line Billing Service after the order date but prior to the start of service, a charge equal to the Special Order Charges, program development costs and any nonrecoverable capital costs computed by the Telephone Company will apply to the IC.

Issued: DEC 2.9 1983

BY R. D. P.B.M.ROSOFVICE Commission By R. D. P.B.M.ROSOFVICE President-Missouri Southwestern MeSP Telephone Company St. Louis, Missouri

₽.	. S	. C	. Mo	No	. 36
----	-----	-----	------	----	------

Access Services Tariff Section 8 Original Sheet 53

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

DEC 201000

. Commission i

- 8.2.2 Private Line Billing Service-(Continued)
- E. Payment Arrangements and Audit Provisions-(Continued)
 - 5. Changes to Special Orders

When an IC requests changes to a pending Special Order for Private Line Billing Service, such changes will be undertaken if they can be accommodated by the Telephone Company. A charge equal to any costs incurred by the Telephone Company because of the change will apply.

- F. Rate Regulations
 - The Private Line Billing Service Special Order Charge applies for each Special Order for Private Line Billing Service, other than establishment of or change to End User's account data, establishment of or changes to End User's account rate element or changes to End User's balances due, accepted by the Telephone Company.
 - 2. The End User's account activity charge applies whenever an End User's account is established or changed pursuant to a Special Order from an IC and/or whenever a nonrecurring or recurring IC rate element or balance due is added to or changed in the account.

An End User's account is a record for a Private Line Billing Service which has a unique name and address and billing number identification, assigned by the Telephone Company, to which a bill is rendered.

The End User's account activity Special Order Charge always applies, and the End User's account establishment and change charge, End User's account rate element rate level change charge or the End User's account rate element rate structure change charge may apply depending on the activity ordered by the IC as set forth id Paragraphs 8.2.2, F.2.a., b., c. and d., following: **83-253**

a. The End User's account activity Special Order Charge applies whenever the IC furnishes to the Telephone Company End User's account information that establishes or changes the information or rates or balances due associated with an End User's account.

Issued: DEC 2 9 1983

Access Services Tariff Section 8 Original Sheet 54

DEC 20

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.2 Private Line Billing Service-(Continued)
- F. Rate Regulations-(Continued)
 - 2. (Continued)
 - The End User's account establishment and change charges applies Ъ. whenever IC-furnished information is used by the Telephone Company to establish or change End User's account data or rate elements or balance due, except for information to change End User's account rate element rate levels or rate structure. End User's account rate element rate level and rate structure changes charges are applied as set forth in Paragraphs 8.2.2, F.2.c. and d., following.

The End User's account establishment and change charge applies for each account established, rate element established, account changed and rate element changed and balance due changed.

- c. The End User's account rate element rate level change charge applies whenever IC-furnished information is used by the Telephone Company to change an End User's account rate element rate level. The charge applies for each End User's account rate element rate level changed.
- d. The End User's account rate element rate structure change charge applies whenever IC-furnished information is used by the Telephone Company to change an End User's account rate element rate structure. The charges to make the End User's account rate element rate structure changes will be determined on an individual-case basis.
- 3. The bill rendering charge applies each month that one or more charges is billed by the issuing of a statement to an End User's account. When both interstate and state Private Line Service charges are billed by the Telephone Company to the End User on the same bill for the TC, the bill rendering charge change times 0.5 applies each month $h_{1/2}$
- 4. The basic per-hour rate and the premin per-hour rate for program development is for the use of one Lour of one Telephone & Smpan 53 programmer's time.

T NUT

Issued: DEC 2 9 1983

3 By R. D. BARRON DUVICE PRESIdent-Missouri Southwaston Pall Tall Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 55

DEC 20 MCC

USOC

Rates

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.2 Private Line Billing Service-(Continued)

F. Rate Regulations-(Continued)

5. The Telephone Company will keep a count of the hours and fraction thereof used by Telephone Company personnel to provide program development and will bill the IC in accordance with these records. The hours for each order will be summed and then rounded to the nearest hour, except that when the total is less than an hour, one hour will be used to determine the charges.

G. Rates and Charges

The rates and charges are:

1. End User's Account Activity - Special Order to receive End User's account data, ICB rates and per Special Order. charges apply - End User's Account establishment CANCELLED and Change, except rate element rate level changes and rate JAN 1 1988 structure changes, charge BY Zul RS per End User's account established Public Service Commission or changed and per recurring and nonrecurring rate element established or changed, each. . . \$.90 - End User's Account Rate Element Rate Level Change Charge, ICB rates and per rate element changed, each charges apply - End User's Account Rate Element Rate structure Change Charge, ICB rates and per rate element changed, each charges apply 83 - 253 2. Bill Rendering Charge, per bill rendered for an BABPR \$.80

Issued: DEC 2 9 1983

Effective: JAN 0 1 1934

	No supplement to this tariff will be issued except for the purpose of canceling this tariff.	Access Services Tariff Section 8 Original Sheet 56
	ACCESS SERVICES	
	8. BILLING AND COLLECTION SERVICES-(Continued)	
	8.2 Billing Service-(Continued)	DEC 2
	8.2.2 Private Line Billing Service-(Continued)	
	G. Rates and Charges-(Continued)	Public Service Learninstich
	The rates and charges are:-(Continued)	USOC Rates
	3. Program Development Charge	
	- Basic, per hour	\$104.00
	 Premium, per hour	B B B B B B B B B B B B B B B B B B B
	4. Inquiry, per bill rendered for an End User's account	BABPQ .09
,	5. Private Line Billing Special Service Order Charge, per Special Order	
)	6. Retention of Records Under Accounting Orders, per order, per month	UNN-1 (SC) 83-253 CB Fates and charges apply.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

. . . . that

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 1st Revised Sheet 57 Replacing Original Sheet 57

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable
 - The Telephone Company will purchase from the IC its accounts receivable that arise from bills rendered by the Telephone Company to that IC's End Users. For IC's subscribing to Message Processing Service, the purchase will be handled as described in (A) following. When an IC does not subscribe to Message Processing Service but does subscribe to Bill Processing Service, the Telephone Company may purchase accounts receivable from an IC who provides the Telephone Company with files of rated messages as described in (B) following. The purchase of accounts receivable will be limited to amounts due the IC when the Telephone Company provides Bill Processing Service and/or Private Line Billing Service for that IC. After an IC orders Bill Processing Service and/or Private Line Billing Service and the Telephone Company is purchasing the IC's accounts receivable, the IC is prohibited from assigning, transferring, selling, exchanging or giving these accounts receivable to any other entity or person. The IC will provide a written assurance to the Telephone Company as such forbearance and any such assignment, transfer, sale, exchange or gift is null and void and will subject the IC to all liabilities, expenses, costs including attorney fees expended and incurred by the Telephone Company in pursuing exclusive ownership to accounts receivable.

The Telephone Company's purchase of an IC's accounts receivable shall be with recourse adjustments as set forth in Paragraph 8.2.3, A.2., and B.2., following, to account for amounts due the IC that the Telephone Company is unable to collect from the End Users which use the IC's services. The amounts due the IC for the purchase of its accounts receivable will be determined as follows:

(AT) A.

Amounts due to IC's subscribing to Message Processing an EP11 Processing For IC's subscribing to Message Processing Security, the amount due for the purchase of its accounts receivable will be determined by the Tele-phone Company as follows: ORSZ phone Company as follows:

Total Current Amount Billed - The Telephone Company for each End User's bill day (i.e., the date on the bill gendered, to an End Use for an IC's service) will determine from its records the the total 1. for an IC's service) will determine from its records the the LEU current amount billed to the IC's End Users for IC services, including all taxes applicable to such services. A Total Current 1986 Amount Billed will be determined for each IC for each End User's bill day.

Public Service Commission

Issued:

AUG 2 6 1986

SEP 26 1986 Effective:

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri.

AUG 25 1986

MISSUURI

(AT)

(FC)

(RT)

(MT)

(AT)

Access Services Tariff Section 8 Original Sheet 57

DEC 20 (LC)

Public Service Commission

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.2 Billing Service-(Continued)

8.2.3 Purchase of Accounts Receivable

The Telephone Company will purchase from the IC its accounts receivable that arise from bills rendered by the Telephone Company to that IC's End Users. The purchase of accounts receivable will be limited to amounts due the IC when the Telephone Company provides Bill Processing Service and/or Private Line Billing Service for that IC. After an IC orders Bill Processing Service and/or Private Line Billing Service and the Telephone Company is purchasing the IC's accounts receivable, the IC is prohibited from assigning, transferring, selling, exchanging or giving these accounts receivable to any other entity or person. The IC will provide a written assurance to the Telephone Company as such forbearance and any such assignment, transfer, sale, exchange or gift is null and void and will subject the IC to all liabilities, expenses, costs including attorney fees expended and incurred by the Telephone Company in pursuing exclusive ownership to accounts receivable.

The Telephone Company's purchase of an IC's accounts receivable shall be with recourse adjustments as set forth in Paragraph 8.2.3, B., following, to account for amounts due the IC that the Telephone Company is unable to collect from the End Users which use the IC's services. The amounts due the IC for the purchase of its accounts receivable will be determined as follows:

- A. <u>Total Current Amount Billed</u> The Telephone Company for each End User's bill day (i.e., the date on the bill rendered to an End User for an IC's service) will determine from its records the total current amount lawfully billed to the IC's End Users for IC services, including all taxes applicable to such services. A Total Current Amount Billed will be determined for each IC for each End User's bill day.
- B. <u>Recourse Adjustments</u> For rach bill day E the Telephone Company will-make recourse adjustments to the Total Current Amount Billed as follows:
 - 1. End User's Adjustments SEP 26 1986 For each bill day, the Telephone, tompany will subtract of the Total Current Amount Billed the LawFulston billed amounts which the Telephone Company removes woron MEnd User's balances due in

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

.

	tarif	fwil	l be	to this issued					ces Tariff Section 8	
				purpose his tariff.					d Sheet 58 1 Sheet 58	
					ACCES	S SERVICES	1	RE	GEIVED	
	8. B	ILLIN	G AN	D COLLECTIO	N SERVICES-(Continued)				
-	8.2	Bill	ing	Service-(Con	ntinued)			UΑ	6 2 5 1986	
	8.2	.3 P	urch	ase of Acco	unts Receiva	ble-(Continued	1)		WISSOURI ervice Commissio	on
(AT)	A.			due to IC's ued)	s subscribin	g to Message l	Processing	and-Bill	Processing-	
(MT) (FC)		2.	mak			each bill day to the Total (
			a.	End User's	Adjustments					
(RT)				the Total (Telephone (accordance each bill (Total Curre value of an gift certin The IC gift	Current Amou Company remo with IC inq day, the Tel ent Amount B ny IC gift c ficates) the	Telephone Con nt Billed the ves from End W uiry instruct ephone Company illed, an amou ertificates (o Telephone Cor es the Telepho	billed amo User's bala ions. In a y will subt unt that eq or its pred npany has i	unts whic nces due ddition, ract from uals the ecessor c n its pos	h the in for the face company's session.	
			b.	Telephone (Company and	IC Adjustments	S			
•				the Total (bills which addition, : adjustment: for amounts or subtract	Current Amou h the Teleph for each bil s to the Tot s on stateme	Telephone Con nt Billed bill one Company de l day, the Tel al Current Ame nts received : End User's bal s.	l amounts f elívers to lephone Com ount Billed from the IC	or End Us the IC. pany may to accou for addi	er's In make int tions	
				ments to t	he Total Cur	he Telephone (rent Amount B: ns for IC or CANCELLY	illed to ac	count for		
(MT)						JANI	Smission	se Public Se	P 26 1986 Prvice Commission	Un
	Issue	d:	AUG	26 1986		BY Orvice By Orvice Public Sorvice Public Sorvice Provide Sorvice	SEP 26 1	<u>98p</u>		
			В			elephone Comp	ivision			-

1

Access Services Tariff

DEC 20 (LU)

Public Service Commission

Original Sheet 58

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable-(Continued)
 - B. Recourse Adjustments-(Continued)
 - 1. End User's Adjustments-(Continued)

accordance with IC inquiry instructions. In addition, for each bill day, the Telephone Company will subtract from the Total Current Amount Billed, an amount that equals the face value of any IC gift certificates (or its predecessor company's gift certificates) the Telephone Company has in its possession. The IC gift certificates the Telephone Company possesses will be returned to the IC.

2. Telephone Company and IC Adjustments

For each bill day, the Telephone Company will subtract from the Total Current Amount Billed bill amounts for End User's bills which the Telephone Company delivers to the IC. In addition, for each bill day, the Telephone Company may make adjustments to the Total Current Amount Billed to account for amounts on statements received from the IC for additions or subtractions to an End User's balance due for services billed in prior periods.

Also, each bill day, the Telephone Company may make adjustments to the Total Current Amount Billed to account for additions and subtractions for IC or Telephone Company prior period errors.

3. Uncollectible Adjustments

For each bill day, the Telephone Company will subtract from the Total Current Amount Billed an amount for uncollectibles. Uncollectibles are amounts billed by the Telephone Company to End Users on final customer bills that are added to the uncollectible (realized) accounts of the Telephone Company. The Telephone Company will determine the IC amount for uncollectibles for each bill day by multiplying the Total Current Amount Billed by the IC uncollectible factor rounded up to the nearest 1/1000 as determined in Paragraphs 8.2.3, B.3.a. or b., following. UNI-160 83-253

SEP 2 6 1986

PUBLIC SERVICE COMMISSION MISSOURI

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984



No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 59

DEC 20 Kulu

UAN = 1 (93.) 8 3 - 2 5 3

able Service Committee

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)
 - B. <u>Recourse Adjustments</u>-(Continued)
 - 3. Uncollectible Adjustments-(Continued)
 - To determine the IC Uncollectible factor, except for the initial a. three months that Bill Processing Services and/or Private Line Billing Service is provided to the IC, The Telephone Company will determine from its records the dollar amount lawfully billed on final customer bills which, after standard collection efforts is completed, is added to its uncollectible (realized) accounts (uncollectible amount) for bills rendered to End Users in the most recent three-month period. This uncollectible amount will include adjustments to account for any payments received by the Telephone Company for outstanding final customer bill amounts that pre-date the most recent three-month period and any deposits held by the Telephone Company for services provided to the End Users where final customer bills have been rendered. The uncollectible amount (including, where necessary, the IC's or its predecessor company's history of uncollectible to develop a full recent three-month period) and will be used by the Telephone Company in an uncollectible apportionment study to determine the realized uncollectible amount for each IC which is provided Bill Processing Service and/or Private Line Billing Service by the Telephone Company for the most recent three-month period. The realized uncollectible amount for an IC will, after the adjustment to account for IC amounts for uncollectibles for the previous three-month period which were greater or lesser than the realized uncollectible amount for the same three-month period, be divided by the Total Current Amount Billed for the IC for the same most recent three-month period to develop an IC uncollectible factor. This factor will be used by the Telephone Company for the next three months to determine the IC amount for uncollectibles. Just prior to the end of the three-month period, the Telephone Company will determine a new IC uncollectible factor in the same manner as above for the ensuing three-



Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this Access Services Tariff tariff will be issued Section 8 except for the purpose 1st Revised Sheet 60 of canceling this tariff. Replacing REG E ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) AUG 2 5 1986 8.2 Billing Service-(Continued) MISSOURI 8.2.3 Purchase of Accounts Receivable-(Continued) Public Service Commission (AT)Α. Amounts due to IC's subscribing to Message Processing and Bill Processing-(Continued) (FC) 2. Recourse Adjustments-(Continued) (MT) Uncollectible Adjustments-(Continued) с. (1) (Continued) for the most recent three-month period. The realized uncollectible amount for an IC will, after the adjustment to account for IC amounts for uncollectibles for the previous three-month period which were greater or lesser than the realized uncollectible amount for the same three-month period, be divided by the Total Current Amount Billed for the IC for the same most recent threemonth period to develop an IC uncollectible factor. This factor will be used by the Telephone Company for the next three months to determine the IC amount for uncollectibles.

Just prior to the end of the three-month period, the Telephone Company will determine a new IC uncollectible factor in the same manner as above for the ensuing three-month period.

(2) When an IC orders Bill Processing Service and/or Private Line Billing Service, the IC at the time that each service is ordered shall provide to the Telephone Company a history of its or its predecessor company's uncollectibles. This history of uncollectibles shall indicate by month its total amounts billed and its uncollectible amounts. The Telephone Company will use these data to develop the IC uncollectible factor for the first three months. To the extent that such IC's or its predecessor company's data do not exist, then the IC uncollectible factor for the first three month period will be determined on an individual-case basis. The Market collectible factor developed either from IC history of on an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period of an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period the first three-month for the first three-month period an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period the first three-month period the first three-month period an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period the first three-month period the first three-month period an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three-month period the first three the first three the first three the first three the first three the first three the first th

(MT) Public Service Commission Hemmiser 26 1986 Issued: AUG 2 6 1986 BYZ By R. D. BARRON, President Messeri Division Southwestern Belly Pelephone Company St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 60

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

DEC 25 (TT)

í,

自動能という。

- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable-(Continued)
- B. Recourse Adjustments-(Continued)
 - 3. Uncollectible Adjustments-(Continued)
 - Ъ. When an IC orders Bill Processing Service and/or Private Line Billing Service, the IC at the time that each service is ordered shall provide to the Telephone Company a history of its or its predecessor company's uncollectibles. This history of uncollectibles shall indicate by month its total amounts billed and its uncollectible amounts. The Telephone Company will use these data to develop the IC uncollectible factor for the first three months. To the extent that such IC's or its predecessor company's data do not exist, then the IC uncollectible factor for the first threemonth period will be determined on an individual-case basis. The IC uncollectible factor developed either from IC history or on an individual-case basis will be used to determine the IC amount for uncollectibles for the first three-month period.
- С. Payments of Net Purchase Amount to the IC
 - The Telephone Company will purchase accounts receivable from the IC 1. on each End User's bill day for an amount (purchase amount) which equals the Total Current Amount Billed as set forth in Paragraph 8.2.3, A., preceding, after known adjustments as set forth in Paragraph 8.2.3, B., preceding, have been made. On the date (payment date) determined by adding 31 days to the End User's bill day or on the next bill date (i.e., same date in the following month as the bill date), whichever is the shortest interval, except as provided herein, the Telephone Company will remit payment to the IC for the purchase amount less additional adjustments as set forth in Paragraphs 8.2.3, B.1. and 2., preceding (net purchase amount) received by the Telephone Company prior to the payment date. The net purchase amount in excess of one hundred thousand dollars (\$100,000) shall be transmitted on the payment date to a designated bank account by electronic funds transfer in immediately available funds. Any amount mot required to be transmitted by elec-tronic funds transfer shall be paid by check or draft and must be, postmarked four business days prior to the payment date 3 If such payment date would cause payment to be due on a Saturday, Sunday or holiday (i.St, New Year's Day, Independence Day, Labor Day,

Issued: DEC 29 1983 BY 1 At 1. S. H. LOU PUBLIC SERVICE COMMISSION Effective: JAN 0 1 1984 By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company

<u>(0)</u>

St. Louis, Missouri
9.3.0. del-Tal 16

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 1st Revised Sheet-61 Replacing origination

ACCESS SERVICES

8.2 Billing Service-(Continued)

8.2.3 Purchase of Accounts Receivable-(Continued)

(AT) Amounts due to IC's subscribing to Message Processing and Bill Processing-A. (Continued)

(FC)

(AT)

(AT)

(CT)

(CT)

íT)

Payments of Net Purchase Amount to the IC 3.

- The Telephone Company will purchase accounts receivable from the IC on each End User's bill day for an amount (purchase amount) which equals the Total Current Amount Billed as set forth in Paragraph 8.2.3, A.1., preceding, after known adjustments as set forth in Paragraph 8.2.3, A.2., preceding, have been made. On the date (payment date) determined by adding 31 days to the End User's bill day or on the next bill date (i.e., same date in the following month as the bill date), whichever is the shortest interval, except as provided herein, the Telephone Company will remit payment to the IC for the purchase amount less additional adjustments as set forth in Paragraphs 8.2.3, A.2.a., and 2.b., preceding (net purchase amount) received by the Telephone Company prior to the payment date. The net purchase amount in excess of one hundred thousand dollars (\$100,000) shall be transmitted on the payment date to a designated bank account by electronic funds transfer in immediately available funds. Any amount not required to be transmitted by electronic funds transfer shall be paid by check or draft and must be postmarked four business days prior to the payment date. If such payment date would cause payment to be due on a Saturday, Sunday or Holiday (i.e., New Year's Day, Independence Day, Labor Day, Thanksgiving Day, Christmas Day, the second Tuesday in November and a Day is legally observed), payment for the net purchashield and a mount will be due to the IC on the first net Will be due to the Kee Will be due to the Kee Will be due to the Kee Will be due to the Kee Will be due to the Kee Will be due to the Kee Will be due to the Kee Will be due to the K JAN 1 following such Saturday, Sunday or Holiday.
- b. Further, if any portion of the net purchase amount is not commission received by the IC by the payment date as set forth is prvice COMMISSION Paragraph 8.2.3, A.3.a., preceding, or if any portion of the COMMISSION net purchase amount is received by the IC is the transformer of the COMMISSION net purchase amount is received by the IC in funds which are not immediately available to the IC, then a law the payment penalty shall be due the IC. The late-payment penalty shall be the portion of the net purchase amount not received 2 by 1986 the payment date times a late factor. The late factor shall be the lesser of:

Public Service Commission

Issued: AUG 2 6 1986 Effective: SEP 26 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

AUG 2.5 1986

MISSUURI Public Service Commission

(MT)

Access Services Tariff Section 8 Original Sheet 61

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)

GANGEL

SEP 2 6 1986

PUBLIC SERVICE COMMISSION

相關的意志

居住的

DEC 26 (CC)

- C. Payments of Net Purchase Amount to the IC-(Continued)
 - 1. (Continued)

Thanksgiving Day, Christmas Day, the second Tuesday in November and a day when Washington's Birthday, Memorial Day or Veterans Day is legally observed), payment for the net purchase amount will be due to the IC on the first non-holiday day following such Saturday, Sunday or holiday.

- 2. Further, if any portion of the net purchase amount is not received by the IC by the payment date as set forth in Paragraph 8.2.3, C.1, preceding, or if any portion of the net purchase amount is received by the IC in funds which are not immediately available to the IC, then a late-payment penalty shall be due the IC. The late-payment penalty shall be the portion of the net purchase amount not received by the payment date times a late factor. The late factor shall be the lesser of:
 - a. The highest interest rate (in decimal value) which may be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from the payment date to and including the date that the Telephone Company actually makes the payment to the IC, or
 - b. 0.000657 per day, compounded daily for the number of days from the payment date to and including the date that the Telephone Company actually makes the payment to the IC.

Any late-payment penalty will be included with the next Telephone Company payment to the ICED



Effective: JAN 0 1 1984

P.S.C. No. 36

No supplement to this Access Services Tariff tariff will be issued Section 8 1st Revised Sheet 62 except for the purpose of canceling this tariff. Replacing 为良伤 ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) AUG 25 1980 8.2 Billing Service-(Continued) MISSUURI Public Service Commission 8.2.3 Purchase of Accounts Receivable-(Continued) (AT)Amounts due to IC's subscribing to Message Processing and Bill Processing-Α. (Continued) (FC) 3. Payments of Net Purchase Amount to the IC-(Continued) (MT) b. (Continued) (1) The highest interest rate (in decimal value) which may be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from the payment date to and including the date that the Telephone Company actually makes the payment to the IC, or (2) 0.000657 per day, compounded daily for the number of days from the payment date to and including the date that the Telephone Company actually makes the payment to the IC. Any late-payment penalty will be included with the next Telephone Company payment to the IC. c. Also, if any adjustment that reduces an End User's balance due is received by the Telephone Company from the IC after the date the Telephone Company billed the charges to be adjusted to the End User plus 45 days (billed-plus date), then a late-payment penalty shall be due the Telephone Company. The late-payment penalty shall be the adjustment amount times a late factor. The late factor shall be the lesser of: (1) The highest interest rate (in decimal value) which may be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from billed plus to and including the date the D the Telephone Company fashs the End User's account Farler 0,88 7 SEP 26 1986 SEP 26 1986 (MT)ESTALLE COMMISSION JAN 1 Public Service Commission Issued: AUG 26 1986 SEP 26 1986 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri ŝ

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 62

DEC 2 0 1000

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)
 - C. Payments of Net Purchase Amount to the IC-(Continued)
 - 3. Also, if any adjustment that reduces an End User's balance due is solon received by the Telephone Company from the IC after the date the Telephone Company billed the charges to be adjusted to the End User plus 45 days (billed-plus date), then a late-payment penalty shall be due the Telephone Company. The late-payment penalty shall be the adjustment amount times a late factor. The late factor shall be the lesser of:
 - a. The highest interest rate (in decimal value) which may be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from billed-plus date to and including the date that the Telephone Company posts the End User's account, or
 - b. 0.000657 per day, compounded daily for the number of days from the billed-plus date to an including the date that the Telephone Company posts the End User's account.

Any late-payment penalty will be included with the adjustment made by the Telephone Company to the IC's Total Current Amount Billed.

D. Netting of IC Access Service Charges and Net Purchase Amounts

When a payment for IC Access Service Charges under this Tariff is due to the Telephone Company from the IC on the same payment date that a net purchase amount is due to the IC from the Telephone Company, the Telephone Company may, with at least 31 days notice to the IC, net the payment for IC Access Service Charges with the net purchase amount. The Telephone Company will pay the net amount to the IC on the payment date when such net amount is due the IC for require the IC to pay the Telephone Company the net amount when such net amount is due the Telephone Company. If either party does not make the payment on the payment date, a latepayment penalty as set forth in Paragraph 8.2.3, C., preceding, or Section 2, Paragraph 2.4.4 [8230, preceding, whichever is appropriate, applies.

Upon request from the IC who is responsible for auditing these amounts.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 62.01

REGEIVED

AUG 2 5 1986

MISSOURI

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued) 8.
- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable-(Continued)
- Public Service Commission A. Amounts due to IC's subscribing to Message Processing and (AT) (Continued)

3. Payments of Net Purchase Amount to the IC-(Continued)

- (Continued) c.
 - (2) 0.000657 per day, compounded daily for the number of days from the billed-plus date to an including the date that the Telephone Company posts the End User's account.

Any late-payment penalty will be included with the adjustment made by the Telephone Company to the IC's Total Current Amount Billed.

4. Netting of IC Access Service Charges and Net Purchase Amounts

When a payment for IC Access Service Charges under this Tariff is due to the Telephone Company from the IC on the same payment date that a net purchase amount is due to the IC from the Telephone Company, the Telephone Company may, with at least 31 days notice to the IC, net the payment for IC Access Service Charges with the net purchase amount. The Telephone Company will pay the net amount to the IC on the payment date when such net amount is due the IC or require the IC to pay the Telephone Company the net amount when such net amount is due the Telephone Company. If either party does not make the payment on the payment date, a latepayment penalty as set forth in Paragraph 8.2.3, A.3., preceding, or Section 2, Paragraph 2.4.1, D., preceding, whichever is appropriate, applies.

Upon request from the IC, the Telephone Company will provide a copy of its methods and procedures used to determine the amounts named in this Section (i.e., Paragraph 8.2.3, Purchase of Accounts Receivable) to the authorized representative the IC who is responsible for auditing these amounts.) に割

CANCEL

Effective Commission

26 1985

SEP 26 1986

Public Service Commission

Issued: AUG 26 1986

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

36.80

(FC)

(CT)CT)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(RT)

(RT)

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)
- (AT) B. Amounts Due the IC Providing Files of Rated Messages

Access Services Tariff Section 8 Original Sheet 62.02

REGEIVED

AUG 2 5 1980

MISSUURI

Public Service Commission

For IC's providing the Telephone Company with files of rated messages, the amount due for the purchase of its accounts receivable will be determined at the option of the Telephone Company as described in 8.2.3, A., preceding or as follows:

1. Total Current Billable Amount

The Telephone Company will upon receipt of files of rated messages, determine from its records the total amount billable to the IC's end users for IC services excluding all taxes applicable to such services. A total amount of billable revenue will be determined for each IC for each file received.

2. Recourse Adjustments

For each settlement period, the Telephone Company will make recourse adjustments to the Total Current Billable Amount as follows:

a. End User Adjustments

For each settlement period, the Telephone Company will subtract from the Total Current Billable Amount the billed amounts which the Telephone Company removes from end users balances due in accordance with IC inquiry instructions. In addition, for each settlement period, the Telephone Company will subtract from the Total Current Billable Amount an amount that equals the face value of any IC gift certificates (or its predecessor company's gift certificates) the Telephone Company has in its possession. The IC gift certificates the Telephone Company possesses will be returned to the IC.



Site .

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)

(AT) B. Amounts Due the IC Providing Files of Rated Messages (Continued)

- 2. Recourse Adjustment-(Continued)
 - b. Telephone Company and IC Adjustments

For each settlement period, the Telephone Company will subtract from the Total Current Billable Amount bill amounts for end user bills which the Telephone Company delivers to the IC. Reductions will be made for those amounts determined to be unbillable by the Telephone Company. Adjustments will also be made, when appropriate, to reflect IC pricing plans such as volume discounts where the billable revenue cannot be determined from an individual file. In addition, the Telephone Company may include adjustments to account for amounts on statements received from the IC for additions or subtractions to an end user balance due for services billed in prior periods.

Also, the Telephone Company will include adjustments to account for additions and subtractions for IC or Telephone Company prior period errors and prior period taxes calculated and billed by the Telephone Company.

c. Uncollectible Adjustments

For the settlement period, the Telephone Company will subtract from the Total Current Billable Amount an amount for uncollectibles. Uncollectibles are amounts billed by the Telephone Company to end users on Final Customer Bills that are added to the Uncollectible (realized) Accounts of the Telephone Company. The Telephone Company will determine the IC amount for uncollectibles for each settlement period by multiplying the Total Current Billable Amount by the IC uncollectible factor rounded up to the nearest 1/1000th as determined in 8.2.3, A.2 preceding.



SEP 26 1986 Public Service Commission

Access Services Tariff

RECE

Original Sheet 62.03

AUG 25 1986

MISSOURI

Public Service Commission

Section 8

Issued: AUG 26 1986

ś

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Oríginal Sheet 62.04 REGEIVED

AUG 25 1980

MISSUURI

Public Service Commission

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)
- (AT) B. Amounts Due the IC Providing Files of Rated Messages-(Continued)
 - 3. <u>Settlement for Amounts Due to the IC Who Provides Files of Rated</u> <u>Messages</u>

ACCESS SERVICES

Settlements will occur as set forth following.

a. Determination of Settlement Date

The settlement date for each file of rated messages will be determined by adding 42 days to the date the files of rated messages are processed by the Telephone Company. The Telephone Company may, in order to facilitate administration of this process, combine payments due for files received on different davs. The settlement date will then be determined by adding 42 days to the midpoint of the period (not to exceed 31 days) over which the billable messages are received. Except as provided herein, the Telephone Company will remit settlement to the IC on said settlement dates. Settlement in excess of one hundred thousand dollars (\$100,000) shall be transmitted on the payment date to a designated bank account by electronic funds transfer in immediately available funds. Any amount not required to be transmitted by electronic funds transfer shall be paid by check or draft and must be postmarked four business days prior to the payment date. If such settlement date would cause payment to be due on a Saturday, Sunday or Holiday (i.e., New Year's Day, Independence Day, Labor Day, Thanksgiving Day, Chirstmas Day, the second Tuesday in November and a day when Washington's Birthday, Memorial Day or Columbus Day is legally observed), payment for the net purchase amount will be due to the IC on the first non-Holiday day following such Saturday, Sunday or Holiday.



Issued: AUG 2 6 1986

Effective: SEP 26 1986

1.3.0. <u>iv.-</u>tu. cu

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 62.05

ACCESS SERVICES

REGEIVED

AUG 2 5 1986

MISSUURI

Public Service Commission

BILLING AND COLLECTION SERVICES-(Continued) 8.

- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable-(Continued)
- Amounts Due the IC Providing Files of Rated Messages-(ContPublic Service Commission (AT) Β.
 - 3. Settlement for Amounts Due to the IC Who Provides Files of Rated Messages-(Continued)
 - b. Late Payment Charges

Further, if any portion of the net settlement amount is received by the IC after the settlement date as set forth in 8.2.3, B.3.a, preceding, then a late payment penalty shall be due the IC. The IC will have the responsibility of notifying the Telephone Company for any applicable late payment charge.

The late payment penalty shall be the portion settlement amount not received by the payment date times a late factor. The late factor shall be the lesser of:

- (1) the highest interest rate (in decimal value) which may . be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from the payment date to and including the date that the Telephone Company actually makes the payment to the IC, or
- (2) 0.000657 per day, compounded daily for the number of days from the payment date to and including the date that the

Telephone Company actually makes the payment to the LEP Any late payment penalty will be included with Gine next Tedephone, Company payment to the IC JAN 1 Company payment to the IC.

c. Penalities Applicable to End User Balance Adjustment

Also, if any adjustment that reduces an end user balance Our in the series of the seri the Telephone Company billed the charges to be adjusted to the end user plus 45 days (billed-plus date), then a late payment [5]] penalty shall be due the Telephone Company. The late payment penalty shall be the adjustment amount times a late factor 26 1986 The late factor shall be the lesser of:

Issued: AUG 2 6 1986

Effective: SEP 26 1986

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 62.06

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.2 Billing Service-(Continued)
 - 8.2.3 Purchase of Accounts Receivable-(Continued)
- RECEIVED AUG 2 5 1986

MISSOURI

- Amounts Due the IC Providing Files of Rated Messages-(CBublic Service Commission (AT)Β.
 - 3. Settlement for Amounts Due to the IC Who Provides Files of Rated Messages-(Continued)
 - Penalties Applicable to End User Balance Adjustments-(Continued) c.
 - (1) the highest interest rate (in decimal value) which may be levied by law for commercial transactions in the state in which the Telephone Company provides Bill Processing Service and/or Private Line Billing Service to the IC, compounded daily for the number of days from the billed-plus date to and including the date that the Telephone Company posts the end user account, or
 - (2) 0.000657 per day, compounded daily for the number of days from the billed plus date to and including the date that the Telephone Company posts the end user account.

Any late payment penalty will be included with the adjustment made by the Telephone Company to the IC's Total Current Billable Amount.

d. Netting of IC Access Charges and Net Settlement Amounts

When a payment for IC Access Service Charges under this tariff is due to the Telephone Company from the IC on the same settlement date that a net settlement amount is due to the IC from the Telephone Company, the Telephone Company may, with at least 31 days notice to the IC, provide the IC with the net settlement amount due (i.e., net of the payment for the IC access and billing service charges and the net purchase amount). In addition, if the payments as previously described do not fall on the same date, netting may occur on the next accounts receivable purchase payment date if determined to be administratively feasible by the Telpehone Company in its routine conduct for the business. The Telephone Company will pay the net amount to the IC or the payment date when such bet amount is due the IC or SEP 26 SEP 26 1986

JAN 1 1988

Public Service Commission

Issued: AUG 2 6 1986

7.2.0.11.20.00

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.2 Billing Service-(Continued)
- 8.2.3 Purchase of Accounts Receivable-(Continued)
- Public Service Commission (AT)Amounts Due the IC Providing Files of Rated Messages-(Gontinued) Β.
 - 3. Settlement for Amounts Due to the IC Who Provides Files of Rated Messages-(Continued)
 - d. Netting of IC Access Charges and Net Settlement Amounts-(Continued)

require the IC to pay the Telephone Company the net amount when such net amount is due the Telephone Company. If either party does not make payment on the payment date, a late payment penalty as set forth in 8.2.3, B.3.b., preceding or 2.4.1., preceding, whichever is appropriate, applies. In those cases where netting does not occur and a payment is due if the IC customer does not make that payment by the date defined in 2.4.1, D.1., preceding, the Telephone Company may subtract the non-disputed past due amount, including the appropriate late payment penalty, from the amount due the IC by the Telephone Company. (Refer to Section 2.4.1., for handling of disputed amounts).

Upon request from the IC, the Telephone Company will provide a copy of its methods and procedures used to determine the uncollectible amounts and the Average Customer Payment Availability Period named in this section to the authorized representative of the IC who is responsible for auditing these amounts.



Access Services Tariff

Original Sheet 62.07

REGEIVED

AUG 25 1980

MISSOURI

Section 8

Issued: AUG 2 6 1986 Effective: SEP 26 1986

Access Services Tariff Section 8 / Original Sheet 63

DEC 20 (m)

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service

At the request of an IC, which is a common carrier, the Telephone Company upon reasonable notice and with reasonable effort will provide Billing Analysis Service for detection, investigation and deterrence of billing evasion activities.

The term "billing evasion activities" used herein denotes Message Billing Abuse and/or Network Abuse. Such abuse is the use of an access line (e.g., End User's line or trunk, Pay Telephone line or other access line or trunk provided by the Telephone Company) where there is intent to circumvent or evade the proper charges in whole or in part for use of telecommunications services over the access line or to conceal the points of origin or termination of telecommunications services.

Such activities include bypass or circumvention of Telephone Company billing equipment, unauthorized use of Telephone Company and IC service offerings and unauthorized use of Telephone Company facilities.

8.3.1 General Description

General Description Billing Analysis Service is the provision of defection, investigation and deterrence of billing evasion activities. JAN 1

Detection is the provision of equipment by the Telephone Company to noi identify and collect information on Network Abuse Such equipment may be located in Telephone Company premises or offices of may be attached to access lines.

Investigation is the provision of investigative services to collect evidence to document that billing evasion activities are occurring, prepare reports, preserve evidence and supply expert witness analysisand testimony. Investigation also includes provision of services to coordinate the investigative activities between exchange telephone companies and/or law enforcement agencies.

Deterrence is the contacting and interviewing of parties identified in billing evasion activities, the recovery of devices or materials... used in billing evasion activities (where legally permitted), the service review for possible suspension of service over access lines identified in billing evasion activities and publicity assistance in publicizing billing evasion deterrence.

Issued: DEC 29 1983

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 64

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service-(Continued)

8.3.2 Undertaking of the Telephone Company

DEC 2

- A. When Billing Analysis Service is ordered under a Special Order by an authorized security representative of the IC, the Telephone Company will provide any one or all of the services as set forth in Paragraphs 8.3.2, B. through D., following, at rates and charges as set forth in Paragraph 8.3.7, following.
- B. Detection Service will be provided for Network Abuse at any central office switch equipped to recognize and record irregular key pulse and multifrequency signals upon receipt of a Special Order from an IC specifying the central office(s) where detection service is requested. Subject to the agreement of the Telephone Company, special construction for the provision of central office equipment to recognize and record irregular key pulse and multifrequency signals may be undertaken at the request of the authorized security representative of the IC to meet the needs of the IC. Such special construction will be provided as set forth in Section 14, following.
 - Reports of Detection Service results will be provided on an officeby-office or line-by-line basis at the request of an authorized security representative of the IC. On an office-by-office basis, a Detection Service report of any signal irregularities will be provided for the previous 60 days for all trunks equipped in the office (i.e., ESS Control Group). Alternatively, for an office requested by the IC, a line-by-line continuous scan for current signal irregularities will be provided. When the continuous lineby-line scan is provided, a weekly report of signal irregularities for the office involved will be furnished.
 - 2. The Detection Service reports as set forth in Paragraph 8.3.2, B.1., preceding, will include for a signaling irregularity observed, the following: (a) the calling NPA and telephone number (i.e., NXX-XXXX), (b) the called number (i.e., NPA and telephone number recorded by Telephone Company billing equipment) if available, (c) signaling irregularity data, (d) the holding time of the call, if available, JAN -1 193 DAN -1 193 DAN 1 193

Lice Commission

JAN 0 1 1984

JAN - 1 198-1 83-253 Add Contas (

Issued: DEC 2 9 1983

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Effèctive:

Access Services Tariff Section 8 Original Sheet 65

DEC 20100

L. C. 在自我的自己

i Public Ser

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.3 Billing Analysis Service-(Continued)
 - 8.3.2 Undertaking of the Telephone Company-(Continued)
 - B. (Continued)
 - 2. (Continued)

The report will be provided as a paper printout or microfiche at the discretion of the Telephone Company and sent to the authorized security representative of the IC by registered first-class U.S. Mail Service. However, an authorized security representative of the IC may pick up the report at a location designated by the Telephone Company. The 60-day report will be available for mailing or pickup two weeks after the end of the 60-day period. Results of the continuous scan will be provided to the authorized security representative of the IC by a written report and/or a telephonic report within six working days after the end of a weekly scan.

- C. Investigative Service will be provided by authorized Telephone Company security personnel upon receipt of a Special Order from an authorized security representative of the IC specifying the line or lines or billing evasion activity (i.e., Message Billing Abuse and/or Network Abuse) to be investigated. The services provided include the provision of an identification report, collection of evidence, provision of a detection and documentation scan, preparation of an affidavit and prosecutive summary, preservation of evidence collected, assistance to law enforcement agencies, provision of expert witness analysis and/or testimony, the coordination of billing evasion investigative services and the review of an IC's billing evasion control programs.
 - 1. Provision of an identification report is the collection by Telephone Company personnel of billing information, party name, party address, service configuration, if any, and the preparation and submission of information pertinent to the IC's service in a report to the authorized security representative of the IC for each line or billing evasion activity specified by the IC man



Issued: DEC 2 9 1985

Effective: JAN 0 1 1984

Access Services Tariff Section 8 Original Sheet 66

DEC 20 CCC

Public Cost o Commissi

ACCESS SERVICES

- BILLING AND COLLECTION SERVICES-(Continued)
 - 8.3 Billing Analysis Service-(Continued)
 - 8.3.2 Undertaking of the Telephone Company-(Continued)
 - C. (Continued)
 - 2. Collection of evidence is the gathering of information pertinent to the line, message or party associated with the billing evasion activity specified by the IC. Collection of evidence includes a written notification to the IC of the results of such gathering efforts. The Telephone Company will determine the information provided in the written notification.
 - Subject to the agreement of the Telephone Company, a detection and 3. documentation scan on an individual line will be provided when an authorized security representative of the IC provides a written request for such a scan. Provision of a detection and documentation scan on an individual line is the continuous scan of line for irregular signals, and when an irregular signal is recognized, the collection by Telephone Company equipment of information to show an irregular signal and use of the line occurred. Any information collected during the continuous scan of a line will be considered the property of the Telephone Company. The Telephone Company will notify the IC of the results of the scan and will permit authorized security representatives of the IC to review the results. The scan and any associated information will not be provided to any person or entity until the Telephone Company is requested to do so by subpoena or lawful demands. Any out-of-pocket payments or travel and/or other expenses of Telephone Company personnel will be billed to the IC $\mathbb{C} \oplus \mathbb{H} \setminus [\mathbb{C} \oplus]$
 - Preparation of an affidavit and prosecutive summary is the whitten 4. documentation of the evidence collected by the Telephone Somp2ng 3 personnel who performed such activities.
 - 5. Preservation of evidence is the placement of the collected evidence in a secure location under the control of Telephone Company security personnel for a period of up to two years. Such preservation of evidence will be continued beyond two years by the Telephone Company when requested by appropriate law enforcement agencies or the authorized security representative of the IC. The charges for preservation of evidence as set fefth in Paragraph 8.3.7, B., fol-lowing, will apply.

miseio

JAN 0 1 1984

Issued: DEC 2 9 1983

bi Service: By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

14N1

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 67

DEC 22 mm

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service-(Continued)

- 8.3.2 Undertaking of the Telephone Company-(Continued) Public Section
- C. (Continued)
 - 6. Assistance to law enforcement is the accompanying of duly authorized law enforcement personnel to a location where billing evasion activities have been determined to exist in order to identify billing evasion activity devices or materials. The Telephone Company personnel will provide such assistance to law enforcement personnel only after law enforcement involvement has been arranged by the IC, or under services as set forth in Paragraph 8.3.2, C.10., following. If such assistance is required at locations outside the operating territory of the Telephone Company involved, any out-of-pocket payments for travel and/or other expenses of Telephone Company personnel will be billed to the IC, except to the extent that such expenses are paid by the judicial system.
 - 7. Provision of expert witness analysis is the review, study and other technical support activities provided by Telephone Company experts to analyze and/or document whether devices and materials associated with an investigation furnished by the authorized security representative of the IC or under services as set forth in Paragraph 8.3.2, C.10, following, operate, perform or contain billing evasion activities. A written analysis report will be provided to the authorized security representative of the IC. The expert will be selected by the Telephone Company.
 - 8. Provision of expert witness testimony is the preparation of testimony and the submission of such testimony in association with an investigation. A copy of the written testimony, if any, will be provided to the authorized security representative of the IC. The expert witness will be selected by the Telephone Company. Any outof-pocket payments for travel and/or other expenses of Telephone Company personnel will be billed to the IC.
 - 9. Coordination of billing evasion investigative services between telephone companies is the referral of investigative information to other telephone company security personnel, provision of billing evasion information to other telephone company security personnelly, -the Geollection of information from other telephone company security gessgnnel, and the tracking, collecting and reporting of the results of such investigations to the authorized security representative of the 0 IC.

JAN 0 1 1984

Issued: DEC 2 9 1983

BY AFFECTIVE JAN 0 1 By R. D. BARRON VICE OUFI By R. D. BARRON VICE OUFI Southwestern Bell Telephone Company St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 68

DEC 2010

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.3 Billing Analysis Service-(Continued)
 - 8.3.2 Undertaking of the Telephone Company-(Continued)
 - C. (Continued)
 - 10. Coordination of billing evasion investigative activities of participating law enforcement agencies is the provision of billing evasion information to the law enforcement agencies, the collection of billing evasion information from the law enforcement agencies and the tracking, collecting and reporting of the results of such participation to the authorized security representative of the IC. This coordination will be provided only upon receipt of written authorization from the authorized security representative of the IC.
 - 11. Review of IC billing evasion deterrence control programs and related activities is advice to and/or training of IC personnel on billing evasion deterrence and preventive controls and the development of billing evasion deterrence and preventive control programs for the IC.

The Telephone Company will, at the request of the IC, provide inves-. tigation service on a premium-time basis. When investigation service is provided on such a basis, premium-time charges as set forth in Paragraph 8.3.7, B., following, will apply.

D. Deterrence services will be provided at any location in the operating territory of the Telephone Company that Telephone Company security personnel can safely and legally enter. Deterrence services will be provided, at the request of the authorized security representative of the IC, by written or telephonic notice. A telephone notice received from the authorized security representative of the IC must be followed by written confirmation within one day. Deterrence services will be provided only after an investigation service has been provided by the Telephone Company and billing evasion activity is found to warrant such actions. Deterrence service includes the contacting and interviewing of parties identified by the Telephone Company as being involved in billing evasion activities, the recovery of devices and/or materials associated with billing evasion activities, service review for possible suspension of Telephone Company service and publicity assistance for publicizing billing evasion activity deterrence

JAN 1 1988 aublic Service Commission 2.131 - ².600

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 69

DEC 20 100

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.3 Billing Analysis Service-(Continued)
 - 8.3.2 Undertaking of the Telephone Company-(Continued)
 - D. (Continued)
 - Contacting and interviewing of parties is the written or verbal notification to, or a face-to-face discussion with, a party or parties by Telephone Company security personnel in order to deter further billing evasion activities. Contacting and interviewing is provided at locations identified by law enforcement agencies, the IC or the Telephone Company.
 - 2. Recovery of devices or materials is the attempt by Telephone Company security personnel to recover devices or materials which are used in association with Telephone Company facilities in billing evasion activities. Such recovery is provided at locations identified by the Telephone Company, law enforcement agencies or the IC.
 - 3. Service review for possible suspension of Telephone Company service is the request for authority to suspend service from appropriate Telephone Company officers, notification to the End User that service will be suspended and, after review of the End User's response, or lack thereof, suspension of service, if warranted, in the Telephone Company's judgment.
 - 4. Publicity assistance for publicizing billing evasion activity deterrence is the provision of information and personnel to aid in publicizing billing evasion activity deterrence. The Telephone Company will determine what information it will provide, if any, and will select the personnel and media, if any, to provide this service.

8.3.3 Liability of the Telephone Company

A. Notwithstanding Section 2, Paragraph 2.1.3, preceding, in the absence of willful misconduct, no liability for damages to the IC or other person or entity shall attach to the Telephone Company for its action or the conduct of its employees in providing Billing Analysis Service.

JAN 1 1988 JAN 71 (93.) this Service Commission 83-253

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 70

DEC 20 (CD)

Public Service Commission

<u>83 - 253</u>

JAN 0 1 1984

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service-(Continued)

- 8.3.4 Obligations of the IC
- A. The authorized security representative of the IC shall order all Billing Analysis Service under a Special Order. The authorized security representative of the IC shall order those Billing Analysis Services it wishes to receive.
- B. With each order, the IC shall designate and identify its authorized security representative who will be responsible to protect the information and to whom the Billing Analysis Service information will be provided. The IC shall assure and take every effort to make sure the Billing Analysis System information is provided to and used only by authorized personnel involved in billing evasion activity matters.
- C. When Billing Analysis Service Investigation is ordered, the authorized security representative of the IC shall furnish all known details of the billing evasion activities, including the access lines, parties or messages to be investigated and shall furnish all necessary End User's information it possesses to the Telephone Company security personnel. The specification of the access line, party or message to be investigated shall be in writing by the authorized security representative of the IC.
- D. When law enforcement agencies are to be brought into the investigation, the authorized security representative of the IC shall secure their participation or authorize the Telephone Company in writing to obtain and coordinate such law enforcement agency participation.
 E. When evidence
- F. When the IC requests that service be suspended for unauthorized use, the IC shall furnish a written request authorized by an officer of the IC. The request shall state the reason for the request and tariff, contract or legal provision permitting such suspension and shall correctly identify the End User's telephone number and the location of the End User's service to be suspended.
- G. All inquiries from the IC's End Users concerning services provided under this Tariff are to be handled by the IC. Any questions to the Telephone. Company shall be made by the authorized security representative of the FIC.

Effective:

Issued: DEC 2 9 1983

No supplement to this .tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.3 Billing Analysis Service-(Continued)
- 8.3.4 Obligations of the IC-(Continued)
- H. Except as set forth in Paragraph 8.3.2, D.4., preceding, publicizing of actions resulting from services provided under this Tariff shall be the responsibility of the IC. The IC shall not publicize that the Telephone Company assisted the IC unless the IC has written permission to do so from the Telephone Company.
- I. When the authorized security representative of the IC orders detection service or a detection and documentation scan of an individual line, it shall specify in writing the offices, lines or parties to be included in the scanning and reports the Telephone Company will provide.
- J. When provision of expert witness analysis is ordered by the IC, the IC shall be responsible for furnishing the evidence to be analyzed unless the services as set forth in Paragraph 8.3.2, C.10., preceding, are ordered by the IC.
- K. When provision of expert witness testimony is ordered by the IC, the IC shall include the name, if any, of the Telephone Company expert it desires and when and where the testimony is needed.
- L. When the IC wants a quotation of the total charges for the service ordered, the IC shall request such at the time of the order. Quotation regulations and charges as set forth in Section 2, Paragraph CANCELLED 2.1.9, preceding, apply.
- 8.3.5 Payment Arrangements
- A. Minimum Periods

5 Payment Arrangements Minimum Periods The minimum period for which Billing Analysis detection service con-tinuous scapping is provided and for which charges aprily is other week. tinuous scanning is provided and for which charges apply is one week.

B. Cancellation of a Special Order

An IC may cancel a Special Order for Billing Analysis Service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the authorized security representative of the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation 1 (92) within ten days. The service date for Billing Analysis Service is the 53 date the Telephone Company security organization receives the Special 53Order.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri

Access Services Tariff Section 8 Original Sheet 71

DEC 20 (CC)

自動物 行行之中 日本中華 5000

ACCESS SERVICES

Access Services Tariff Section 8 Original Sheet 72

DEC 2012

JAIN 1 1988 PS 7

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.3 Billing Analysis Service-(Continued)
 - 8.3.5 Payment Arrangements-(Continued)
 - B. Cancellation of a Special Order-(Continued)

When an IC cancels a Special Order for Billing Analysis Service after the order date but prior to the start of service, a charge as listed following shall apply:

- 1. For detection service, the per-report charge for each 60-day report ordered and the per-week charge for each office where a continuous scan is ordered.
- 2. For investigative service, two times the appropriate hourly charge for the service ordered and the per-occurrence charge for each documentation scan ordered.
- 3. For deterrence service, two times the contact and interview of parties hourly charge for the service ordered.
- C. Changes to Special Orders

IC-requested changes to a pending Special Order for Billing Analysis Service will be undertaken if they can be accommodated by the Telephone Company. However, all changes to pending Special Orders for detection service will be considered as a discontinuance of the pending order and the placement of a new order and appropriate charges will apply. Any additional time required on the part of the Telephone Company personnel will be billed to the IC at the appropriate hourly charges.

8.3.6 Rate Regulations

- A. The charge per report for the 60-day report as set for in Raragraph 8.3.7, following, applies for each report provided to is if ic even though no signaling irregularities are found.
- B. The charge per office, per week for continuous scan as set forth in Paragraph 8.3.7, following, applies for each week of service even though no signaling irregularities are found.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 Repl ACCESS SERVICES FEB 2 MISSOURI Public Service Commission

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.3 Billing Analysis Service-(Continued)
 - 8.3.6 Rate Regulations-(Continued)
 - C. The per-hour rate for investigative service and/or deterrence service is for the use of one hour of one Telephone Company security person's time."
 - D. The Telephone Company will keep a count of the hours and fraction thereof used by Telephone Company personnel to provide the service the IC ordered and will bill the IC in accordance with these records. The hours for each service ordered will be summed and then rounded to the nearest hour, except that when the total is less than one hour, one hour will be used to determine charges.
 - Ε. Premium time is all hours for work performed outside the Telephone Company's normal work schedule and/or which requires additions to the work force.
 - F. The provision of Billing Analysis Service per Special Order Charge applies for each Special Order accepted by the Telephone Company for any Billing Analysis Service.
 - 8.3.7 Rates and Charges

(C) (0) The rates and charges are:

		USOC	<u>Rates</u>
Detection Service, - 60-Day Report per office, per - Continuous Scan per office, pe		BAATE	\$640.00 290.00
<pre>Investigative Service, - Identification Report, per hou - Collection of Evidence, per hou - Documentation Scan per line, per occurrence - Preparation of Affidavit and Prosecutive Summary, per hour. - Preservation of Evidence, per - Assist Law Enforcement, per we - Provision of Expert Witness Am per hour</pre>	week JAW 1 week	Commissi SCUPI	E115.00(1)
			ce Commission
By R. D. BARRON, Vice Southwestern Bell St. Louis,	e President-Miss Telephone Compa	ouri	-

Access Services Tariff

DEC 20 CED

felse Street Comision

Original Sheet 73

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service-(Continued)

- 8.3.6 Rate Regulations-(Continued)
- C. The per-hour rate for investigative service and/or deterrence service is for the use of one hour of one Telephone Company security person's time.
- D. The Telephone Company will keep a count of the hours and fraction thereof used by Telephone Company personnel to provide the service the IC ordered and will bill the IC in accordance with these records. The hours for each service ordered will be summed and then rounded to the nearest hour, except that when the total is less than one hour, one hour will be used to determine charges.
- E. Premium time is all hours for work performed outside the Telephone Company's normal work schedule and/or which requires additions to the work force.
- F. The provision of Billing Analysis Service per Special Order Charge applies for each Special Order accepted by the Telephone Company for any Billing Analysis Service.

8.3.7	Rates and Charges				
	MAR 2 8 1984				
	The rates and charges are: By $\int \frac{dt}{R} \frac{5.273}{73}$	USOC Rates			
	Detection Service, PUBLIC SERVICE COMMISSION	BAATE			
	- 60-Day Report per office, per report	\$290.00			
	- Continuous Scan per office, per week	640.00			
	Investigative Service,	BAAVE			
	- Identification Report, per hour	114.00(1)			
	- Collection of Evidence, per hour	101.00(1)			
	- Documentation Scan				
	per line, per occurrence	610.00			
	 Preparation of Affidavit and 				
	Prosecutive Summary, per hour	114.00(1)			
	- Preservation of Evidence, per week	83.00			
	- Assist Law Enforcement, per week	- IIIL82.00			
	- Provision of Expert Witness Analysis,				
	per hour	JAN -114(300(1)			
	- Provision of Expert Witness Testimony, per hour	$83 - 253_{4.00(1)}$			
	- Coordination services, per hour	115:00(1)			
	- Review services, per hour	115.00(1)			
	niumstime newshappy wate is the times the newshappy re	to			

(1) Premium-time per-hour rate is two times the per-hour rate.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

t	No supplement to this tariff will be issued except for the purpose of canceling this tariff.	P.S.C. MoNo. 36 Replace	Access Services Tariff Section 8 2nd Revised Sheet 74 ing 1st Revised Sheet 74
		ACCESS SERVICES	REGEIVED
8	8. BILLING AND COLLECTION SERVI	CES-(Continued)	
	8.3 Billing Analysis Service-((Continued)	AUG 2 8 1986
	8.3.7 Rates and Charges-(Cont	inued)	MISSUURI Dublic Service Commission
	The rates and charges a	ore:-(continued)	Public Service Commission USOC Rates
	Deterrence Service, - Recovery of devices, p - Contact and Interview - Service Review, per ho - Publicity Assistance, Provision of Billing Ana per Special Order	of Parties, per hour . our per hour alvsis Service.	BAATR \$113.00(1) 94.00(1) 115.00(1) 115.00(1) 115.00(1)
	8.4 Billing Information Service	ce JAN J	1988 95.00 R57 ission
	8.4 Billing Information Service At the request of the IC, to the IC from its End Use	the Telephone Company wi er's records, billing fil	Commission Sond account data base.
	8.4.1 General Description	. (
(RT)	from Telephone Company Information System (CRI Such Billing Information information to an IC re vided by that IC. Info	evice is the provision of record systems labeled a (S) and Customer Name and on Service will be limite elating exclusively to En ormation relating to serv be provided, except as sp wing.	s Customer Records Address Bureau (CNA). d to the provision of d User's services pro- ices provided by any
	listings which is not l	as any entry in the reco listed as proprietary to oprietary to the Telephon	the Telephone Company.
	8.4.2 Undertaking of the Tele	ephone Company	
	A. When Billing Information pany will provide informa Paragraphs 8.4.2, B. thro forth in Paragraph 8.4.7	ation on a request-by-req bugh M., following, at th	
	(1) Premium-time per-hour rate :	is two times the per-hour	rate. Public Service Commission
-	Issued: AUG 29 1986	Effective: SEP	2 9 1986
	Southwest	DN, President-Missouri Di tern Bell Telephone Compa St. Louis, Missouri	

P.S.C. Mo	
No supplement to this tariff will be issued except for the purpose	Access Services Tariff Section 8 <u>1st Revised Sheet 74</u>
of canceling this tariff.	Replacing REGENCED
ACCESS SERV	TICES
8. BILLING AND COLLECTION SERVICES-(Contin	nued) FEB 1 1 1986
8.3 Billing Analysis Service-(Continued)	MISSOURI
8.3.7 Rates and Charges-(Continued)	Public Service Commission
The rates and charges are:-(contin	
	<u>USOC</u> <u>Rates</u>
Deterrence Service, - Recovery of devices, per hour - Contact and Interview of Parties, - Service Review, per hour - Publicity Assistance, per hour .	per hour 94.00(1) 115.00(1)
Provision of Billing Analysis Servi per Special Order	
8.4 Billing Information Service	
At the request of the IC, the Telepho to the IC from its End User's records 8.4.1 General Description	s, billing files and account data base.
from Telephone Company record syst Information System (CRIS), Nonsent and Customer Name and Address Bure Service will be limited to the pro ing exclusively to End User's serv	t-Paid Indicator Data Base (DBAS) eau (CNA). Such Billing Information ovision of information to an IC relat- vices provided by that IC. Information any other entity will not be provided,
listings which is not listed as ph	ryNin the records, data base or bureau roprietary to the Telephone Company. o the Telephone Company will not be SEP 2 9 1900
8.4.2 Undertaking of the Telephone Comp; BY 6 A. When Billing Information Service FS	ATTAL R. S. # 74
pany will provide information on a m	request-by-request basis as follows in llowing, at the rates <u>and charges</u> as set
(1) Premium-time per-hour rate is two times	
Issued: FEB 1 1 1986 Ef:	fective: MAR 1 3 1986 Public Service Commis
By R. D. BARRON, Presider Southwestern Bell To St. Louis, M	elephone Company

بكشا 12

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 74

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.3 Billing Analysis Service-(Continued)

8.3.7 Rates and Charges-(Continued)

The rates and charges are:-(continued)

DEC 20 1003

Rates

USOC

Deterrence Service,	BAATR	
- Recovery of devices, per hour		\$113.00(1)
- Contact and Interview of Parties, per hour .		94.00(1)
- Service Review, per hour		115.00(1)
- Publicity Assistance, per hour		115.00(1)
Provision of Billing Analysis Service,		
per Special Order		95.00

8.4 Billing Information Service

At the request of the IC, the Telephone Company will provide information to the IC from its End User's records, billing files and account data base.

8.4.1 General Description

Billing Information Service is the provision of information to the IC from Telephone Company record systems labeled as Customer Records Information System (CRIS), Nonsent-Paid Indicator Data Base (DBAS) and Customer Name and Address Bureau (CNA). Such Billing Information Service will be limited to the provision of information to an IC relating exclusively to End User's services provided by that IC. Information relating to services provided by any other entity will not be provided.

Information is desired as any fitry in the records, data base or bureau listings which is not listed as proprietary to the Telephone Company, Any entry listed as proprietary to the Telephone Company will not be provided. MAR 1 3 1986

8.4.2 Undertaking st the Aceptin Tompany

JAN - 1 1934 83 - 253

- Public SERVICE COMMISSION A. When Billing Informations Service is ordered by the IC, the Telephone Company will provide information on a request-by-request basis as follows in Paragraphs 8.4.2, B. through M., following, at the rates and charges as set forth in Paragraph 8.4.7, following.
- (1) Premium-time per-hour rate is two times the per-hour rate.

Issued: **DEC** 2 9 1983

Effective: JAN 0 1 1984

Access Services Tariff

Replacing Original

1st Revised Sheet 75

FEB 1 1 1986

MISSUURI Public Service Commission

Section 8

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.4 Billing Information Service-(Continued)
- 8.4.2 Undertaking of the Telephone Company-(Continued)
- B. Upon request from an IC and when the IC has ordered Message Billing Service Bill Processing Service, the Telephone Company will provide information from its CRIS records as follows:
 - 1. Message detail for a message End User.
 - 2. Account detail for a message End User.
 - 3. Service and equipment detail for a message End User.

Message detail is message-billed records in Exchange Message Record (EMR) format in the CRIS file.

Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.

Service and equipment detail is data associated with the IC's rate elements.

A message End User is an account with IC message or bulk-billed detail (for a bill period) or an account which is marked as set forth in Paragraph 8.4.2, G., following, or established as an End User of the IC's message- or bulk-billed services.

Message detail, account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall execute a written agreement with the Telephone Company as specified in 8.4.4, following.

- C. Upon request from an IC and when the IC has ordered Private Line Billing Service, the Telephone Company will provide information from its CRIS records as follows:
 - 1. Account detail for a private line End User.

2. Service and equipment detail for a private line End User CANCELLED MAR 1 3 1986 JAN 1 'ElON Public Service Commission Co Issued: FEB 1 1 1986 Mar 1 3 **1986** By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri

(CP)

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 75

DEC 2 Chan

Public Versice Commit

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.4 Billing Information Service-(Continued)
- 8.4.2 Undertaking of the Telephone Company-(Continued)
- B. Upon request from an IC and when the IC has ordered Message Billing Service Bill Processing Service, the Telephone Company will provide information from its CRIS records as follows:
 - 1. Message detail for a message End User.
 - Account detail for a message End User.
 - 3. Service and equipment detail for a message End User.

Message detail is message-billed records in Exchange Message Record (EMR) format in the CRIS file.

Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.

Service and equipment detail is data associated with the IC's rate elements.

A message End User is an account with IC message or bulk-billed detail (for a bill period) or an account which is marked as set forth in Paragraph 8.4.2, G., following, or established as an End User of the IC's message- or bulk-billed services.

Message detail, account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall secure written permission from the End User to obtain the information from the Telephone Company. The IC shall furnish the Telephone Company the End User's written permission for the information to be released.

- C. Upon request from an IC and when the IC has ordered Private Line Billing Service, the Telephone Company will provide information from its CRIS records as follows:
 - 1. Account detail for a private line End User.
 - 2. Service and equipment detail 501986private line End User. JAN 1 1934

LIC SERVICE COMMISSION

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

!

(

 tariff will be issued except for the purpose of canceling this tariff. ACCESS SERVICES BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) C. (Continued) Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail. Service and equipment detail is data associated with the IC's rate elements. A private line End User is an account with rate element detail (for a bil period) or an account which is established, at the IC's request, as an En User of the IC's private line (non-message) services. Account detail and/or service and equipment detail which is confidential due to legal, national security. End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall execute a vritten agreement with the Telephone Company as specified in 8.4.4, following. (RT) D. Where the Telephone Company facilities are available and subject to the agreement of the Telephone Company, CRIS information may be provided on an interrogation basis at the request of the IC. The interrogation basis will permit the IC to access a data file which contains the data base information file will be provided during normal Telephone Company business hours. (RT) (RT) D. Where the Telephone Company facilities are available and subject to the fold User's information. The interrogation file will be provided on an interrogation basis at the request of the IC. The interrogation basis will permit the IC to access a data file which contains the data base information file will be provided during normal Telephone Company business hours. 		Issued	+ AUG 2 9 1986	Effective: ARRON, President-Misso	: SEP 29 1986
 except for the purpose 2nd Revised Sheet 76 of canceling this tariff. Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 ACCESS SERVICES BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) C. (Continued) Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail. Service and equipment detail is data associated with the IC's rate elements. A private line End User is an account with rate element detail (for a bil period) or an account which is established, at the IC's request, as an En User of the IC's private line (non-message) services. Account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall execute a written agreement with the Telephone Company as specified in 8.4.4, following. (RT) (FC) D. Where the Telephone Company facilities are available and subject to the agreement of the Telephone Company, CRIS information may be provided on an interrogation basis at the request of the IC. The interrogation basis will permit the IC to access a data file which contains the data base information from a data processing terminal at the IC location, furnish an End User's telephone number and, after verification that the information is authorized for the IC's use, receive the End User's information. The interrogation file will be provided during normal Telephone Company torus hors. 				CANCELLEN JAN 1 1988 BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS BY 24 RS	SEP 29 1986 mmission RI Public Service Commis
 except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) C. (Continued) Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail. Service and equipment detail is data associated with the IC's rate elements. A private line End User is an account with rate element detail (for a bil period) or an account which is established, at the IC's request, as an En User of the IC's private line (non-message) services. Account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall execute a written agreement with the Telephone Company as specified in 8.4.4, following. (RT) (FC) D. Where the Telephone Company facilities are available and subject to the agreement of the Telephone Company, CRIS information may be provided on 	(RT)		contains the data bas the IC location, furn fication that the inf End User's informatio	e information from a d ish an End User's tele ormation is authorized n. The interrogation any business hours.	lata processing terminal at ephone number and, after veri- l for the IC's use, receive the file will be provided during
<pre>except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) C. (Continued) Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail. Service and equipment detail is data associated with the IC's rate elements. A private line End User is an account with rate element detail (for a bil period) or an account which is established, at the IC's request, as an En User of the IC's private line (non-message) services. Account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate require- ments will not be provided. If the IC requires this information in order to bill its services, it shall execute a written agreement with the Tele- phone Company as specified in 8.4.4, following.</pre>	(FC)	D.	agreement of the Tele	phone Company, CRIS in	formation may be provided on
except for the purpose of canceling this tariff.2nd Revised Sheet 76 Replacing 1st Revised Sheet 76ACCESS SERVICESReplacing 1st Revised Sheet 768. BILLING AND COLLECTION SERVICES-(Continued)AUG 2 & 19868.4.2 Undertaking of the Telephone Company-(Continued)AUG 2 & 1986C. (Continued)MISSUURI Public Service CommissionAccount detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.Service and equipment detail is data associated with the 1C's rate elements.A private line End User is an account with rate element detail (for a bil period) or an account which is established, at the IC's request, as an End			due to legal, nationa ments will not be pro- to bill its services,	l security, End User o vided. If the IC requ it shall execute a wr	or other appropriate require- nires this information in orden ritten agreement with the Tele-
except for the purpose of canceling this tariff.2nd Revised Sheet 76 Replacing 1st Revised Sheet 76ACCESS SERVICESACCESS SERVICES8. BILLING AND COLLECTION SERVICES-(Continued)AUG 2 & 19868.4 Billing Information Service-(Continued)AUG 2 & 19868.4.2 Undertaking of the Telephone Company-(Continued)MISSUURI Public Service CommissionC. (Continued)Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.Service and equipment detail is data associated with the IC's rate			period) or an account	which is established,	at the IC's request, as an Er
except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) C. (Continued) Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or				detail is data associ	ated with the IC's rate
except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued) B.4.2 Undertaking of the Telephone Company-(Continued)			address and billing p	arameters other than m	
except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) AUG 28 1986			-		
except for the purpose of canceling this tariff. ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 2nd Revised Sheet 76 Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 Replacing 1st Revised Sheet 76 Revised Sheet 76 Replacing 1st Revised Sheet 76 Revised Sheet			0		AND SCHOOL CH
except for the purpose 2nd Revised Sheet 76 of canceling this tariff. Replacing 1st Revised Sheet 76 ACCESS SERVICES RECEIVED					AUG 2.8 1986
except for the purpose 2nd Revised Sheet 76 of canceling this tariff. Replacing 1st Revised Sheet 76		0 DT			REGEIVED
	(except	for the purpose		Replacing 1st Revised Sheet 76

241 101 412

No supplement to this	Access Services Tariff
tariff will be issued	Section 8
except for the purpose	lst Revised Sheet 76
of canceling this tariff. Repla	cing Original Sheet 76
ACCESS SERVICES	REGEIVED
8. BILLING AND COLLECTION SERVICES-(Continued)	
8.4 Billing Information Service-(Continued)	FEB 1 1 1986
8.4.2 Undertaking of the Telephone Company-(Continued)	MISSUURI Public Service Commission
C. (Continued)	**************************************

Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.

Service and equipment detail is data associated with the IC's rate elements.

A private line End User is an account with rate element detail (for a bill period) or an account which is established, at the IC's request, as an End User of the IC's private line (non-message) services.

Account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall execute a written agreement with the Telephone Company as specified in 8.4.4, following.

- D. Upon request from an authorized supervisor of the IC or End User's information when Automatic Number Identification (ANI) service is provided to the IC by the Telephone Company or when the IC offers a telecommunications service for which the billing is based on authorized calling or called parties, the Telephone Company will provide information from its DBAS records. Only current information which resides in the data base will be provided.
- Ε. Where the Telephone Company facilities are available and subject to the agreement of the Telephone Company, CRIS and/or DBAS information may be provided on an interrogation basis at the request of the IC.

The interrogation basis will permit the IC to access a data file which contains the data base information from a data processing terminal at the IC location, furnish an End User's telephone number and, after verification that the information is authorized for the IC's use, receive the End User's information. The interrogation fille will be provided during normal Telephone Company business thours The DBAS interrogation file will be updated each business day to reflect current End User's information.

BY LIC SERVICE COMMISSION	
	MAR 1 3 1986
	PBGblic Service Commissio

(CP)

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access Services Tariff Section 8 Original Sheet 76

DEC 20 (CCS

Peblic Canica Commission

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

8.4.2 Undertaking of the Telephone Company-(Continued)

C. (Continued)

Account detail is data that furnishes the End User's name, billing address and billing parameters other than message detail and/or service and equipment detail.

Service and equipment detail is data associated with the IC's rate elements.

A private line End User is an account with rate element detail (for a bill period) or an account which is established, at the IC's request, as an End User of the IC's private line (non-message) services.

Account detail and/or service and equipment detail which is confidential due to legal, national security, End User or other appropriate requirements will not be provided. If the IC requires this information in order to bill its services, it shall secure written permission from the End User to obtain the information from the Telephone Company. The IC shall furnish the Telephone Company the End User's written permission for the information to be released.

- D. Upon request from an authorized supervisor of the IC for End User's information when Automatic Number Identification (ANI) service is provided to the IC by the Telephone Company or when the IC offers a telecommunications service for which the billing is based on authorized calling or called parties, the Telephone Company will provide information from its DBAS records. Only current information which resides in the data base will be provided.
- E. Where the Telephone Company facilities are available and subject to the agreement of the Telephone Company, CRIS and/or DBAS information may be provided on an interrogation basis at the request of the IC.

The interrogation basis will permit the IC to access a data file which contains the data base information from a data processing terminal at the IC location, furnish an End User's telephone number and, after verification that the information is authorized for the IC's use, receive the End User's information of the interpopation file will be provided during normal Telephone Company business hours. The DBAS interrogation file will be updated each business day to reflect current End User's information.

Ъ	e updated each	business day to reflect curr MAR 1 3 1986	rent End User's information. 83-253
- <u></u>		stR.5#76	
Issued:	DEC 2 9 1983	BY PUBLIC SERVICE COMMISSION OF MISSOURI Effective:	JAN 0 1 1984
	By	R. D. BARRON, Vice Presiden	
		Southwestern Bell Telephone St. Louis, Missouri	

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.4 Billing Information Service-(Continued)
 - 8.4.2 Undertaking of the Telephone Company-(Continued)
 - D. (Continued)

The CRIS interrogation file will be updated each bill day (day bills are prepared and dated for an End User for an IC's service) and will be updated daily when information is available and when the Telephone Company updates the file on a daily basis to reflect current End User's information.

- (RT) The Telephone Company will develop the IC's CRIS information order into a retrieval and interrogation program. Program development charges, as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs.
- (RT)(FC) E. CRIS information will be provided on a total-file and/or file-update basis as follows:
 - 1. The total-file basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, magnetic tape or fiche. The total-file output will contain End User's information for the current billing period. The billing period will be set by the Telephone Company. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. After the information system ordered by the IC is in service, the paper printout, magnetic tape or fiche will be available from the Telephone Company within ten working days of the IC's request.

Program development charges as set forth fallaragraph 8.4.7, following, apply for the hours required to presign, develop, test and maintain the necessary programs that are used to provide the paper output, magnetic tape or fiche.

magnetic tape or fiche. Once available, the paper printout, magnetic tape or fiche will be sent to the IC via first-class U.S. MaideService. At the option of the IC, the IC may pick up the paper printout, magnetic tape or fiche at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When-the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the casumer requests other means of transmission such as air express, the charges will be determined on an individual-case basis. SEP 29 1986

Issued: AUG 29 1986

Effective: SEP 29.19 Bublic Service Commission

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri



Access Services Tariff

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

Access S	Service	es Tar	iff
	S	Section	n 8
lst Re	evised	Sheet	77
Replacing Ori	iginal	Sheet	77

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
 - 8.4 Billing Information Service-(Continued)
 - 8.4.2 Undertaking of the Telephone Company-(Continued)
 - E. (Continued)

(AT)

The CRIS interrogation file will be updated each bill day (day bills are prepared and dated for an End User for an IC's service) and will be updated daily when information is available and when the Telephone Company updates the file on a daily basis to reflect current End User's information.

The Telephone Company will develop the IC's CRIS and DBAS information order into a retrieval and interrogation program. Program development charges, as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs.

- F. CRIS and/or DBAS information will be provided on a total-file and/or fileupdate basis as follows:
 - 1. The total-file basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, magnetic tape or fiche. The total-file output will contain End User's information for the current billing period. The billing period will be set by the Telephone Company. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. After the information system ordered by the IC is in service, the paper printout, magnetic tape or fiche will be available from the Telephone Company within ten working days of the IC's request.

Program development charges as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs that are used to provide the paper output, magnetic tape or fiche.

Once available, the paper papirton magnetic tape or fiche will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper printout, magnetic tape or fiche at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis.

		EN ED
Issued:	Effective: MAY 20 1985	MAY 20 1985
APR 19 1985	By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri	Public Service Commission

REGEIVED APR 10 1860 MISSOURI Public Service Commission

Access Services Tariff Section 8 Original Sheet 77

ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continue) DELLED DEC 200 MAY 2 0 1985 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company (Sentraned) PUBLIC SERVICE COMMISSION BY . OF MISSOURI E. (Continued)

The CRIS interrogation file will be updated each bill day (day bills are prepared and dated for an End User for an IC's service) and will be updated daily when information is available and when the Telephone Company updates the file on a daily basis to reflect current End User's information.

The Telephone Company will develop the IC's CRIS and DBAS information order into a retrieval and interrogation program. Program development charges, as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs.

- F. CRIS and/or DBAS information will be provided on a total-file and/or fileupdate basis as follows:
 - 1. The total-file basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, magnetic tape or fiche. The total-file output will contain End User's information for the current billing period. The billing period will be set by the Telephone Company. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. After the information system ordered by the IC is in service, the paper printout, magnetic tape or fiche will be available from the Telephone Company within ten working days of the IC's request.

Program development charges as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs that are used to provide the paper output, magnetic tape or fiche.

Once available, the paper printout, magnetic tape or fiche will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper printout, magnetic tape or fiche at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the date transmission charges will be determined on an individual-case basis.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 2nd Revised Sheet 78 Replacing 1st_Revised_Sheet_78 RECENVED AUG 28 1986 MISSOURI Public Service Commission

- 8.4.2 Undertaking of the Telephone Company-(Continued)

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

E. (Continued)

(RT)

2. The file-update basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, fiche or magnetic tape. The file-update output will contain End User's information for the current billing period only. The current billing period is the period associated with the most recent bill rendered to an End User. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. The file updates will include those records added and those records deleted, if any (deleted records may not be retained by the Telephone Company). For CRIS information, the file updates will be provided.

Program development charges as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs that are used to provide the paper output, fiche or magnetic tape.

Once available, the file update paper printout, fiche or magnetic tape will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis.



SEP 29 1986

Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986

Access Services Tariff Section 8 1st Revised Sheet 78 Replacing Original Sheet 78

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.4 Billing Information Service-(Continued)
- 8.4.2 Undertaking of the Telephone Company-(Continued)



- F. (Continued)
 - 2. The file-update basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, fiche or magnetic tape. The file-update output will contain End User's information for the current billing period only. The current billing period is the period associated with the most recent bill rendered to an End User. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. The file updates will include those records added and those records deleted, if any (deleted records may not be retained by the Telephone Company). For CRIS information, the file updates will be provided for each business day.

Program development charges as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs that are used to provide the paper output, fiche or magnetic tape.

Once available, the file update paper printout, fiche or magnetic tape will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined for an individual-case basis.

SEP 2 9 1986

PUBLIC SERVICE COMMISSION

OF MISSOURI



By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

Effective: MAY 20 1985

1月1111日回 MAY 201985

Public Service Commission



(AT)
No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 78

Public Sci

DEC 20 (11)

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

8.4.2 Undertaking of the Telephone Company-(Continued)

F. (Continued)

2. The file-update basis will permit the IC to receive, at the IC's option, all the End User's information that is authorized for the IC's use on paper printout, fiche or magnetic tape. The file-update output will contain End User's information for the current billing period only. The current billing period is the period associated with the most recent bill rendered to an End User. The magnetic tapes will be provided without the return of previously supplied tapes. The Telephone Company will supply the magnetic tapes. The file updates will include those records added and those records deleted, if any (deleted records may not be retained by the Telephone Company). For CRIS information, the file updates will be provided for each business day.

Program development charges as set forth in Paragraph 8.4.7, following, apply for the hours required to design, develop, test and maintain the necessary programs that are used to provide the paper output, fiche or magnetic tape.

Once available, the file update paper printout, fiche or magnetic tape will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis.



|7 ...|∟|3|.) JAN = 1 183 : SAM = 2 5 3

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

F.5.C. II	0NO. 30	
No supplement to this	Acc	ess Services Tariff
tariff will be issued		Section 8
except for the purpose	2	2nd Revised Sheet 79
of canceling this tariff.	Replacing 1	lst Revised Sheet 79
ACCESS S	ERVICES	RECEIVED
8. BILLING AND COLLECTION SERVICES-(Con	tinued)	aug 2 8 1986
8.4 Billing Information Service-(Conti	nued)	
8.4.2 Undertaking of the Telephone Co	mpany-(Continued)	MISSOURI Public Service Commission
E. (Continued)	,	میں ایک اور میں ایک میں میں ایک میں میں ایک اور میں میں ایک اور میں میں میں ایک اور میں میں ایک ایک میں میں ایک ایک اور این ایک ایک ایک ایک ایک ایک ایک ایک ایک ایک

- 3. The total-file output and the file-update output will, at the option of the IC, be provided on a quick turnaround basis. Such quick turn
 - around output will be provided one working day after the information that the IC ordered is available. Once available, the output will be provided on paper printout, fiche or magnetic tape and will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis.
- (FC) F. The Telephone Company will, at the request of the IC, mark any messagebilled message End User's account, other than End User's accounts with IC credit cards or rate elements, as a user of the IC's message services. After marking is ordered, the End User's account will be marked as an IC End User's account at the time the first message is posted to the End User's account. If not marked at the request of the IC, such an End User's account will not be identified as an IC account unless there are IC message details associated with the account for the bill period for which message detail is ordered by the IC. The mark will be removed at the request of the IC. Charges apply to mark the account and maintain the mark in future months as set forth in Paragraph 8.4.7,G., following. IC bulk-billed message End User's accounts and message End User's accounts with IC credit cards or IC rate elements are counted as IC accounts.



SFP 29 1986 Public Service Commission

Issued: AUG 29 1986

Effective: SEP 29 1986

ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

tinued)	MISSOURI Public Service Commission					
	APR 15 CCC					
Replacing Original Sheet 79 RECEIVED						
Access Services Tariff Section 8 1st Revised Sheet 79						

8.4.2 Undertaking of the Telephone Company-(Con

BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

F. (Continued)

8.

(AT)

- 3. The total-file output and the file-update output will, at the option of the IC, be provided on a quick turnaround basis. Such quick turnaround output will be provided one working day after the information that the IC ordered is available. Once available, the output will be provided on paper printout, fiche or magnetic tape and will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis. When the customer requests other means of transmission such as air express, the charges will be determined on an individual-case basis.
- G. The Telephone Company will, at the request of the IC, mark any messagebilled message End User's account, other than End User's accounts with IC credit cards or rate elements, as a user of the IC's message services. After marking is ordered, the End User's account will be marked as an IC End User's account at the time the first message is posted to the End User's account. If not marked at the request of the IC, such an End User's account will not be identified as an IC account unless there are IC message details associated with the account for the bill period for which message detail is ordered by the IC. The mark will be removed at the request of the IC. Charges apply to mark the account and maintain the mark in future months as set forth in Paragraph 8.4.7G., following. IC bulk-billed message End User's accounts and message End User's accounts with IC credit cards or IC rate elements are counted as IC accounts.



Issued: APR 191985 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri Public Service Commission

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 79

DEC 20 (CC)

Public Service Commission

ACCESS SERVICES

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.4 Billing Information Service-(Continued)
 - 8.4.2 Undertaking of the Telephone Company-(Continued)
 - F. (Continued)
 - 3. The total-file output and the file-update output will, at the option of the IC, be provided on a quick turnaround basis. Such quick turnaround output will be provided one working day after the information that the IC ordered is available. Once available, the output will be provided on paper printout, fiche or magnetic tape and will be sent to the IC via first-class U.S. Mail Service. At the option of the IC, the IC may pick up the paper output, fiche or magnetic tape at a location designated by the Telephone Company or request the information be data-transmitted to the IC. When the information is data-transmitted to the IC, the data transmission charges will be determined on an individual-case basis.
 - G. The Telephone Company will, at the request of the IC, mark any messagebilled message End User's account, other than End User's accounts with IC credit cards or rate elements, as a user of the IC's message services. After marking is ordered, the End User's account will be marked as an IC End User's account at the time the first message is posted to the End User's account. If not marked at the request of the IC, such an End User's account will not be identified as an IC account unless there are IC message details associated with the account for the bill period for which message detail is ordered by the IC. The mark will be removed at the request of the IC. Charges apply to mark the account and maintain the mark in future months as set forth in Paragraph 8.4.7, G., following. IC bulk-billed message End User's accounts and message End User's accounts with IC credit cards or IC rate elements are counted as IC accounts.



停制上屋的 JAN - 1 (934 83 - 253 7. KB

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 8 except for the purpose of canceling this tariff. Replacing ACCESS SERVICES AHG 28 1986 8. BILLING AND COLLECTION SERVICES-(Continued) MISSOURI 8.4 Billing Information Service-(Continued) Public Service Commission 8.4.2 Undertaking of the Telephone Company-(Continued) (FC) G. Upon acceptance by the Telephone Company of a Special Order for Billing Information Service from an IC, the Telephone Company will determine the period of time to implement such service on an individual-order basis. (FC) Η. The Telephone Company will provide the format for interrogation of its data files and the format of any printer, magnetic tape or fiche output (RT) from its CRIS. (FC) Upon request from an authorized supervisor of the IC who furnishes the 1. account code assigned by the Telephone Company, the Telephone Company will provide name and town information from its CNA bureau. The CNA name and town data, but not street address, will be provided only when the IC needs the information to authorize a call, to bill a call or to handle an emergency situation. The information will be provided on a request-by-request basis by voice telecommunications. Name, town and state will be provided for a telephone number. A request includes the handling of one call and providing the data for one telephone number. The Telephone Company will specify the location where requests are to be received and the format in which the request is to be made. If the name and address associated with the telephone number is restricted due to the request of the End User, legal authority or LED enforcement agency, no name or town location will be provided. ELLA At the request of the IC, written confirmation of the name and town J. If the IC requests the information ordered by the IC be resumption by the Telephone Company because of incorrect IC specifications of errors, the Telephone Company will resupply the information in accordance with a new IC order and all appropriate charges as a set of the location will be sent to the authorized supervisor making the request (FC) (FC) Where facilities are available and subject to the agreement of the Tele-Κ. phone Company, updating of the IC data bases or files from Telephone Company data processing terminals or equipment in Telephone Company peations may be undertaken at the request of the IC. The charges for such a service will be determined on an individual-case basis. SFP 29 1986 Effective: SEP 29 1986 Service Commission Issued: AUG 29 1986 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company

St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

- 8.4.2 Undertaking of the Telephone Company-(Continued)
- H. Upon acceptance by the Telephone Company of a Special Order for Billing Information Service from an IC, the Telephone Company will determine the period of time to implement such service on an individual-order basis.
- I. The Telephone Company will provide the format for interrogation of its data files and the format of any printer, magnetic tape or fiche output from its CRIS and DBAS files.
- J. Upon request from an authorized supervisor of the IC who furnishes the account code assigned by the Telephone Company, the Telephone Company will provide name and town information from its CNA bureau. The CNA name and town data, but not street address, will be provided only when the IC needs the information to authorize a call, to bill a call or to handle an emergency situation. The information will be provided on a request-by-request basis by voice telecommunications. Name, town and state will be provided for a telephone number. A request includes the handling of one call and providing the data for one telephone number.

The Telephone Company will specify the location where requests are to be received and the format in which the request is to be made.

If the name and address associated with the telephone number is restricted due to the request of the End User, legal authority or law enforcement agency, no name or town location will be provided.

At the request of the IC, written confirmation of the name and town location will be sent to the authorized supervisor making the request by first-class U.S. Mail Service.

- K. If the IC requests the information fordered by the IC be resupplied by the Telephone Company because of incorrect IC specifications or errors, the Telephone Company will resupply the Sinformation in accordance with a new IC order and all appropriate charges as set forth in Paragraph 8.4.7, following, will apply. BY SHOW BY SHOW 83-253
- L. Where facilities are available and subject to the agreement of the Telephone Company, updating of the IC data bases or files from Telephone Company data processing terminals or equipment in Telephone Company locations may be undertaken at the request of the IC. The charges for such a service will be determined on an individual-case basis.

Issued: DEC 29 1983

Effective: JAN 0 1 1984

By R. D. BARRON, Vice President-Missouri Southwestern Bell Telephone Company St. Louis, Missouri DEC 2 0 200

Public Service Commission

Original Sheet 80

Section 8

Access Services Tariff

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Ro ACCESS SERVICES 8. BILLING AND COLLECTION SERVICES-(Continued) 8.4 Billing Information Service-(Continued) 8.4.2 Undertaking of the Telephone Company-(Conta (FC) L. The Telephone Company will provide Billing In Special Order. For all Billing Information S

MISSOURI	n
AUG 28 1986	
2nd Revised Sheet 81 Replacing Ist Revised Sheet 81 REGEIVED	
Access Services Tariff Section 8	

Public Service Commission

-) L. The Telephone Company will provide Billing Information Service under a Special Order. For all Billing Information Services, the Billing Information Service Special Order Charge as forth in Paragraph 8.4.7, following, applies.
 - 8.4.3 Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, in the absence of willful misconduct, no liability for damages to the IC or other person or entity shall attach to the Telephone Company for its action or the conduct of its employees in providing Billing Information Service.

- 8.4.4 Obligations of the IC
- A. The IC shall order Billing Information Service under a Special Order. The IC shall order those Billing Information Services for the state where it wishes to receive the services and shall specify how often it wishes the service to be provided.
- B. With each order, the IC shall identify the authorized individual and address to receive the Billing Information Service output. When interrogation is ordered, the IC shall identify the data processing terminals authorized to receive the information and the authorized individual who will be responsible for all terminal activities. When CNA service is ordered, the IC will identify in writing and include the account codes assigned by the Telephone Company of all authorized individuals who will contact the CNA bureau.

C. The IC shall take every effort to make sure that Billing Information Service output and interrogation capabilities are provided only to authorized personnel of the IC or third parties performing billing services-related work for the IC. The IC shall execute a written agreement satisfactory to the Telephone Company, that the IC will treat the information in accordance with this tariff, will treat the information as confidential and proprietary data, and will use the information only for the purpose of billing its End Users. The written agreement shall also provide for idemfification of the Telephone Company for damages Fesulting from improper release of the information.

100

SEP 29 1986

Issued: AUG 29 1986

ACCESS SERVICES

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

	Access Services Tariff Section 8 1st Revised Sheet 81 cing Original Sheet 81				
RECEIVED					
	FEB 1 1 1986				

WI2200KI

Public Service Commission

8. BILLING AND COLLECTION SERVICES-(Continued)

- 8.4 Billing Information Service-(Continued)
- 8.4.2 Undertaking of the Telephone Company-(Continued) Public Service Commission
 - M. The Telephone Company will provide Billing Information Service under a Special Order. For all Billing Information Services, the Billing Information Service Special Order Charge as forth in Paragraph 8.4.7, following, applies.
- 8.4.3 Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, in the absence of willful misconduct, no liability for damages to the IC or other person or entity shall attach to the Telephone Company for its action or the conduct of its employees in providing Billing Information Service.

- 8.4.4 Obligations of the IC
- A. The IC shall order Billing Information Service under a Special Order. The IC shall order those Billing Information Services for the state where it wishes to receive the services and shall specify how often it wishes the service to be provided.
- B. With each order, the IC shall identify the authorized individual and address to receive the Billing Information Service output. When interrogation is ordered, the IC shall identify the data processing terminals authorized to receive the information and the authorized individual who will be responsible for all terminal activities. When CNA service is ordered, the IC will identify in writing and include the account codes assigned by the Telephone Company of all authorized individuals who will contact the CNA bureau.

C. The IC shall take every effort to make sure that Billing Information Service output and interrogation capabilities are provided only to authorized personnel of the IC or third parties performing billing services-related work for the IC. The IC shall execute a written agreement satisfactory to the Telephone Company, that the IC will treat the information in accordance with this tariff, will treat the information as confidential and proprietary data, and will use the information billing for the purpose of pilling its End Users. The written agreement shall also provide for information of the Telephone Company for damages resulting from inproper release of the information.

Effective:

MAR 1 3 1980

Issued: FEB 1 1 1986

(CP)

By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company St. Louis, Missouri

1 cm 2

PUBLIC SERVICE

Access Services Tariff Section 8 Original Sheet 81

ACCESS SERVICES

DEC 2 8 (283

- 8. BILLING AND COLLECTION SERVICES-(Continued)
- 8.4 Billing Information Service-(Continued)
 - 8.4.2 Undertaking of the Telephone Company-(Continued)
 - M. The Telephone Company will provide Billing Information Service under a Special Order. For all Billing Information Services, the Billing Information Service Special Order Charge as forth in Paragraph 8.4.7, following, applies.
 - 8.4.3 Liability of the Telephone Company

Notwithstanding Section 2, Paragraph 2.1.3, preceding, in the absence of willful misconduct, no liability for damages to the IC or other person or entity shall attach to the Telephone Company for its action or the conduct of its employees in providing Billing Information Service.

- 8.4.4 Obligations of the IC
- A. The IC shall order Billing Information Service under a Special Order. The IC shall order those Billing Information Services for the state where it wishes to receive the services and shall specify how often it wishes the service to be provided.
- B. With each order, the IC shall identify the authorized individual and address to receive the Billing Information Service output. When interrogation is ordered, the IC shall identify the data processing terminals authorized to receive the information and the authorized individual who will be responsible for all terminal activities. When CNA service is ordered, the IC will identify in writing and include the account codes assigned by the Telephone Company of all authorized individuals who will contact the CNA bureau.
- C. The IC shall take every effort to make sure that Billing Information Service output and interrogation capability is provided only to authorized personnel. The IC shall agree in writing to the Telephone Company that the IC will not provide the Billing Information Service outputs to third parties for any use by such third parties, except for work for the IC and which is under complete control of the IC re-

which is under complete control of the HELED JAN - 1 1934 MAR 1 3 1986 83-253 NR.S.#-8 Public Service Pra SERVICE COMMISSION OF MISSOUN Issued: Effective: DEC 2 9 1983 JAN 0 1 1984

Access Services Tariff

Replacing Origina

1st Revised Sheet 82

AUG 28 1986

MISSUURI Public Service Commission

Section 8

82

speet

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

8.4.4 Obligations of the IC-(Continued)

- D. The IC shall furnish to the Telephone Company when interrogation service is ordered, all information necessary to allow the Telephone Company to establish an interrogation program. In addition, the IC shall furnish the Telephone Company for each data base and file where the interrogation is ordered, an estimate of the number of requests per business day that the Telephone Company data bases and files will be asked to handle. The IC's terminals used to interrogate the Telephone Company data bases and files must be capable of working with Telephone Company equipment and software.
- E. The IC shall be responsible for all contacts and inquiries from its End Users concerning Billing Information Service.
- F. The IC shall not publicize or represent to others that the Telephone Company jointly participates with the IC in the development of the IC's End User's records, accounts, data bases or market data, records, files and data bases or other systems it assembles through the use of Billing Information Service.
- (RT)
- (FC) When the IC orders marking of non-IC credit card message-billed message G. End User's accounts, all accounts containing that IC's messages will be marked starting with the next bill period and marking will continue until the IC orders marking discontinued. The IC shall, when it orders marking removed, furnish the telephone number of each End User's account for which the mark is to be removed. The mark will be removed prior to the next bill period.

8.4.5 Payment Arrangements

A. Minimum Periods and Minimum Monthly Charges

The minimum period for which Billing Information Service CRIS file in-(RT) terrogation is provided and for which tharges apply is on CANCELLER 1988 SEP 29 1986 BEION Public Service Commi Public Service Commission MERFECTIVE: SEP 29 1986

Issued: AUG 29 1986

Access Services Tariff Section 8 Original Sheet 82

DEC 2 CO

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

8.4.4 Obligations of the IC-(Continued)

- D. The IC shall furnish to the Telephone Company when interrogation service is ordered, all information necessary to allow the Telephone-Company to establish an interrogation program. In addition, the IC shall furnish the Telephone Company for each data base and file where the interrogation is ordered, an estimate of the number of requests per business day that the Telephone Company data bases and files will be asked to handle. The IC's terminals used to interrogate the Telephone Company data bases and files must be capable of working with Telephone Company equipment and software.
- E. The IC shall be responsible for all contacts and inquiries from its End Users concerning Billing Information Service.
- F. The IC shall not publicize or represent to others that the Telephone Company jointly participates with the IC in the development of the IC's End User's records, accounts, data bases or market data, records, files and data bases or other systems it assembles through the use of Billing Information Service.
- G. When the IC wants a quotation of the charges for the service ordered, the IC shall request such at the time of the order. Quotation regulations as set forth in Section 2, Paragraph 2.4.3, preceding, apply.
- H. When the IC orders marking of non-IC credit card message-billed message End User's accounts, all accounts containing that IC's messages will be marked starting with the next bill period and marking will continue until the IC orders marking discontinued. The IC shall Evhen it orders marking removed, furnish the telephone number of each End User's account for which the mark is to be removed. The mark will be removed prior to the next bill period.

SEP 2 9 1986

8.4.5 Payment Arrangements

BY JATRS, #82

A. Minimum Periods and Minimum Monthly Ghargesice COMMISSION

The minimum period for which Billing Information Service CRIS and/or DBAS file interrogation is provided and for which charges apply is one year.

Issued: **DEC 2 9 1983**

Effective: JAN 0 1 1984 JAN - 1 (53) R. D. BARRON, Vice President-Missouri 83-253 Southwestern Bell Telephone Company Public Service Commission By R. D. BARRON, Vice President-Missouri St. Louis, Missouri

No supplement to this tariff will be issued except for the purpose of canceling this tariff.

(RT)

Access Services Tariff Section 8 1st Revised Sheet 83 Replacing Original Sheet 83

ACCESS SERVICES

BILLING AND COLLECTION SERVICES-(Continued) 8.

8.4 Billing Information Service-(Continued)

8.4.5 Payment Arrangements-(Continued)



AUG 28 1986

MISSOURI

Public Service Commission

- A. Minimum Periods and Minimum Monthly Charges-(Continued Public Service Commission
- The minimum monthly charges for CRIS file interrogation are the charges for the total number of requests per business day furnished by the IC as set forth in Paragraph 8.4.4, D., preceding, times 18 (i.e., 20 business days per month times 0.9).

When the IC discontinues the service prior to the end of the one-year minimum period, the minimum monthly charge for the data base interrogation will apply for each remaining month and fraction of month.

B. Cancellation of a Special Order

An IC may cancel a Special Order for Billing Information Service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for a Billing Information Service is the date the Telephone Company notifies the IC that the Telephone Company is ready to provide Billing Information Service reports or receive interrogation requests.

When an IC cancels a Special Order for Billing Information Service after the order date but prior to the start of service, charges as listed following shall apply:

- For any service, the appropriate per-hour rate for all hours, LED 1. expended by the Telephone Company to provide the service CA
- Fublic Service Commission 2. For any service, any expense for equipment obtained for they service where such equipment cannot be reused within six months.
- C. Changes to Special Orders

When an IC requests changes for a pending Special Order for Billing Information Service, they will be undertaken if they can be accommodated by the Telephone Company. Any additional time required on the part of Telephone Company personnel will be billed to the IC at the appropriate Siderly charges.

Issued: AUG 29 1986

Effective: SEP 29 1986-

Access Services Tariff Section 8 Original Sheet 83

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

DEC 25 KCB

8.4 Billing Information Service-(Continued)

8.4.5 Payment Arrangements-(Continued)

A. Minimum Periods and Minimum Monthly Charges-(Continued)

The minimum monthly charges for CRIS and/or DBAS file interrogation are the charges for the total number of requests per business day furnished by the IC as set forth in Paragraph 8.4.4, D., preceding, times 18 (i.e., 20 business days per month times 0.9).

When the IC discontinues the service prior to the end of the one-year minimum period, the minimum monthly charge for the data base interrogation will apply for each remaining month and fraction of month.

B. Cancellation of a Special Order

An IC may cancel a Special Order for Billing Information Service on any date prior to the service date. The cancellation date is the date the Telephone Company receives written or verbal notice from the IC that the Special Order is to be cancelled. The verbal notice must be followed by written confirmation within ten days. The service date for a Billing Information Service is the date the Telephone Company notifies the IC that the Telephone Company is ready to provide Billing Information Service reports or receive interrogation requests.

When an IC cancels a Special Order for Billing Information Service after the order date but prior to the start of service, charges as listed following shall apply:

- 1. For any service, the appropriate per-hour rate for all hours expended by the Telephone Company for provide the service.
- 2. For any service, any expense for equipment obtained for the service where such equipment cannot be reused within six months.

BY AtR. S. # 83 PUBLIC SERVICE COMMISSION

C. Changes to Special Orders

When an IC requests changes for a pending Special Order for Billing Information Service, they will be undertaken if they can be accommodated by the Telephone Company. Any additional time required on the part of Telephone Company personnel will be billed to the IC at the appropriate hourly charges.

Issued: DEC 2 9 1983

Effective: JAN 0 1 1984

³⁴ JAN = 1 1982 **8 3 - 2 5 3** Public Service Commission

Access Services Tariff Section 8 Original Sheet 84

DEC 20 203

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

8.4 Billing Information Service-(Continued)

8.4.6 Rate Regulations

- A. The number and type of records for which charges apply as set forth in Paragraph 8.4.7, following, will be accumulated by the Telephone Company, and the Telephone Company will bill the IC in accordance with these accumulations. A record is a logical grouping of information as described in the programs that process the information, print the paper output and load the magnetic tape or file used to supply the detail which is datatransmitted or put on fiche. For each service and type of output ordered, the number of records processed by the Telephone Company to prepare the output will be used to determine the charges. The number of records processed will be determined using the number of records input to or the number of records output from the programs that process the information, print the paper output and load the magnetic tape or data file, whichever number of records is higher.
- B. The number of hours and fraction thereof for which charges apply as set forth in Paragraph 8.4.7, following, will be accumulated by the Telephone Company. The per-hour rate is for the use of one hour of one Telephone Company's programmer. The Telephone Company will bill the IC for hourly charges in accordance with these accumulations. The accumulated hours for each order will be summed and rounded to the nearest hour, except that when the total is less than one hour, one hour will be used to determine the charges.
- C. When a CNA request is received, the Telephone Company with Creep a count of the requests. The Telephone Company will bill the CC in accordance with these records even though the Telephone Company was mothable to provide a name and town location for all requests.
- D. When records are entered on a data file or magnetic tape in redries to provide information to an IC, the per-tape charge applies for each data file or tape prepared. In addition, the per-record charge applies for each record entered on the data file or tape. The Telephone Company will determine the charges based on the number of data files or tapes prepared and on its count of the records entered on the data file or tape.
- E. When marking of message End User's accounts is ordered, the marking charge applies for each End User's account marked. Once an account is marked, the maintenance of mark charge applies for each month following the month the account is marked until the IC requests the mark be removed. No charges apply to remove the mark.

Issued: DEC 2 9 1983

Effective: JAN

JAN 0 1 1984 JAN - 1 1984 **83 - 253** Missouri Public Centre Commany

			thwestern	Bell Te	nt-Missouri elephone Co Missouri			
	Issued:	AUG 29 1986		Ef	fective: S	EP 29 1986		
	(1) The the	Quick Turnaround p ten-working-day, p	er-record er-record	charge	and the pe	r-tape charg	SEP 29 198 se is three the second	6 times MISSIUIII
(RT)		CRIS File Interro per request rece	ived	^{1FS+} م	ANI 1 1988	nmisel pr	.20	
(DR)						F	lates	
		- Fiche output, per record pro	cessed .	. 1FSC+	.0005	.0005	.0005	
		- Magnetic tape, per record pro per tape or da			.0005	.0005	.0005	\$41.00
		- Paper output, per record pro	cessed .	. 1FSC+	\$.0005	\$.0005	\$. 0 005	
		CRIS Ten-Working- Information Servi						
				USOC	Message Detail	Account Detail	and Equipment Detail	Detail on _Tape
		The rates and cha	rges are:			Rates	(1) Service	
	8.4.7	Rates and Charges					-	
	a	he provision of Bi pplies for each Sp ny Billing Informa	ecial Ord	er accep				
	8.4.6	Rate Regulations-	(Continue	d)		Public	Service Comm	ission
	8.4 Bi	lling Information	Service-(Continue	d)		MISSOURI	
	8. BILI	ING AND COLLECTION					\ UG 2 8 1986	
}	of cance	ling this tariff.	۵۵۵	ESS SERV		Replacing_Or		
	tariff w except f	ill be issued or the purpose				lst R	Sectio evise <u>d S</u> heet	n 8 85
	No suppl	ement to this	P.S	.C. Mo	No. 36	Access	Services Tar	iff

باستفحار ر

No supplement to this tariff will be issued except for the purpose of canceling this tariff. Access Services Tariff Section 8 Original Sheet 85

ACCESS SERVICES

8. BILLING AND COLLECTION SERVICES-(Continued)

DEC 2)] 3

- 8.4 Billing Information Service-(Continued)
- 8.4.6 Rate Regulations-(Continued)
- F. The provision of Billing Information Service per Special Order Charge applies for each Special Order accepted by the Telephone Company for any Billing Information Service.

8.4.7 Rates and Charges

The rates and charges are:

The rates and charges are	2:		Rates(1)			
					Service	
		<u>USOC</u>	Message Detail	Account Detail	and Equipment Detail	Detail on <u>Tape</u>
CRIS Ten-Working-Day Information Service						
- Paper output, per record processed	• • • •	1FSC+	\$.0005	\$.0005	.\$.0005	
- Magnetic tape, per record processed per tape or data filo			.0005	0005	.0005	\$41.0 0
- Fiche output, per record processed	• • • •	1FSC+	.0005	.0005	.0005	
				Rat	es	
DBAS Information Service	,					
- Paper output, per record processed		1FSD+		\$.0005	
- Magnetic tape, per record processed per tape or data fil - Fiche output,		1FSD+ 1FSD+ ଜଣ	NGEI	LEDA	.0005 .00	
per record processed		1FSD+	SEP 2 9	1986	.0005	
CRIS File or DBAS File Interrogation, per request received.		1FS+PY PU	ptl SA	COMMISSION	. 20	
(1) The Quick Turnaround per-reco the ten-working-day, per-reco	-			-	is three ti	
Issued: DEC 2 9 1983	Ef	fective	JAN 0	1 1984	HAN - 1 490 88 - 25	8
By R. D. BARR Southweste St		elephor	ne Company		o S <mark>er</mark> vida Cona	

		ement to this ill be issued	Access	Services Tariff Section 8
-	except fo	or the purpose	lst F	Revised Sheet 86
	of cancel	ling this tariff.	Replacing Or	riginal Sheet 86
		ACCESS SERVICES	I	REGEIVED
	8. BILL	ING AND COLLECTION SERVICES-(Continued)		
	8.4 Bi	lling Information Service-(Continued)		APR 15 1865
	8.4.7	Rates and Charges-(Continued)		MISSOURI
		The rates and charges are:-(Continued)	Waaa	Public Service Commission
			USOC	Rates
		Program Development Charge,		Å107 00
		- Basic, per hour	•	\$104.00
		- Premium, per hour		128.00
		CNA Information Service,		
		- CNA interrogation, per request received	. CNT	.35
		- CNA interrogation confirmation, per request confirmed		.50
			FID	
		Data transmission to an IC location of Billing Information	BOD I	ነጥ
		Service details,	501 1	ICB rates and
		per record transmitted	•	charges apply.
	:)	JAIN 1 JAIN 1 BY 24 AUSTIC BORVICE	ED 1988 RS 7 Commission Commission	
	·		<u></u>	EN ED
	Issued:	Effective:	MAY 20 1985	MAY 2 0 1985
	APR 19 1	By R. D. BARRON, President-Missouri Southwestern Bell Telephone C St. Louis, Missouri		Public Service Commission
			5 A.	

P.S.C. Mo.-No. 36 No supplement to this Access Services Tariff tariff will be issued Section 8 except for the purpose Original Sheet 86 of canceling this tariff. ACCESS SERVICES BILLING AND COLLECTION SERVICES-(Continued) DEC 2) ())) 8.4 Billing Information Service-(Continued) 8.4.7 Rates and Charges-(Continued) The rates and charges are:-(Continued) USOC Rates Program Development Charge, \$104.00 - Basic, per hour (applicable to work performed within the Telephone Company's normal work schedule and using CANGELLED the normal work force) - Premium, per hour 128.00 (applicable to work performed MAY 2 0 198 outside the Telephone Company's normal work schedule and/or which requires additions to PUBLIC SERVICE COMMISSION the work force) BY OF MISSOURI CNA Information Service, - CNA interrogation, CNT .35 - CNA interrogation confirmation, per request confirmed.50 FID Data transmission to an IC location BOD DT of Billing Information Service Service details, ICB rates and per record transmitted charges apply. Marking of Message End User's Accounts, - Marking, per End User's account \$.10 - Maintenance of mark, per End User's .005 Updating of IC Data Bases or Files, ICB rates and per record transmitted charges apply. Provision of Bill Information Service, 1 \$¹ 95:00 253 JAN 0 1 1984 83 DFC 29 1983 Effective: Issued: By R. D. BARRON, Vice President-Missouri

Southwestern Bell Telephone Company St. Louis, Missouri

and the second second second second second second second second second second second second second second second

- **1**

	P.S.C. MoNo. 36	
tariff except	plement to this will be issued for the purpose celing this tariff.	Access Services Tariff Section 8 Original Sheet 87
8. BII	ACCESS SERVICES LLING AND COLLECTION SERVICES-(Continued)	REGEIVED
	Billing Information Service-(Continued)	APR 15 100
8.4.7	7 Rates and Charges-(Continued)	MISSOURI Public Service Commission
	The rates and charges are:-(Continued)	USOC Rates
(MT)	Marking of Message End User's Accounts,	
	- Marking, per End User's account	\$.10
	- Maintenance of mark, per End User's account, per month	.005
(MT)	Updating of IC Data Bases or Files, per record transmitted	ICB rates and charges apply.
	Provision of Bill Information Service, per Special Order	\$ 95.00
(NR)	Other transmission to a customer location per tape or data file	ICB rates and charges apply.



णडागार MAY 20 1985 Issued: Effective: . APR 1 9 1985 MAY 20 1985 By R. D. BARRON, President-Missouri Division Southwestern Bell Telephone Company Public Service Commission St. Louis, Missouri